ETSI TS 123 205 V9.4.0 (2011-10)



Universal Mobile Telecommunications System (UMTS); Bearer-independent circuit-switched core network; Stage 2 (3GPP TS 23.205 version 9.4.0 Release 9)



Reference RTS/TSGC-0423205v940

> Keywords UMTS

ETSI

650 Route des Lucioles F-06921 Sophia Antipolis Cedex - FRANCE

Tel.: +33 4 92 94 42 00 Fax: +33 4 93 65 47 16

Siret N° 348 623 562 00017 - NAF 742 C Association à but non lucratif enregistrée à la Sous-Préfecture de Grasse (06) N° 7803/88

Important notice

Individual copies of the present document can be downloaded from: http://www.etsi.org

The present document may be made available in more than one electronic version or in print. In any case of existing or perceived difference in contents between such versions, the reference version is the Portable Document Format (PDF). In case of dispute, the reference shall be the printing on ETSI printers of the PDF version kept on a specific network drive within ETSI Secretariat.

Users of the present document should be aware that the document may be subject to revision or change of status. Information on the current status of this and other ETSI documents is available at http://portal.etsi.org/tb/status/status.asp

If you find errors in the present document, please send your comment to one of the following services: http://portal.etsi.org/chaircor/ETSI_support.asp

Copyright Notification

No part may be reproduced except as authorized by written permission. The copyright and the foregoing restriction extend to reproduction in all media.

> © European Telecommunications Standards Institute 2011. All rights reserved.

DECTTM, PLUGTESTSTM, UMTSTM and the ETSI logo are Trade Marks of ETSI registered for the benefit of its Members. **3GPP**[™] and **LTE**[™] are Trade Marks of ETSI registered for the benefit of its Members and of the 3GPP Organizational Partners.

GSM® and the GSM logo are Trade Marks registered and owned by the GSM Association.

Intellectual Property Rights

IPRs essential or potentially essential to the present document may have been declared to ETSI. The information pertaining to these essential IPRs, if any, is publicly available for **ETSI members and non-members**, and can be found in ETSI SR 000 314: "Intellectual Property Rights (IPRs); Essential, or potentially Essential, IPRs notified to ETSI in respect of ETSI standards", which is available from the ETSI Secretariat. Latest updates are available on the ETSI Web server (http://ipr.etsi.org).

Pursuant to the ETSI IPR Policy, no investigation, including IPR searches, has been carried out by ETSI. No guarantee can be given as to the existence of other IPRs not referenced in ETSI SR 000 314 (or the updates on the ETSI Web server) which are, or may be, or may become, essential to the present document.

Foreword

This Technical Specification (TS) has been produced by ETSI 3rd Generation Partnership Project (3GPP).

The present document may refer to technical specifications or reports using their 3GPP identities, UMTS identities or GSM identities. These should be interpreted as being references to the corresponding ETSI deliverables.

The cross reference between GSM, UMTS, 3GPP and ETSI identities can be found under <u>http://webapp.etsi.org/key/queryform.asp</u>.

Contents

Intelle	ectual Property Rights	2
Forew	/ord	2
Forew	/ord	15
1	Scope	16
2	References	16
3	Symbols and abbreviations	
3.1	Symbols	
3.2	Abbreviations	
3.3	Definitions	19
4	Main Concepts	19
4.1	General	
4.2	Bearer-Independent Call Control	
4.3	H.248/MEGACO	20
5	General Circuit Switched Core Network Domain Architecture	
5.1	Logical Architecture	
5.1.1	CS Core Network Nodes	20
5.1.1.1	MSC Server	
5.1.1.2	C GMSC Server	20
5.1.1.3	B Media Gateway	20
5.1.2	CS Core Network Interfaces and Reference Points	21
5.1.2.1	Mc Interface	21
5.1.2.2	Nc Interface	21
5.1.2.3	Nb Interface	21
5.2	Network Interworking	21
5.2.1	Interworking on the Nc Reference Point	21
5.2.2	Interworking on the Nb Reference Point	21
6	Call Establishment	
6.1	Basic Mobile Originating Call	
6.1.1	Forward bearer establishment	
6.1.1.1	MGW selection	21
6.1.1.2	2 Initial addressing	22
6.1.1.3	8 Network side bearer establishment	22
6.1.1.4	Access bearer assignment	22
6.1.1.5	Framing protocol initialisation	22
6.1.1.6	6 Confirmation of bearer establishment	22
6.1.1.7	Through-Connection	23
6.1.1.8	Interworking function	23
6.1.1.9	Codec handling	
6.1.1.1	0 Voice Processing function	
0.1.1.1	Failure handling in MSC server	
0.1.1.1	2 Example	
6121	MGW selection	
6122	No w selection	20 26
6123	Initial addressing	20 ?6
6.1 2.3	Access bearer assignment	
6.1.2.5	Framing protocol initialisation	
6.1.2.6	5 Confirmation of bearer establishment	
6.1.2.7	Through-Connection	
6.1.2.8	Interworking function	
6.1.2.9	Codec handling	
6.1.2.1	0 Voice Processing function	

6.1.2.11	Failure handling in MSC server	28
6.1.2.12	Example	28
6.1.3	Originating Call Establishment For Iu Interface on IP	30
6.1.4	Forward bearer establishment with Trace Session Activation	31
6.1.4.1	Trace activation	31
6.1.4.2	Example	32
6.1.5	Originating Call Establishment for A interface over IP	34
6.2	Basic Mobile Terminating Call	35
621	Forward bearer establishment	35
6211	GMSC server	35
62111	MGW selection	35
62112	Initial addressing	36
62113	Outgoing side boarer establishment	
0.2.1.1.3	Uugoing side beater establishment	
0.2.1.1.4	Energine and to be in the light of the state	
0.2.1.1.5	Framing protocol initialisation	30
0.2.1.1.0	I nrougn-Connection	30
6.2.1.1./	Confirmation of bearer establishment	36
6.2.1.1.8	Voice Processing function	37
6.2.1.1.9	Failure handling in GMSC server	37
6.2.1.2	MSC server	37
6.2.1.2.1	Paging	37
6.2.1.2.2	Call setup	37
6.2.1.2.3	MGW selection	37
6.2.1.2.4	Network side bearer establishment	37
6.2.1.2.5	Access bearer assignment	37
6.2.1.2.6	Framing protocol initialisation	38
6.2.1.2.7	Called party alerting	38
6.2.1.2.8	Called party answer	38
6.2.1.2.9	Through-Connection	38
621210	Interworking function	39
621211	Codec handling	39
621212	Voice Processing function	39
621212	Failure handling in MSC server	30
621214	Fyampla	30
622	Recleverd hours establishment	42
6221		42
0.2.2.1	MGW selection	42
0.2.2.1.1	MOW selection.	42
0.2.2.1.2		42
6.2.2.1.3	Initial addressing	42
6.2.2.1.4	Incoming side bearer establishment	42
6.2.2.1.5	Framing protocol initialisation	42
6.2.2.1.6	Through-Connection	43
6.2.2.1.7	Confirmation of bearer establishment	43
6.2.2.1.8	Voice Processing function	43
6.2.2.1.9	Failure handling in GMSC server	43
6.2.2.2	MSC server	43
6.2.2.2.1	Paging	43
6.2.2.2.2	Call setup	43
6.2.2.2.3	MGW selection	44
6.2.2.2.3	Network side bearer establishment	44
6.2.2.2.4	Access bearer assignment	44
6.2.2.2.5	Framing protocol initialisation	44
6.2.2.2.6	Called party alerting	45
6.2.2.2.7	Called party answer	45
6.2.2.2.8	Through-Connection	45
6.2.2.2.9	Interworking function	45
6.2.2.2.10	Codec handling	45
6.2.2.2.11	Voice Processing function	45
6.2.2.2.12	Failure handling in MSC server	.45
622213	Example	45
623	Terminating Call Establishment For In Interface on IP	<u>4</u> 0
624	Terminating Call Establishment For A Interface over IP	. . 51
J		

	Call Clearing	51
7.1	Network Initiated	
7.1.1	GMSC server	
7.1.1.1	Call clearing from the incoming side	
7.1.1.2	Call clearing to the outgoing side	
7.1.2	MSC server	
7.1.2.1	Call clearing from the network side	
7.1.2.2	Call clearing to the UE	
7.1.2.3	Example	53
7.2	User Initiated	54
7.2.1	Void	55
7.2.2	MSC server	55
7.2.2.1	all clearing from the UE	55
7.2.2.2	Call clearing to the network side	55
7.2.2.3	Example	55
7.3	(G)MSC server Initiated	56
7.3.1	GMSC server	56
7.3.1.1	Call clearing to the destination side	56
7.3.1.2	Call clearing to the originating side	56
7.3.2	MSC server	57
7.3.2.1	Call clearing to the UE	57
7.3.2.2	Call clearing to the network side	57
7.3.2.3	Example	57
7.4	MGW Initiated	59
7.4.1	GMSC server	59
7.4.1.1	Bearer released on the destination side	59
7.4.1.2	Bearer released on the originating side	59
7.4.2	MSC server	59
7.4.2.1	Bearer released on the access side	59
7.4.2.2	Bearer released on the network side	
7.4.2.3	Example	
7.5	Call Clearing for Iu Interface on IP	61
8	Handover/Relocation	62
0.1		
8.1	UMTS to UMTS	63
8.1 8.1.1	UMTS to UMTS Intra-MSC SRNS/SBSS Relocation	
8.1 8.1.1 8.1.1.1	UMTS to UMTS Intra-MSC SRNS/SBSS Relocation Relocation Required	
8.1 8.1.1 8.1.1.1 8.1.1.2	UMTS to UMTS Intra-MSC SRNS/SBSS Relocation Relocation Required Relocation Command/Relocation Detect	
8.1 8.1.1 8.1.1.1 8.1.1.2 8.1.1.3	UMTS to UMTS Intra-MSC SRNS/SBSS Relocation Relocation Required Relocation Command/Relocation Detect Relocation Complete	
8.1 8.1.1 8.1.1.1 8.1.1.2 8.1.1.3 8.1.1.4	UMTS to UMTS Intra-MSC SRNS/SBSS Relocation Relocation Required Relocation Command/Relocation Detect Relocation Complete Interworking function.	
8.1 8.1.1 8.1.1.1 8.1.1.2 8.1.1.3 8.1.1.4 8.1.1.5	UMTS to UMTS Intra-MSC SRNS/SBSS Relocation Relocation Required Relocation Command/Relocation Detect Relocation Complete Interworking function Codec handling	
8.1 8.1.1 8.1.1.1 8.1.1.2 8.1.1.3 8.1.1.4 8.1.1.5 8.1.1.6	UMTS to UMTS Intra-MSC SRNS/SBSS Relocation Relocation Required Relocation Command/Relocation Detect Relocation Complete Interworking function Codec handling Voice Processing function	
8.1 8.1.1 8.1.1.1 8.1.1.2 8.1.1.3 8.1.1.4 8.1.1.5 8.1.1.6 8.1.1.7	UMTS to UMTS. Intra-MSC SRNS/SBSS Relocation Relocation Required Relocation Command/Relocation Detect Relocation Complete Interworking function. Codec handling Voice Processing function. Handling of multiple bearers (multicall).	
8.1 8.1.1 8.1.1.2 8.1.1.3 8.1.1.4 8.1.1.5 8.1.1.6 8.1.1.7 8.1.1.8	UMTS to UMTS. Intra-MSC SRNS/SBSS Relocation Relocation Required. Relocation Command/Relocation Detect Relocation Complete. Interworking function. Codec handling Voice Processing function. Handling of multiple bearers (multicall). Failure Handling in MSC server	
8.1 8.1.1 8.1.1.1 8.1.1.2 8.1.1.3 8.1.1.4 8.1.1.5 8.1.1.6 8.1.1.7 8.1.1.8 8.1.1.9	UMTS to UMTS. Intra-MSC SRNS/SBSS Relocation Relocation Required. Relocation Command/Relocation Detect	
8.1 8.1.1 8.1.1.1 8.1.1.2 8.1.1.3 8.1.1.4 8.1.1.5 8.1.1.6 8.1.1.7 8.1.1.8 8.1.1.9 8.1.2	UMTS to UMTS. Intra-MSC SRNS/SBSS Relocation Relocation Required. Relocation Command/Relocation Detect	
8.1 8.1.1 8.1.1.1 8.1.1.2 8.1.1.3 8.1.1.4 8.1.1.5 8.1.1.6 8.1.1.7 8.1.1.8 8.1.1.9 8.1.2 8.1.2.1	UMTS to UMTS. Intra-MSC SRNS/SBSS Relocation Relocation Required. Relocation Command/Relocation Detect Relocation Complete. Interworking function. Codec handling Voice Processing function. Handling of multiple bearers (multicall). Failure Handling in MSC server Example. Basic Inter-MSC SRNS/SBSS Relocation. MSC-A/MGW-A	
8.1 8.1.1 8.1.1.1 8.1.1.2 8.1.1.3 8.1.1.4 8.1.1.5 8.1.1.6 8.1.1.7 8.1.1.8 8.1.1.9 8.1.2 8.1.2.1 8.1.2.1	UMTS to UMTS. Intra-MSC SRNS/SBSS Relocation Relocation Required. Relocation Command/Relocation Detect Relocation Complete. Interworking function. Codec handling Voice Processing function. Handling of multiple bearers (multicall). Failure Handling in MSC server Example. Basic Inter-MSC SRNS/SBSS Relocation. MSC-A/MGW-A .1 Relocation Required.	
8.1 8.1.1 8.1.1.1 8.1.1.2 8.1.1.3 8.1.1.4 8.1.1.5 8.1.1.6 8.1.1.7 8.1.1.8 8.1.1.9 8.1.2 8.1.2.1 8.1.2.1 8.1.2.1	UMTS to UMTS. Intra-MSC SRNS/SBSS Relocation Relocation Required Relocation Command/Relocation Detect Relocation Complete Interworking function. Codec handling Voice Processing function. Handling of multiple bearers (multicall) Failure Handling in MSC server Example Basic Inter-MSC SRNS/SBSS Relocation MSC-A/MGW-A 1 Relocation Required 2 Bearer establishment between MGW-A and MGW-B	
8.1 8.1.1 8.1.1.1 8.1.1.2 8.1.1.3 8.1.1.4 8.1.1.5 8.1.1.6 8.1.1.7 8.1.1.8 8.1.1.9 8.1.2 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1	UMTS to UMTS. Intra-MSC SRNS/SBSS Relocation Relocation Required. Relocation Command/Relocation Detect Relocation Complete Interworking function. Codec handling Voice Processing function. Handling of multiple bearers (multicall). Failure Handling in MSC server Example Basic Inter-MSC SRNS/SBSS Relocation. MSC-A/MGW-A .1 Relocation Required .2 Bearer establishment between MGW-A and MGW-B .3 Relocation Command/Relocation Detect	
8.1 8.1.1 8.1.1.1 8.1.1.2 8.1.1.3 8.1.1.4 8.1.1.5 8.1.1.6 8.1.1.7 8.1.1.8 8.1.1.9 8.1.2 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1	UMTS to UMTS. Intra-MSC SRNS/SBSS Relocation Relocation Required. Relocation Command/Relocation Detect Relocation Complete Interworking function. Codec handling Voice Processing function. Handling of multiple bearers (multicall) Failure Handling in MSC server Example Basic Inter-MSC SRNS/SBSS Relocation MSC-A/MGW-A .1 Relocation Required .2 Bearer establishment between MGW-A and MGW-B .3 Relocation Complete .4 Relocation Complete	
8.1 8.1.1 8.1.1.1 8.1.1.2 8.1.1.3 8.1.1.4 8.1.1.5 8.1.1.6 8.1.1.7 8.1.1.8 8.1.1.9 8.1.2 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1	UMTS to UMTS. Intra-MSC SRNS/SBSS Relocation Relocation Required. Relocation Command/Relocation Detect Relocation Complete Interworking function. Codec handling Voice Processing function. Handling of multiple bearers (multicall) Failure Handling in MSC server Example Basic Inter-MSC SRNS/SBSS Relocation. MSC-A/MGW-A .1 Relocation Required .2 Bearer establishment between MGW-A and MGW-B .3 Relocation Complete .4 .5 Interworking function	
8.1 8.1.1 8.1.1.1 8.1.1.2 8.1.1.3 8.1.1.4 8.1.1.5 8.1.1.6 8.1.1.7 8.1.1.8 8.1.1.9 8.1.2 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1	UMTS to UMTS. Intra-MSC SRNS/SBSS Relocation Relocation Required. Relocation Command/Relocation Detect Relocation Complete Interworking function. Codec handling Voice Processing function. Handling of multiple bearers (multicall) Failure Handling in MSC server Example Basic Inter-MSC SRNS/SBSS Relocation. MSC-A/MGW-A 1 Relocation Required .2 Bearer establishment between MGW-A and MGW-B .3 Relocation Complete .4 Relocation Complete .5 Interworking function .6 Codec handling.	
8.1 8.1.1 8.1.1.1 8.1.1.2 8.1.1.3 8.1.1.4 8.1.1.5 8.1.1.6 8.1.1.7 8.1.1.8 8.1.1.9 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1	UMTS to UMTS. Intra-MSC SRNS/SBSS Relocation Relocation Required Relocation Complete Interworking function. Codec handling Voice Processing function. Handling of multiple bearers (multicall) Failure Handling in MSC server Example Basic Inter-MSC SRNS/SBSS Relocation. MSC-A/MGW-A 1 Relocation Command/Relocation Detect .2 Bearer establishment between MGW-A and MGW-B .3 Relocation Command/Relocation Detect .4 Relocation Command/Relocation Detect .5 Interworking function .6 Codec handling .7 Voice Processing function .8 Handling of multiple bearers (multicall)	
8.1 8.1.1 8.1.1.1 8.1.1.2 8.1.1.3 8.1.1.4 8.1.1.5 8.1.1.6 8.1.1.7 8.1.1.8 8.1.1.9 8.1.2.1	UMTS to UMTS. Intra-MSC SRNS/SBSS Relocation Relocation Required. Relocation Command/Relocation Detect Relocation Complete Interworking function. Codec handling Voice Processing function. Handling of multiple bearers (multicall) Failure Handling in MSC server Example Basic Inter-MSC SRNS/SBSS Relocation. MSC-A/MGW-A .1 Relocation Required .2 Bearer establishment between MGW-A and MGW-B .3 Relocation Command/Relocation Detect .4 Relocation Complete .5 Interworking function .6 Codec handling .7 Voice Processing function .8 Handling of multiple bearers (multicall)	
8.1 8.1.1 8.1.1.1 8.1.1.2 8.1.1.3 8.1.1.4 8.1.1.5 8.1.1.6 8.1.1.7 8.1.1.6 8.1.1.7 8.1.2.1	UMTS to UMTS. Intra-MSC SRNS/SBSS Relocation Relocation Required Relocation Command/Relocation Detect Relocation Complete Interworking function Codec handling Voice Processing function Handling of multiple bearers (multicall) Failure Handling in MSC server Example Basic Inter-MSC SRNS/SBSS Relocation MSC-A/MGW-A 1 Relocation Command/Relocation Detect .4 Relocation Command/Relocation Detect .4 .7 Voice Processing function .8 .9 Failure Handling in MSC server </td <td></td>	
8.1 8.1.1 8.1.1.1 8.1.1.2 8.1.1.3 8.1.1.4 8.1.1.5 8.1.1.5 8.1.1.6 8.1.1.7 8.1.1.8 8.1.1.9 8.1.2.1	UMTS to UMTS. Intra-MSC SRNS/SBSS Relocation Relocation Required Relocation Command/Relocation Detect Relocation Complete Interworking function Codec handling Voice Processing function Handling of multiple bearers (multicall) Failure Handling in MSC server Example Basic Inter-MSC SRNS/SBSS Relocation MSC-A/MGW-A 1 Relocation Complete .3 Relocation Complete .4 .5 Interworking function .6 .7 Voice Processing function .8 Handling of multiple bearers (multicall) .9 Failure Handling in MSC server	
8.1 8.1.1 8.1.1.1 8.1.1.2 8.1.1.3 8.1.1.4 8.1.1.5 8.1.1.6 8.1.1.7 8.1.1.8 8.1.1.9 8.1.2 8.1.2.1	UMTS to UMTS. Intra-MSC SRNS/SBSS Relocation Relocation Required. Relocation Command/Relocation Detect Relocation Complete Interworking function. Codec handling Voice Processing function. Handling of multiple bearers (multicall) Failure Handling in MSC server Example Basic Inter-MSC SRNS/SBSS Relocation. MSC-A/MGW-A 1 Relocation Required 2 Bearer establishment between MGW-A and MGW-B 3 Relocation Complete .5 Interworking function. .6 Codec handling. .7 Voice Processing function. .6 Codec handling. .7 Voice Processing function. .8 Handling of multiple bearers (multicall) .9 Failure Handling in MSC server. MSC-B/MGW-B	
8.1 8.1.1 8.1.1.1 8.1.1.2 8.1.1.3 8.1.1.4 8.1.1.5 8.1.1.6 8.1.1.7 8.1.1.8 8.1.1.9 8.1.2 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.2 8.1.2.	UMTS to UMTS. Intra-MSC SRNS/SBSS Relocation Relocation Required. Relocation Command/Relocation Detect Relocation Complete. Interworking function. Codec handling Voice Processing function. Handling of multiple bearers (multicall) Failure Handling in MSC server Example Basic Inter-MSC SRNS/SBSS Relocation. MSC-A/MGW-A 1 Relocation Complete .3 Relocation Command/Relocation Detect .4 Relocation Command/Relocation Detect .4 Relocation Command/Relocation Detect .5 Interworking function .6 Codec handling .7 Voice Processing function .6 Codec handling .7 Voice Processing function .8 Handling of multiple bearers (multicall) .9 Failure Handling in MSC server. .1 MGW selection .2 Bearer establishment towards RNC-B. .3 Bearer establishment towards RNC-B. .3 Bearer establishment towards RNC-B. .4 MGW sele	
8.1 8.1.1 8.1.1.1 8.1.1.2 8.1.1.3 8.1.1.4 8.1.1.5 8.1.1.6 8.1.1.7 8.1.1.8 8.1.1.9 8.1.2 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.2 8.1.2.	UMTS to UMTS. Intra-MSC SRNS/SBSS Relocation Relocation Required. Relocation Command/Relocation Detect Relocation Complete Interworking function. Codec handling Voice Processing function. Handling of multiple bearers (multicall). Failure Handling in MSC server Example Basic Inter-MSC SRNS/SBSS Relocation. MSC-A/MGW-A 1 Relocation Complete .3 Relocation Complete .5 Interworking function. .6 Codec handling. .7 Voice Processing function .6 Codec handling. .7 Voice Processing function .6 Codec handling. .7 Voice Processing function .6 Codec handling in MSC server. MSC-B/MGW-B. MGW selection. .1 MGW selection. .2 Bearer establishment towards RNC-B. .3 Bearer establishment between MGW-A and MGW-B.	
8.1 8.1.1 8.1.1.1 8.1.1.2 8.1.1.3 8.1.1.4 8.1.1.5 8.1.1.6 8.1.1.7 8.1.1.8 8.1.1.9 8.1.2 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.1 8.1.2.2 8.1.2.	UMTS to UMTS. Intra-MSC SRNS/SBSS Relocation Relocation Required. Relocation Command/Relocation Detect. Relocation Complete Interworking function. Codec handling Voice Processing function. Handling of multiple bearers (multicall) Failure Handling in MSC server Example Basic Inter-MSC SRNS/SBSS Relocation MSC-A/MGW-A 1 Relocation Command/Relocation Detect .4 Relocation Command/Relocation Detect .5 Inter-MSC SRNS/SBSS Relocation MSC-A/MGW-A .1 Relocation Required .2 Bearer establishment between MGW-A and MGW-B .3 Relocation Commend/Relocation Detect .4 Relocation Complete .5 Interworking function .6 Codec handling .7 Voice Processing function .8 Handling of multiple bearers (multicall) .9 Failure Handling in MSC server .1 MGW selection .2 Bearer establishment towards RNC-B	

0100c	Handling of multiple because (multipli)	(0
8.1.2.2.0	Handing of multiple bearers (multical)	
8.1.2.2.7	Failure Handling in MSC server	68
8.1.2.2.8	Example	69
8.1.3	Subsequent Inter-MSC SRNS/SBSS Relocation back to the Anchor MSC	71
8.1.3.1	MSC-A/MGW-A	71
8.1.3.1.1	Relocation Request	71
8.1.3.1.2	Relocation Command/Relocation Detect	71
81313	Relocation Complete	72
81314	Interworking function	
0.1.3.1.4		
8.1.3.1.5	Codec nandling	
8.1.3.1.6	Voice Processing function	
8.1.3.1.7	Handling of multiple bearers (multicall)	72
8.1.3.1.8	Failure Handling in MSC server	72
8.1.3.2	MSC-B/MGW-B	72
8.1.3.2.1	Relocation Required	72
8.1.3.2.2	Relocation Complete	72
8.1.3.2.3	Release of bearer towards MGW-A	
81324	Handling of multiple bearers (multicall)	73
81325	Fyample	73
0.1. <i>3.2.3</i> 9.1. <i>1</i>	Subsequent Inter MSC SDNS/SDSS Delegation to a third MSC	
0.1.4	Subsequent Inter-MISC SKNS/SDSS Relocation to a tintu MISC	
8.1.5	SKINS/SDSS Relocation with the on the second s	
ð.1.0		
8.1.6.1		
8.1.6.2	Enhanced Relocation Complete Request	
8.1.6.3	Enhanced Relocation Complete Confirm	77
8.1.6.4	Interworking function	77
8.1.6.5	Codec handling	77
8.1.6.6	Voice Processing function	77
8.1.6.7	Handling of multiple bearers (multicall)	77
8.1.6.8	Failure Handling in MSC server	77
8.1.6.9	Example	77
8.1.7	Intra-MSC enhanced SRNS Relocation with Iu on IP	79
8.2	UMTS to GSM	80
8.2.1	Intra-MSC UMTS to GSM Handover	80
8211	Relocation Required	80
8212	Handover Request Acknowledge	00 01
8213	Relocation Command/Handover Detect	
8214	Relocation Command, Handover Detect.	
0.2.1.4	Handovar Complete	
8.2.1.5	Handover Complete	
0010	Handover Complete Interworking function	
8.2.1.6	Handover Complete Interworking function Voice Processing function	81 81 81 81 81 81 81
8.2.1.6 8.2.1.7	Handover Complete Interworking function Voice Processing function Handling of multiple bearers (multicall)	81 81 81 81 81 81 81
8.2.1.6 8.2.1.7 8.2.1.8	Handover Complete Interworking function Voice Processing function Handling of multiple bearers (multicall) Failure Handling in MSC server	81 81 81 81 81 81 81 81 81
8.2.1.6 8.2.1.7 8.2.1.8 8.2.1.9	Handover Complete Interworking function Voice Processing function Handling of multiple bearers (multicall) Failure Handling in MSC server Example	81 81 81 81 81 81 81 81 81 81
8.2.1.6 8.2.1.7 8.2.1.8 8.2.1.9 8.2.1.10	Handover Complete Interworking function Voice Processing function Handling of multiple bearers (multicall) Failure Handling in MSC server Example Intra-MSC UMTS to GSM Handover for A interface over IP	81 81 81 81 81 81 81 81 81 84
8.2.1.6 8.2.1.7 8.2.1.8 8.2.1.9 8.2.1.10 8.2.2	Handover Complete Interworking function Voice Processing function Handling of multiple bearers (multicall) Failure Handling in MSC server Example Intra-MSC UMTS to GSM Handover for A interface over IP Basic Inter-MSC UMTS to GSM Handover	81 81 81 81 81 81 81 81 81 84 85
8.2.1.6 8.2.1.7 8.2.1.8 8.2.1.9 8.2.1.10 8.2.2 8.2.2.1	Handover Complete Interworking function Voice Processing function Handling of multiple bearers (multicall) Failure Handling in MSC server Example Intra-MSC UMTS to GSM Handover for A interface over IP Basic Inter-MSC UMTS to GSM Handover MSC-A/ MGW-A	81 81 81 81 81 81 81 81 81 84 85 85
8.2.1.6 8.2.1.7 8.2.1.8 8.2.1.9 8.2.1.10 8.2.2 8.2.2.1 8.2.2.1 8.2.2.1.1	Handover Complete Interworking function Voice Processing function Handling of multiple bearers (multicall) Failure Handling in MSC server Example Intra-MSC UMTS to GSM Handover for A interface over IP Basic Inter-MSC UMTS to GSM Handover MSC-A/ MGW-A Bearer establishment between MGW-A and MGW-B	81 81 81 81 81 81 81 81 81 81 84 85 85 85 85
8.2.1.6 8.2.1.7 8.2.1.8 8.2.1.9 8.2.1.10 8.2.2 8.2.2.1 8.2.2.1.1 8.2.2.1.1 8.2.2.1.2	Handover Complete Interworking function Voice Processing function Handling of multiple bearers (multicall) Failure Handling in MSC server Example Intra-MSC UMTS to GSM Handover for A interface over IP Basic Inter-MSC UMTS to GSM Handover MSC-A/ MGW-A Bearer establishment between MGW-A and MGW-B Relocation Command/Handover Detect	81 81 81 81 81 81 81 81 81 81 84 85 85 85 85 85 85
8.2.1.6 8.2.1.7 8.2.1.8 8.2.1.9 8.2.1.10 8.2.2 8.2.2.1 8.2.2.1.1 8.2.2.1.1 8.2.2.1.2 8.2.2.1.3	Handover Complete Interworking function Voice Processing function Handling of multiple bearers (multicall) Failure Handling in MSC server Example Intra-MSC UMTS to GSM Handover for A interface over IP Basic Inter-MSC UMTS to GSM Handover MSC-A/ MGW-A Bearer establishment between MGW-A and MGW-B Relocation Command/Handover Detect Handover Complete	81 81 81 81 81 81 81 81 81 81 84 85 85 85 85 85 85 85 85 85 85 85 85 86
8.2.1.6 8.2.1.7 8.2.1.8 8.2.1.9 8.2.1.10 8.2.2 8.2.2.1 8.2.2.1.1 8.2.2.1.1 8.2.2.1.2 8.2.2.1.3 8.2.2.1.4	Handover Complete Interworking function Voice Processing function Handling of multiple bearers (multicall) Failure Handling in MSC server Example Intra-MSC UMTS to GSM Handover for A interface over IP Basic Inter-MSC UMTS to GSM Handover MSC-A/ MGW-A Bearer establishment between MGW-A and MGW-B Relocation Command/Handover Detect Handover Complete Interworking function	81 81 81 81 81 81 81 81 81 81 81 84 85 85 85 85 85 85 85 85 85 85 85 85 85
8.2.1.6 8.2.1.7 8.2.1.8 8.2.1.9 8.2.1.10 8.2.2 8.2.2.1 8.2.2.1.1 8.2.2.1.2 8.2.2.1.3 8.2.2.1.4 8.2.2.1.4	Handover Complete Interworking function Voice Processing function Handling of multiple bearers (multicall) Failure Handling in MSC server Example Intra-MSC UMTS to GSM Handover for A interface over IP Basic Inter-MSC UMTS to GSM Handover MSC-A/ MGW-A Bearer establishment between MGW-A and MGW-B Relocation Command/Handover Detect Handover Complete Interworking function Voice Processing function	81 81 81 81 81 81 81 81 81 81
8.2.1.6 8.2.1.7 8.2.1.8 8.2.1.9 8.2.1.10 8.2.2 8.2.2.1 8.2.2.1.1 8.2.2.1.2 8.2.2.1.2 8.2.2.1.3 8.2.2.1.4 8.2.2.1.5 8.2.2.1.5	Handover Complete Interworking function Voice Processing function Handling of multiple bearers (multicall) Failure Handling in MSC server Example Intra-MSC UMTS to GSM Handover for A interface over IP Basic Inter-MSC UMTS to GSM Handover MSC-A/ MGW-A Bearer establishment between MGW-A and MGW-B Relocation Command/Handover Detect Handover Complete Interworking function Voice Processing function Voice Processing function	81 81 81 81 81 81 81 81 81 81
8.2.1.6 8.2.1.7 8.2.1.8 8.2.1.9 8.2.1.10 8.2.2 8.2.2.1.1 8.2.2.1.1 8.2.2.1.2 8.2.2.1.2 8.2.2.1.3 8.2.2.1.4 8.2.2.1.5 8.2.2.1.6 8.2.2.1.6	Handover Complete Interworking function Voice Processing function Handling of multiple bearers (multicall) Failure Handling in MSC server Example Intra-MSC UMTS to GSM Handover for A interface over IP Basic Inter-MSC UMTS to GSM Handover MSC-A/ MGW-A Bearer establishment between MGW-A and MGW-B Relocation Command/Handover Detect Handover Complete Interworking function Voice Processing function Voice Processing function Handling of multiple bearers (multicall) Failure Handling in MSC server	81 81 81 81 81 81 81 81 81 81 81 84 85 85 85 85 85 85 85 85 85 85 85 86 86 86 86 86 86 86 86 86 86
8.2.1.6 8.2.1.7 8.2.1.8 8.2.1.9 8.2.1.10 8.2.2 8.2.2.1.1 8.2.2.1.1 8.2.2.1.2 8.2.2.1.2 8.2.2.1.3 8.2.2.1.4 8.2.2.1.5 8.2.2.1.6 8.2.2.1.7 8.2.2.1.7	Handover Complete Interworking function Voice Processing function Handling of multiple bearers (multicall) Failure Handling in MSC server Example Intra-MSC UMTS to GSM Handover for A interface over IP Basic Inter-MSC UMTS to GSM Handover MSC-A/ MGW-A Bearer establishment between MGW-A and MGW-B Relocation Command/Handover Detect Handover Complete Interworking function Voice Processing function Voice Processing function Handling of multiple bearers (multicall) Failure Handling in MSC server	81 81 81 81 81 81 81 81 81 81
8.2.1.6 8.2.1.7 8.2.1.8 8.2.1.9 8.2.1.10 8.2.2 8.2.2.1 8.2.2.1.1 8.2.2.1.2 8.2.2.1.3 8.2.2.1.4 8.2.2.1.5 8.2.2.1.6 8.2.2.1.7 8.2.2.1.7 8.2.2.2	Handover Complete Interworking function Voice Processing function Handling of multiple bearers (multicall) Failure Handling in MSC server Example Intra-MSC UMTS to GSM Handover for A interface over IP Basic Inter-MSC UMTS to GSM Handover MSC-A/ MGW-A Bearer establishment between MGW-A and MGW-B Relocation Command/Handover Detect Handover Complete Interworking function Voice Processing function Voice Processing function Handling of multiple bearers (multicall) Failure Handling in MSC server MSC-B/MGW-B MGW selection	81 81 81 81 81 81 81 81 81 81
8.2.1.6 8.2.1.7 8.2.1.8 8.2.1.9 8.2.1.10 8.2.2 8.2.2.1.1 8.2.2.1.1 8.2.2.1.2 8.2.2.1.3 8.2.2.1.3 8.2.2.1.4 8.2.2.1.5 8.2.2.1.6 8.2.2.1.7 8.2.2.1.7 8.2.2.2 8.2.2.2.1 8.2.2.2.1	Handover Complete	81 81 81 81 81 81 81 81 81 81
8.2.1.6 8.2.1.7 8.2.1.8 8.2.1.9 8.2.1.10 8.2.2 8.2.2.1.1 8.2.2.1.1 8.2.2.1.2 8.2.2.1.3 8.2.2.1.3 8.2.2.1.4 8.2.2.1.5 8.2.2.1.6 8.2.2.1.7 8.2.2.2 8.2.2.2.1 8.2.2.2.1 8.2.2.2.1 8.2.2.2.2 8.2.2.2.1	Handover Complete Interworking function	81 81 81 81 81 81 81 81 81 81
8.2.1.6 8.2.1.7 8.2.1.8 8.2.1.9 8.2.1.10 8.2.2 8.2.2.1.1 8.2.2.1.1 8.2.2.1.2 8.2.2.1.3 8.2.2.1.3 8.2.2.1.4 8.2.2.1.5 8.2.2.1.6 8.2.2.1.7 8.2.2.2 8.2.2.2.1 8.2.2.2.2 8.2.2.2.1 8.2.2.2.2 8.2.2.2.2 8.2.2.2.2 8.2.2.2.2	Handover Complete	81 81 81 81 81 81 81 81 81 81
$\begin{array}{c} 8.2.1.6\\ 8.2.1.7\\ 8.2.1.8\\ 8.2.1.9\\ 8.2.1.0\\ 8.2.2\\ 8.2.2.1\\ 8.2.2.1.1\\ 8.2.2.1.2\\ 8.2.2.1.3\\ 8.2.2.1.3\\ 8.2.2.1.4\\ 8.2.2.1.5\\ 8.2.2.1.6\\ 8.2.2.1.7\\ 8.2.2.2\\ 8.2.2.2.1\\ 8.2.2.2.1\\ 8.2.2.2.2\\ 8.2.2.2.1\\ 8.2.2.2.3\\ 8.2.2.2.4\\ 8.2.2.2.4\\ 8.2.2.5\\ 8.2.5\\$	Handover Complete Interworking function	81 81 81 81 81 81 81 81 81 81 81 81 81 8
$\begin{array}{c} 8.2.1.6\\ 8.2.1.7\\ 8.2.1.8\\ 8.2.1.9\\ 8.2.1.0\\ 8.2.2\\ 8.2.2.1\\ 8.2.2.1.1\\ 8.2.2.1.2\\ 8.2.2.1.3\\ 8.2.2.1.3\\ 8.2.2.1.3\\ 8.2.2.1.4\\ 8.2.2.1.5\\ 8.2.2.1.6\\ 8.2.2.1.7\\ 8.2.2.2\\ 8.2.2.2.1\\ 8.2.2.2.1\\ 8.2.2.2.2\\ 8.2.2.2.3\\ 8.2.2.2.4\\ 8.2.2.2.5\\ 8.2.5\\ 8.$	Handover Complete	81 81 81 81 81 81 81 81 81 81 81 81 81 8
$\begin{array}{c} 8.2.1.6\\ 8.2.1.7\\ 8.2.1.8\\ 8.2.1.9\\ 8.2.1.0\\ 8.2.2\\ 8.2.2.1\\ 8.2.2.1.1\\ 8.2.2.1.2\\ 8.2.2.1.2\\ 8.2.2.1.3\\ 8.2.2.1.3\\ 8.2.2.1.4\\ 8.2.2.1.5\\ 8.2.2.1.6\\ 8.2.2.1.7\\ 8.2.2.2\\ 8.2.2.2.1\\ 8.2.2.2.2\\ 8.2.2.2.1\\ 8.2.2.2.2\\ 8.2.2.2.3\\ 8.2.2.2.4\\ 8.2.2.2.5\\ 8.2.2.2.6\\ 8.2.2.2\\ 8.2.2.2.2\\ 8.2.2.2.2\\ 8.2.2.2.2\\ 8.$	Handover Complete Interworking function	81 81 81 81 81 81 81 81 81 81 81 81 81 8
$\begin{array}{r} 8.2.1.6\\ 8.2.1.7\\ 8.2.1.8\\ 8.2.1.9\\ 8.2.1.10\\ 8.2.2\\ 8.2.2.1\\ 8.2.2.1.1\\ 8.2.2.1.2\\ 8.2.2.1.2\\ 8.2.2.1.3\\ 8.2.2.1.3\\ 8.2.2.1.4\\ 8.2.2.1.5\\ 8.2.2.1.6\\ 8.2.2.1.7\\ 8.2.2.2\\ 8.2.2.2.1\\ 8.2.2.2.2\\ 8.2.2.2.1\\ 8.2.2.2.2\\ 8.2.2.2.3\\ 8.2.2.2.4\\ 8.2.2.2.5\\ 8.2.2.2.6\\ 8.2.2.3\\ 8.2.3$	Handover Complete Interworking function	81 81

8.2.3.1	MSC-A	90
8.2.3.1.1	Handover Request	90
8.2.3.1.2	Handover Request Acknowledge	91
8.2.3.1.3	Relocation Command/Handover Detect	91
8.2.3.1.4	Handover Complete	91
8.2.3.1.5	Interworking function	91
8.2.3.1.6	Voice Processing function	91
8.2.3.1.7	Handling of multiple bearers (multicall)	91
8.2.3.1.8	Failure Handling in MSC server	91
8.2.3.2	MSC-B	91
8.2.3.2.1	Handover Complete	91
8.2.3.2.2	Release of bearer towards MGW-A	91
8.2.3.2.3	Handling of multiple bearers (multicall)	92
8.2.3.2.4	Example	92
8.2.3.3	Subsequent Inter-MSC UMTS to GSM Handover back to the anchor MSC for A Interface	over
011	IP Subsequent later MSC UMTS to CSM Hendever to a third MSC	94
8.2.4 9.2	Subsequent miler-MSC UM15 to GSM Handover to a unit MSC	
0.5	USIM 10 UM15	
0.3.1	Hundover Dogwingd	
0.3.1.1	Handover Command/Palasetian Datast	
0.3.1.2	Palocation Complete	
0.3.1.3	Interworking function	
0.3.1.4 8 3 1 5	Codec handling	90
8316	Voice Processing function	
8317	Failure Handling in MSC server	
8318	Fyample	96. 96
832	Basic Inter-MSC GSM to UMTS Handover	98
8321	MSC-A	90
83211	Handover Required	99
83212	Bearer establishment between MGW-A and MGW-B	99
83213	Handover Command/Handover Detect	99
83214	Handover Complete	99
8.3.2.1.5	Interworking function	
8.3.2.1.6	Codec handling	
8.3.2.1.7	Voice Processing function	
8.3.2.1.8	Failure Handling in MSC server	
8.3.2.2	MSC-B	
8.3.2.2.1	MGW selection	
8.3.2.2.2	Bearer establishment towards RNC-B	
8.3.2.2.3	Bearer establishment between MGW-A and MGW-B	100
8.3.2.2.4	Codec handling	100
8.3.2.2.5	Voice Processing function	100
8.3.2.2.6	Failure Handling in MSC server	100
8.3.2.2.7	Example	101
8.3.3	Subsequent Inter-MSC GSM to UMTS Handover back to the Anchor MSC	103
8.3.3.1	MSC-A	103
8.3.3.1.1	Handover Request	103
8.3.3.1.2	Handover Command/Relocation Detect	104
8.3.3.1.3	Relocation Complete	104
8.3.3.1.4	Interworking function	104
8.3.3.1.5	Codec handling	104
8.3.3.1.6	Voice Processing function	104
8.3.3.1.7	Failure Handling in MSC server	104
8.3.3.2	MSC-B / MGW-B	104
8.3.3.2.1	Handover Required	104
8.3.3.2.2	Handover Complete	105
8.3.3.2.3	Release of bearer towards MGW-A	105
8.3.3.2.4	Example	105
8.3.4	Subsequent Inter-MSC GSM to UMTS Handover to a third MSC	107
8.3.5	GSM to UMTS Handover with Iu on IP	108
8.4	GSM to GSM	109

9/1	Intro MSC Intor DSS CSM to CSM Handovar	100
0.4.1	Hundover Doquired	109
8/12	Handover Request Acknowledge	110
8/13	Handover Command/Handover Detect	110
8/11/	Handover Command/Handover Detect	110
8/115	Interworking function	110
8416	Voice Processing function	110
8/17	Failure Handling in MSC server	110
8/18	Fyample	111
8/10	Intra-MSC GSM to GSM Handover for A interface over IP	113
8/17	Basic Inter-MSC GSM to GSM Handover	114
8421	MSC-A / MGW-A	114
8/21	1 Bearer establishment between MGW-A and MGW-B	11/
8/21	2 Handover Command/Handover Detect	11/
8/21	3 Handover Complete	115
8421	4 Interworking function	115
8/21	5 Voice Processing function	115
8421	6 Failure Handling in MSC server	115
8422	MSC-B / MGW-B	115
8422	1 MGW selection	115
8/22	2 Rearer establishment towards RSC-R	115
8422	3 Bearer establishment between MGW-A and MGW-B	115
8/22	A Voice Processing function	115
8422	5 Failure Handling in MSC server	115
8/22	6 Evample	115
8/23	Basic Inter-MSC GSM to GSM Handover for A Interface over IP	118
8/3	Subsequent Inter MSC GSM to GSM Handover back to the Anchor MSC	110
8/31	MSC-A / MGW-A	110
8/31	1 Handover Request Acknowledge	110
8/31	2 Handover Command/Handover Detect	120
8/31	3 Handover Complete	120
8/31	A Interworking function	120
8/31	5 Voice Processing function	120
8/31	6 Failure Handling in MSC server	120
8/37	MSC B/MGW B	120
8/32	1 Handover Complete	120
8432	2 Release of hearer towards MGW-A	120
8432	3 Fxample	120
8433	Subsequent Inter-MSC GSM to GSM Handover back to the anchor MSC for A Interface over IP	120
844	Subsequent first to GSM Handover to a third MSC	124
845	BSS Internal Handover	125
8451	Introduction	125
8452	Internal Handover Required	125
8453	Internal Handover Command/Handover Detect	125
8454	Handover Complete	125
8455	Interworking function	125
8456	Voice Processing function	125
8457	Failure Handling in MSC server	125
8.4.5.8	Example for a successful BSS Internal Handover	.126
8.5	Handling of GSM Services after UMTS to GSM Handover	.128
0.0		0
9	Compatibility Issues	.128
9.1	Interworking with GERAN (A i/f)	.128
9.1.1	Introduction	.128
9.1.2	A interface over TDM	.129
10	General (G)MSC server_MGW Procedures	120
10 1	MCW Unavailable	129
10.1	MGW Available	120
10.2	MCW Decovery	.130
10.5	GMSC server Recovery	122
10.4	General	122
10.4.1	University of the second s	

10.4.2	(G)MSC Server Restoration.	133
10.5	MGW Re-register	133
10.6	MGW Re-registration Ordered by (G)MSC server	134
10.7	Removal from Service of a Physical Termination	134
10.8	Restoration to Service of a Physical Termination	135
10.9	Audit of MGW	135
10.9.1	Audit of Value	135
10.9.2	Audit of Capability	136
10.10	MGW Capability Change	136
10.11	Void	137
10.12	(G)MSC Server Out of service	137
10.13	MGW Resource Congestion Handling - Activate	137
10.14	MGW Resource Congestion Handling -Indication	137
10.15	Control association monitoring	138
10.16	Hanging termination detection	138
11	Identities	.139
11.1	Bearer Address and Binding Reference	139
11.2	MGW-Id	139
11.3	(G)MSC server Address	139
10		120
12	Operational Aspects	.139
12.1	Charging	139
13	Interactions with Other Services	.139
13.1	Enhanced Multi-Level Precedence and Pre-emption service (eMLPP)	140
13.2	Call Deflection Service	140
13.2.1	MGW selection and incoming side bearer establishment	140
13.2.2	IU/A-interface release	140
13.2.3	Notification to the Calling Subscriber	140
13.2.4	Initial addressing	140
13.2.5	Establishment of bearer towards the forwarded-to subscriber	140
13.2.6	Failure handling in MSC server	140
13.2.7	Example	140
13.3	Line identification Services	142
13.3.1	Calling Line Identification Presentation (CLIP)	142
13.3.2	Calling Line Identification Restriction (CLIR)	142
13.3.3	Connected Line Identification Presentation (COLP)	142
13.3.4	Connected Line Identification Restriction (COLR)	142
13.4	Call Forwarding Services	142
13.4.1	Call Forwarding Unconditional (CFU)	142
13.4.1.	1 MGW selection	142
13.4.1.	2 Incoming side bearer establishment	142
13.4.1.	3 Notification to the Calling Subscriber	142
13.4.1.	4 Initial addressing	142
13.4.1.	5 Establishment of bearer towards the forwarded-to subscriber	143
13.4.1.	6 Confirmation of bearer establishment	143
13.4.1.	7 Failure handling in GMSC server	143
13.4.1.	8 Example	
13.4.2	Call Forwarding on mobile subscriber Busy (CFB)	144
13.4.2	1 Network Determined User Busy (NDUB)	.144
13.4.2	1.1 MGW selection	.144
13.4.2	1.2 Incoming side dearer establishment. 1.3 Notification to the calling subscriber	.144
13.4.2	1.5 Nouncation to the canning subscriber	.144
13.4.2	1.4 IIIIIdi auditessiig	143 1 <i>4</i> 5
13.4.2	1.5 Establishment of bearer establishment	1/5
13.4.2	1.5 Contraction of Dealer establishment	1143
13.4.2	1.0 Failute hallothig in NioC server 1.7 Evample	143 1 <i>4</i> 5
13.4.2	 Drampte	.14J 1/6
13.4.2	2 Ost Determined User Dusy (UDUD)	.140 1/6
13.4.2	2.1 INTO W SELECTION	1/7
13.4.2	2.2 medining side ocale establishinent	1/7
13.4.2.	2.5 IU/A-IIIU/1400 IU/480	+/

13.4.2.2.4	Notification to the Calling Subscriber	
13.4.2.2.5	Initial addressing	
13.4.2.2.6	Establishment of bearer towards the forwarded-to subscriber	147
13.4.2.2.7	Confirmation of bearer establishment	
13.4.2.2.8	Failure handling in MSC server	
13.4.2.2.9	Example	
13.4.3	Call Forwarding on No Reply (CFNRy)	
13.4.3.1	MGW selection and incoming side bearer establishment	
13.4.3.2	IU/A-interface release	
13.4.3.3	Notification to the Calling Subscriber	
13.4.3.4	Initial addressing	
13.4.3.5	Establishment of bearer towards the forwarded-to subscriber	
13.4.3.6	Failure handling in MSC server	150
13.4.3.7	Example	150
13.4.4	Call Forwarding on mobile subscriber Not Reachable (CFNRc).	
13441	Rerouting by HLR	151
13442	Rerouting by VLR	151
13.4.4.2.1	MGW selection	152
13.4.4.2.2	Incoming side bearer establishment	152
134423	Notification to the calling subscriber	152
134424	Initial addressing	152
13 4 4 2 5	Establishment of bearer towards the forwarded-to subscriber	152
134426	Confirmation of bearer establishment	153
13.4.4.2.0	Failure handling in MSC server	
13.4.4.2.7	Fyamle	
13.4.4.2.0	Call Waiting (CW)	
13.5	Call Confirmation of the waiting call	
13.5.1	Acceptance of the Waiting Call	130
13.5.2	Call Hold (CH)	
13.61	Hold request	
13.0.1	Retrieval request	
13.0.2	Setting up another call	
13.0.5	Alternate from one call to the other	130
13.0.4	Disconnect	130
13.0.5	Eailura handling in MSC sorver	
13.0.0	Francisco nationing in MiSC Sciver	
13.0.7	Example 1	130
13.0.8	Multiparty (MPTV)	
13.7	Beginning the Multi Derty coll	104
13.7.1	Managing an active Multi Party call	104
13.7.2	Disconnect	104
13.7.5	Englura handling in MSC sorver	105
1275	Fanule handling in Misc Server	
1276	Example 2 — Private call establishment during multiparty call	105
12.7.0	Closed User Group (CUG)	100
13.0	Advise of Charge (AoC)	
13.9 12.10	Auvice of Charge (AOC)	
13.10	Coll Derring Conviges	
13.11	Daming of outgoing calls	
12.11.1	Darring of outgoing calls	
13.11.2	Darning of incoming cans	
13.12	Connection of remote portion	
13.12.1	UI/A interface release	
13.12.2	TU/A-IIICITACE TETERSE	
13.12.3	Famule mandling in MISC server	
13.12.4	Example	
13.13	Completion of Calls to Busy Subscriber (CCBS)	
13.13.1	Uterring when tones/announcements are provided to the calling subscriber	
13.13.2	Network initiated mobile originated call	
13.13.2.1	Early Traffic Channel Assignment	
13.13.2.1.	I Example	
13.13.3	CCBS Information conveyed by Call Signalling	
13.14	wumple Subscriber Profile (MSP)	1/4

10.15	N 12 11	174
13.15	Multical	.1/4
13.15.1	Mobile Originating	.1/4
13.15.2	Mobile Terminating	1/4
13.10	Calling Name Presentation (CNAP)	.1/4
13.17	Alternate Speech/Fax	.1/4
13.18	Modification of Decrea Characteristics	.1/5
13.18.1	Modification of Bearer Characteristics	1/5
13.18.1.1	IU mode	1/5
13.18.1.2	A/GD mode	1/5
13.18.2	IWF PTOIOCOI Change	1/5
13.19	GSM Fax	.1/5
13.20	Voice group call service (VGCS), voice broadcast service (VBS)	.170
13.20.1	Tallan above in Value and all (1 above land)	.170
13.20.2	Talker change in voice group call (1 channel model)	1/0
13.20.3	Disconnect	1//
13.20.4	Failure nandling in MISC server	1//
15.20.5	Example	1//
14 Int	eractions with Other Network Features and Services	181
14.1	Customised Applications for Mobile network Enhanced Logic (CAMEL)	181
14.1.1	Play Announcement/Send Tone	182
14.1.1.1	Example of playing announcement by the gsmSRF	182
14.1.2	User Interaction	182
14.1.2.1	Play announcement	182
14.1.2.2	Detect DTMF tones	
14.1.2.3	Report DTMF tones	183
14.1.2.4	Cancel prompt and collect user information	183
14.1.3	Call Party Handling (CPH)	183
14.1.3.1	Call Party Handling concepts	184
14.1.3.2	Initiate Call Attempt procedure	184
14.1.3.2.1	Example	
14.1.3.3	Move Leg procedure	185
14.1.3.4	Split Leg procedure	186
14.1.3.5	CAMEL User interaction procedure	187
14.1.3.6	Failure handling in the MSC-S	188
14.2	IST	188
14.3	Operator Determined Barring (ODB)	188
14.3.1	Barring of Outgoing Calls	188
14.3.2	Barring of Incoming Calls	188
14.4	DTMF	189
14.4.1	DTMF Tone Generation	189
14.4.1.1	Inband DTMF Tone Generation	189
14.4.1.1.1	Start DTMF	189
14.4.1.1.2	Stop DTMF	189
14.4.1.1.3	Example	189
14.4.1.2	Out-of-Band DTMF Tone Generation	190
14.4.1.2.1	Start DTMF	190
14.4.1.2.2	Stop DTMF	190
14.4.1.2.3	Example	190
14.4.2	DTMF Detection	191
14.4.2.1	Inband DTMF Detection	191
14.4.2.2	Out-of-Band DTMF Detection	192
14.5	OR	192
14.5.1	Optimal routeing for basic mobile-to-mobile calls	192
14.5.2	Optimal routeing for conditional call forwarding; Early call forwarding	192
14.5.3	Optimal routeing for conditional call forwarding: Late call forwarding.	192
14.5.3.1	MSC server	192
14.5.3.1.1	Resume Call Handling and clearing of connection to GMSC server	192
14.5.3.1.2	IU release	192
14.5.3.2	GMSC server	193
14.5.3.2.1	Resume Call Handling and Clearing of Connection to visited MSC server	193
14.5.3.2.2	MGW selection	193

14.5.3.2.3	Incoming side bearer establishment	
145324	Notification to the Calling Subscriber	193
14.5.3.2.5	Establishment of call and bearer towards the forwarded-to subscriber	
145326	Example	193
14.6 Pro	viding tones or announcements	197
14.6.1	Preconditions when providing in-band information to the calling subscriber	197
14.6.2	Preconditions when providing in-band information to the called subscriber	197
1463	Preconditions when providing in-band information to use called subscribers	197
14.6.5	Request to play an appouncement/tone	197
1465	Stopping an appouncement/tone	197
14.6.6	Announcement/tone completed	198
14.6.7	Fxample 1	198
14.6.8	Example 7	199
14.0.0 14.7 Gld	hal Text Telenhony	200
14.7 GR	ergency Calls	201
14.0 En	scriber and equipment trace	201
14.10 Cu	tomized Alerting Tone	201
14.10 1 Get	peral	201
14.10.2	Audio C A T	201
14.10.2	Audio CAT Activated by the Calling Party	201
14.10.2.1	Audio CAT Activated by the Called Derty	
14.10.2.2	Audio CAT Activated by the Called Party	
14.10.5	Introduction	
14.10.3.1	Introduction	
14.10.3.2	Multimedia CAT Activated by the Called Party	
14.10.3.3	Multimedia CAT Activated by the Called Party	
14.10.3.3.1	Mobile originated multimedia call.	
14.10.3.3.2	Mobile terminated multimedia call	
14.10.3.3.3	Example	
15 Tunne	lling	205
15.1 For	ward Bearer Fstablishment	205
15.1.1	Outgoing Side	205
15.1.1	Tunnel Selection	205
15.1.1.1	Initial addressing	205
15.1.1.2	Fast forward tunnelling	205
15.1.1.5	Delayed forward tunnelling	206
15.1.1.4	Bearer control signalling transfer	206
15.1.1.5	Incoming Side	
15.1.2	Initial addressing	
15.1.2.1	Tunnal Salection	
15.1.2.2	Fact forward tunnalling	
15.1.2.5	Dalayad forward tunnalling	
15.1.2.4	Delayed for ward turnering	
15.1.2.5	Every le	
15.1.2.0	Example	
15.2 Dat	walu Dealei Establisiiliell	
15.2.1	Tunnal Solartian	
15.2.1.1	I united Selection	
15.2.1.2	Initial addressing	
15.2.1.5	Fast forward tunneling	
15.2.1.4	Delayed backward tunnelling	
15.2.1.5	Bearer control signalling transfer	
15.2.2	Incoming Side	
15.2.2.1	Turnal Colorising	
15.2.2.2	I unnel Selection	
15.2.2.3	Fast forward tunnelling	
15.2.2.4	Delayed backward tunnelling	
15.2.2.5	Bearer control signalling transfer	
15.2.2.6	Example	
16 Messa	ges/Procedures and their contents	
16.1 Me	ssages between (G)MSC servers	213
16.2 Pro	cedures between (G)MSC server and MGW	
110		

16.2.1	Change Flow Direction	215
16.2.2	Join Bearer Termination	216
16.2.3	Isolate Bearer Termination	216
16.2.4	Establish Bearer	217
16.2.5	Prepare Bearer	219
16.2.6	Reserve Circuit	
1627	Change Through-Connection	222
16.2.8	Activate Interworking Function	222
16.2.9	Release Rearer	222
16.2.10	Rearer Established	223
16.2.10	Bearer Released	223
16.2.11	Release Termination	224
16.2.12	Tunnal Information Un	224
16.2.13	Tunnel Information Down	225
16.2.14	Sand Tana	225
10.2.15	Sten Tone	220
16.2.10	Stop Tone	220
16.2.17	Play Announcement	227
16.2.18	Stop Announcement.	227
16.2.19	Announcement Completed	228
16.2.20	Tone Completed	228
16.2.21	Detect DTMF	229
16.2.22	Stop DTMF Detection	229
16.2.23	Report DTMF	230
16.2.24	Send DTMF	230
16.2.25	Stop DTMF	231
16.2.26	MGW Out-of-Service/ Maintenance locking	231
16.2.27	MGW Communication Up	232
16.2.28	MGW Restoration	232
16.2.29	MGW Register	233
16.2.30	MGW Re-register	234
16.2.31	(G)MSC Server Ordered Re-register	234
16.2.32	(G)MSC Server Restoration	235
16.2.33	(G)MSC Server Out of Service	235
16.2.34	Termination Out-of-Service	236
16.2.35	Termination Restoration	236
16.2.36	Audit Value	237
16.2.37	Audit Capability	237
16.2.38	Capability Update	238
16.2.39	Command Reject	238
16.2.40	Activate Voice Processing Function	239
16.2.41	Modify Bearer Characteristics	240
16.2.42	Protocol Negotiation Result	241
16.2.43	MGW Resource Congestion Handling - Activate	241
16.2.44	MGW Resource Congestion Handling - Indication	
162.45	Bearer Modification Support	242
16246	CTM report	243
16247	Prenare IP Transnort	244
16248	Modify IP Transport Address	2/15
16.2.40	Emergency Call Indication	245
16.2.49	Trace Activation	246
16.2.50	Trace Deactivation	240
16.2.51	Trace Dedutivation	240
16.2.52	Continuity Check Tone	247
16.2.55	Continuity Check Tolle	247
16.2.54	Continuity Check Verify	247
10.2.55	Continuity Check Kesponse	248
10.2.56	Kale Unange	248
16.2.57	Bearer Modifed	249
16.2.58	Bearer Modification Failed	249
16.2.59	I ermination heartbeat indication	250
16.2.60	Inactivity timeout activation	250
16.2.61	Inactivity timeout indication	251
16.2.62	Reserve RTP Connection Point	251

16.2.6	3 Configure RTP Connection Point	254
16.2.64	4 Reserve and Configure RTP Connection Point	257
16.2.6	5 Realm Availability Change - Activation	259
16.2.6	6 Realm Availability Change - Indication	260
16.2.6	7 Media Inactivity Notification	260
17	Bearer Redirect	261
17.1	Example of use of Bearer Redirect with Call Forwarding on No Reply (CFNRy)	261
18	(G)MSC MGW Tandeming	263
18.1	Example of use of MSC MGW Tandeming during call setup to provide bearer access to specialised	
	MGW resources	263
19	Timers for bearer independent CS core network	264
20	Multiple IP Realms	264
Anne	x A (informative). Change History	266
1 Mille	A 11 (mit) multive). Chunge 115001 j minimum minimum minimum minimum minimum minimum minimum minimum minimum mi	
Histor	۲y	269

Foreword

This Technical Specification has been produced by the 3rd Generation Partnership Project (3GPP).

The contents of the present document are subject to continuing work within the TSG and may change following formal TSG approval. Should the TSG modify the contents of the present document, it will be re-released by the TSG with an identifying change of release date and an increase in version number as follows:

Version x.y.z

where:

- x the first digit:
 - 1 presented to TSG for information;
 - 2 presented to TSG for approval;
 - 3 or greater indicates TSG approved document under change control.
- y the second digit is incremented for all changes of substance, i.e. technical enhancements, corrections, updates, etc.
- z the third digit is incremented when editorial only changes have been incorporated in the document.

1 Scope

The present document defines the stage 2 description for the bearer independent CS core network. The stage 2 shall cover the information flow between the GMSC server, MSC server and media gateways. Note that nothing in the present document shall preclude an implementation of a combined MSC Server and MGW. The present document shall show the CS core network termination of the Iu interface in order to cover the information flow stimulus to the core network and describe the interaction with the supplementary and value added services and capabilities.

For the purposes of the present document, the protocol used over the Nc interface is an enhanced call control protocol supporting call bearer separation such as BICC (which is specified in [22]). The protocol used over the Mc interface is H.248 (which is specified in 3GPP TS 29.232 [6]). Existing specifications and recommendations shall not be repeated, as such the relevant specification shall be referred to.

SIP-I based CS core network is further specified in 3GPP TS 23.231 [42].

The present document is applicable only for ATM or IP transport in the CS core network.



Figure 1: CS core network logical architecture

The CAP interfaces and the interfaces towards the HLR are outside the scope of the present document.

Details of Transcoder-Free Operation are outside the scope of the present document. Please see 3GPPTS 23.153 [3] for more information.

2 References

The following documents contain provisions which, through reference in this text, constitute provisions of the present document.

- References are either specific (identified by date of publication, edition number, version number, etc.) or non-specific.
- For a specific reference, subsequent revisions do not apply.
- For a non-specific reference, the latest version applies. In the case of a reference to a 3GPP document (including a GSM document), a non-specific reference implicitly refers to the latest version of that document *in the same Release as the present document*.
- [1] 3GPP TR 21.905: "Vocabulary for 3GPP Specifications".
- [2] 3GPP TS 23.002: "Network architecture".

- [3] 3GPP TS 23.153: "Out of Band Transcoder Control; Stage 2".
- [4] 3GPP TS 24.008: "Mobile radio interface Layer 3 specification; Core network protocols; Stage 3".
- [5] ITU-T Recommendation H.248: "Gateway control protocol".
- [6] 3GPP TS 29.232: "Media Gateway Controller (MGC) Media Gateway (MGW) interface; Stage 3".
- [7] 3GPP TS 29.415: " Core Network Nb Interface User Plane Protocols ".
- [8] 3GPP TS 23.009: "Handover procedures".
- [9] 3GPP TS 23.072: "Call Deflection Supplementary Service; Stage2".
- [10] 3GPP TS 23.078: "Customized Applications for Mobile network Enhanced Logic (CAMEL); Stage 2".
- [11] 3GPP TS 23.079: "Support of Optimal Routeing (SOR); Technical Realization; Stage2".
- [12] 3GPP TS 23.082: "Call Forwarding (CF) Supplementary Services; Stage 2".
- [13] 3GPP TS 23.083: "Call Waiting (CW) and Call Hold (HOLD) Supplementary Services; Stage 2".
- [14] 3GPP TS 23.084: "MultiParty (MPTY) Supplementary Service; Stage 2".
- [15] 3GPP TS 23.091: "Explicit Call Transfer (ECT) Supplementary Service; Stage 2".
- [16] 3GPP TS 23.093: "Technical realization of Completion of Calls to Busy Subscriber (CCBS); Stage 2".
- [17] 3GPP TS 23.135: "Multicall supplementary service; Stage 2".
- [18] 3GPP TS 23.108: "Mobile radio interface layer 3 specification core network protocols; Stage 2".
- [19] 3GPP TS 42.032: "Immediate Service Termination (IST); Service description; Stage 1".
- [20] 3GPP TS 25.415: "UTRAN Iu interface user plane protocols".
- [21] 3GPP TS 29.414: "Core Network Nb data transport and transport signalling".
- [22] 3GPP TS 29.205: "Application of Q.1900 series to bearer independent circuit-switched core network architecture; Stage 3".
- [23] 3GPP TS 29.010: "Information Element Mapping between Mobile Station Base Station System (MS - BSS) and Base Station System - Mobile-services Switching Centre (BSS - MSC); Signalling Procedures and the Mobile Application Part (MAP)".
- [24] 3GPP TS 43.045: "Technical Realization Of Facsimile Group 3 service Transparent".
- [25] 3GPP TS 23.146: "Technical realization of facsimile group 3 non-transparent".
- [26] 3GPP TS 25.413: "UTRAN Iu Interface RANAP signalling"
- [27] 3GPP TS 48.008: 'Mobile Switching Centre Base Station system (MSC BSS) interface layer 3 Specification'
- [28] 3GPP TS 23.226: "Global Text Telephony (GTT); Stage 2"
- [29] 3GPP TS 43.051: " GSM/EDGE Radio Access Network (GERAN) overall description; Stage 2;"
- [30] 3GPP TS 25.412: "UTRAN Iu interface signalling transport".
- [31] 3GPP TS 25.410: "UTRAN Iu Interface: General Aspects and Principles".
- [32] 3GPP TS 25.414: "UTRAN Iu interface data transport and transport signalling".

- [33] 3GPP TS 23.014: "Technical Specification Group Core Network; Support of Dual Tone Multi-Frequency (DTMF) signalling".
- [34] 3GPP TS 32.421: "Subscriber and equipment trace: Trace concepts and requirements".
- [35] 3GPP TS 32.422: "Subscriber and equipment trace: Trace control and configuration management".
- [36] 3GPP TS 32.423: "Subscriber and equipment trace: Trace data definition and management".
- [37] 3GPP TS 29.007: "General requirements on interworking between the Public Land Mobile Network (PLMN) and the Integrated Services Digital Network (ISDN) or Public Switched Telephone Network (PSTN)".
- [38] 3GPP TS 23.172 : "Technical Specification Group Core Network and Terminals;Technical realization of Circuit Switched (CS);multimedia service UDI/RDI fallback and service modification;Stage 2
- [39] 3GPP TS 43.068 : " Voice Group Call Service (VGCS)"
- [40] 3GPP TS 43.069 : " Voice Broadcast Service (VBS)"
- [41] IETF RFC 2663: "IP Network Address Translator (NAT) Terminology and Considerations ".
- [42] 3GPP TS 23.231: "SIP-I based Circuit Switched Core Network ; Stage 2"
- [43] 3GPP TS 29.002: "Mobile Application Part (MAP) specification".
- [44] 3GPP TS 29.235: "Interworking between SIP-I based circuit-switched core network and other networks".
- 3 Symbols and abbreviations

3.1 Symbols

For the purposes of the present document, the following symbols apply:

- Iu Interface between the RNS and the core network. It is also considered as a reference point.
- Mc Interface between the server and the media gateway.
- Nb Interface between media gateways.
- Nc The NNI call control interface between (G)MSC servers.

3.2 Abbreviations

For the purposes of the present document, the following abbreviations apply:

BCF	Bearer Control Function
BICC	Bearer Independent Call Control
CIC	Call Instance Code
CCF	Call Control Function
CS	Circuit Switched
GERAN	GSM/EDGE Radio Access Network
IAM	Initial Address Message
IETF	Internet Engineering Task Force
IP	Internet Protocol
IPv4	Internet Protocol version 4
IPv6	Internet Protocol version 6
MGW	Media GateWay
MGC	Media Gateway Controller
MSC-S	MSC Server
MTP2	Message Transfer Part layer 2
MTP3	Message Transfer Part layer 3

NNI	Network-Network interface
RAB	Radio Access Bearer
RANAP	Radio Access Network Application Protocol
SBSS	Serving Base Station System
SRNS	Serving Radio Network Subsystem
TCAP	Transaction Capabilities Application Part
TFO	Tandem Free Operation
TRAU	Transcoder and Rate Adapter Unit
TrFO	Transcoder Free Operation
UDP	User Datagram Protocol
UTRAN	UMTS Terrestrial Radio Access Network

3.3 Definitions

UE: User equipment. This specification makes no distinction between MS (mobile station) and UE.

A/Gb mode: mode of operation of the UE when connected to the Core Network via GERAN and the A and/or Gb interfaces. Throughout this specification the term GSM refers to GERAN A/Gb mode.

Iu mode: mode of operation of the UE when connected to the Core Network via GERAN or UTRAN and the Iu interface. Throughout this specification the term UMTS refers to UTRAN or GERAN Iu mode.

4 Main Concepts

4.1 General

The circuit switched core network enables the support of different transports (e.g. ATM or IP) in a bearer-independent fashion. For the ATM and IP transport, there is a strict separation between the call control level and the bearer control level. In the case of ATM or IP transport, the passage of compressed speech at variable bit rates is possible through the CS core network.

The CS core network shall employ the MSC server, GMSC server and media gateways. The GMSC server and MSC server shall provide the call control and mobility management functions, and the media gateway shall provide the bearer control and transmission resource functions. The media gateway shall contain the stream manipulating functions.

The GMSC server and MSC servers are connected to the media gateway via the Mc reference point. The MSC servers and GMSC servers are connected with the Nc reference point. There may be a number of call control transit nodes between the MSC server and GMSC server in the Nc reference point. The MGWs are connected with the Nb reference point.

The users connected to the CS core network shall not be aware whether a MSC server – media gateway combination is used, or a monolithic MSC is used.

4.2 Bearer-Independent Call Control

The protocol used on the Nc interface shall be a call control protocol supporting IP and ATM transports in a bearer-independent manner for the ISDN service set, allowing the physical separation of the call control entities from the bearer control entities.

An exception to this bearer independence concept is if Iu interface is on IP and the IP addresses are to be exchanged via call control plane signalling (known by the MSC due to configuration data). In this case the specific handling is described separately.

Another exception to the bearer independence concept is the case of A interface user plane over IP (AoIP). In that case the IP addresses/UDP port numbers are to be exchanged via call control plane signalling. Specific handling is also described separately for this case.

4.3 H.248/MEGACO

H.248/MEGACO has been jointly developed within the ITU-T and the IETF, and supports a separation of call control entities from bearer control entities, and a separation of bearer control entities from transport entities. H.248 is used on the Mc interface between the (G)MSC servers and the media gateway.

5 General Circuit Switched Core Network Domain Architecture

5.1 Logical Architecture

The overall CS core network logical architecture is shown in figure 1.

5.1.1 CS Core Network Nodes

5.1.1.1 MSC Server

The MSC server mainly comprises the call control and mobility control parts of a GSM/UMTS MSC as described in 3GPP TS 23.002 [2]. It is also integrated with a VLR to hold the mobile subscriber's service data and CAMEL related data.

The MSC server terminates the user-network signalling (see 3GPP TS 24.008 [4]) and translates it into the signalling over the Nc interface. The MSC Server terminates the Iu control plane signalling and its transport bearer (see 3GPP TS 25.413 [26] and 3GPP TS 25.412 [30]). It also terminates the signalling over the Mc interface with the Media Gateway.

The MSC server controls the parts of the call state model that pertain to connection control for media channels in an MGW. It also contains the "Call Control Function" in the BICC model.

5.1.1.2 GMSC Server

The GMSC server mainly comprises the call control and mobility control parts of a GSM/UMTS GMSC as described in 3GPP TS 23.002 [2].

The GMSC server terminates the signalling over the Nc interface and the call control interfaces to the external networks. It also terminates the signalling over the Mc interface towards the Media Gateway.

The GMSC server controls the parts of the call state model that pertain to connection control for media channels in an MGW. It also contains the "Call Control Function" in the BICC model.

5.1.1.3 Media Gateway

The Media Gateway terminates the signalling over the Mc interface from the (G)MSC servers. The Media Gateway terminates the Iu transport network control plane signalling with its transport bearer (see 3GPP TS 25.410 [31] and 3GPP TS 25.414 [32]). The Media Gateway also terminates the Iu user plane protocol (see 3GPP TS 25.415 [20]). It also terminates the bearer control signalling and the transport bearer over the Nb interface (see 3GPP TS 29.414 [21] and 3GPP TS 29.415 [7]).

The Media Gateway contains bearer terminations and media manipulation equipment (e.g. transcoders, echo cancellers, or tone senders). It may perform media conversion and framing protocol conversion.

5.1.2 CS Core Network Interfaces and Reference Points

5.1.2.1 Mc Interface

The Mc reference point in the present document considers the aspects of the interface between the (G)MSC server and the MGW. The H.248 protocol [5] together with 3GPP specific extensions/packages shall be used over the Mc interface.

5.1.2.2 Nc Interface

The Network-Network based call control is used over the Nc interface. Any suitable call control protocol may be used over the Nc interface (e.g. BICC).

5.1.2.3 Nb Interface

The bearer control signalling and transport are carried over the Nb interface.

5.2 Network Interworking

5.2.1 Interworking on the Nc Reference Point

Interworking between the Nc reference point, call control protocols and ISUP is defined within the 3GPP stage 3 documentation for each protocol (or by references specified in stage 3 documentation [6]).

5.2.2 Interworking on the Nb Reference Point

The interworking is specified in 3GPP TS 29.415 [7] and 3GPP TS 29.414 [21].

6 Call Establishment

- NOTE1: All message sequence charts in this clause are examples. All valid call establishment message sequences can be derived from the example message sequences and associated message pre-conditions.
- NOTE2: The continuity indication in the IAM is not used to indicate that a continuity check will be performed on the current leg of the call, but it is used to indicate that a Continuity message can be expected as a result of a continuity check on a preceding ISUP circuit, or establishment of a preceding bearer connection.

6.1 Basic Mobile Originating Call

6.1.1 Forward bearer establishment

The mobile originating call shall be established in accordance with 3GPP TS 23.108 [17]. The following clauses describe the additional requirements for the bearer independent CS core network. If out-of-band transcoder control is applied for a speech call, it shall be performed in accordance with 3GPP TS 23.153 [3].

6.1.1.1 MGW selection

The MSC server shall select an MGW for the bearer connection before it performs the access bearer assignment or the network side bearer establishment. This may happen either before sending the IAM or after receiving the Bearer Information message. In the latter case, the MGW selection may be based on a possibly received MGW-id from the succeeding node (bullet 1 or bullet 2 in figure 6.2). For GSM, if performing Service based handover (see 3GPP TS 48.008 [27]) the MSC Server may omit MGW selection at this time.

6.1.1.2 Initial addressing

The MSC server shall indicate in the IAM that forward bearer establishment is to be used. If access bearer assignment has not been completed, the MSC server shall indicate that the Continuity message will follow. However, if late access bearer assignment (assignment after alerting or answer) is used the MSC server shall not indicate that the Continuity message will follow. The MSC server provides the bearer characteristics to the succeeding node in the IAM. If the MGW is selected at an earlier stage the MGW-id may also be provided in the IAM (bullet 1 in figure 6.2).

6.1.1.3 Network side bearer establishment

The MSC server shall select bearer characteristics for the network side bearer connection before sending the IAM. After the succeeding node has provided a bearer address and a binding reference in the Bearer Information message the MSC server uses the Establish Bearer procedure to request the MGW to establish a bearer towards the destination MGW. The MSC server provides the MGW with the bearer address, the binding reference and the bearer characteristics (bullet 2 in figure 6.2).

6.1.1.4 Access bearer assignment

The MSC server shall select bearer characteristics for the access bearer.

For UMTS, before the MSC server starts the access bearer assignment, the MSC server requests the MGW to prepare for the access bearer establishment using the Prepare Bearer procedure. The MSC server requests the MGW to provide a bearer address and a binding reference, provides the MGW with the bearer characteristics and requests notification that the bearer can be modified. For speech calls, the MSC server shall provide the MGW with the speech coding information and conditionally GTT related information in accordance with 3GPP TS 23.226 [28] for the bearer. For a non-speech call the MSC server also provides the MGW with a PLMN Bearer Capability [4]. After the MGW has replied with the bearer address and the binding reference the MSC server requests access bearer assignment using the provided bearer address and binding reference (bullet 3 in figure 6.2) in accordance with 3GPP TS 25.413 [26]. The MSC shall only be notified by the MGW using Bearer Modification Support procedure if the existing link characteristics of the access bearer can be modified at a later stage, see subclause 13.18.1.

For GERAN Iu mode the MSC Server receives the GERAN capabilities within the RANAP INITIAL UE MESSAGE, indicating the services (e.g. for CS speech services the supported codec types and, for an adaptive codec type, the supported codec modes (for definition see [27])), which will be available at the RAB establishment procedure. The MSC server shall take the indicated GERAN capabilities into account as well as the received MS capabilities when negotiating a service. Additionally, when requesting the access bearer assignment the MSC server shall indicate to the GERAN the selected service (e.g. selected codec type). The MSC server shall not set codec information in the NAS Synchronisation Indicator (see [4]). Instead it shall set codec information in the GERAN BSC container.

For GSM, before the MSC server starts the access bearer assignment, the MSC server uses the Reserve Circuit procedure to seize a TDM circuit. For a non-speech call the MSC server also provides the MGW with a PLMN Bearer Capability [4] and a GSM channel coding. After the MGW has replied to the TDM circuit seizure, the MSC server requests access bearer assignment (bullet 4 in figure 6.2) in accordance with 3GPP TS 48.008 [27]. If performing Service based handover (see 3GPP TS 48.008 [27]) the MSC Server may omit to perform Reserve Circuit procedure.

6.1.1.5 Framing protocol initialisation

In 3GPP CS CN speech and data shall be carried using the Iu/Nb User Plane Protocol. The specification for the Iu UP protocol is defined in [20] and the Nb UP Protocol in [7] and [21]. The Iu/Nb UP Protocol is established through the CN in a forward direction. This is established independently of the bearer establishment direction. The MGW derives the forward direction from information sent by the MSC server within the Establish Bearer and Prepare Bearer procedures [6].

6.1.1.6 Confirmation of bearer establishment

If the IAM which was sent to the succeeding node indicated that the Continuity message will follow, the MSC server sends the Continuity message when the access bearer assignment has been completed (bullet 5 in figure 6.2).

6.1.1.7 Through-Connection

During any one of the Prepare Bearer, Reserve Circuit and Establish Bearer procedures, the MSC server will use the Change Through-Connection procedure to request the MGW to through-connect the bearer terminations so that the bearer will be backward through-connected (bullet 2, and bullet 3 or 4 in figure 6.2).

For a multimedia call, the MSC may request the MGW to both-way through-connect the bearer using the Change Through-Connection procedure to generate a multimedia CAT (see subclause 14.10.3.1).

Otherwise when MSC server receives the answer indication, it requests the MGW to both-way through-connect the bearer using the Change Through-Connection procedure (bullet 6 in figure 6.2).

6.1.1.8 Interworking function

The MGW may use an interworking function that is based on the PLMN Bearer Capability [4] of the bearer termination. The activation of the possible interworking function in both bearer terminations will be requested by the MSC server at reception of the answer indication using the Activate Interworking Function procedure (bullet 6 in figure 6.2).

6.1.1.9 Codec handling

The MGW may include a speech transcoder based upon the speech coding information provided to each bearer termination.

6.1.1.10 Voice Processing function

A voice processing function located on the MGW may be used to achieve desired acoustic quality on the bearer terminations. The MSC server shall request the activation of voice processing functions in the bearer terminations. For non-speech calls, the MSC server has the ability to instruct the MGW to disable the voice processing functions (bullet 6 in figure 6.2).

6.1.1.11 Failure handling in MSC server

If any procedure between the MSC server and the MGW has not completed successfully or the MSC server receives a Bearer Released procedure from the MGW, the call shall be cleared as described in clause 7.3, (G)MSC server initiated call clearing or in clause 7.4, MGW initiated call clearing. Alternatively, the MSC server may only release the resources in the MGW that caused the failure, possibly select a new MGW for the bearer connection and continue the call establishment using new resources in the selected MGW.

6.1.1.12 Example

Figure 6.1 shows the network model for the mobile originating call. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling (not applicable in A/Gb mode for the A-interface) and the bearer. The MSC server seizes one context with two bearer terminations in the MGW. The bearer termination T1 is used for the bearer towards the RNC/BSC and the bearer termination T2 is used for the bearer towards the succeeding MGW.



Figure 6.1: Basic Mobile Originating Call, Forward Bearer Establishment (network model)

Figure 6.2 shows the message sequence chart example for the mobile originating call. In the example the MSC server requests seizure of the network side bearer termination and establishment of the bearer when the Bearer Information message is received from the succeeding node. After the network side bearer termination is seized the MSC server requests seizure of the access side bearer termination. When the MSC server receives an answer indication, it shall requests the MGW to both-way through-connect the bearer terminations. The MSC shall also request the possible activation of the interworking function in both terminations and the possible activation of the voice processing functions for the bearer terminations.

3GPP TS 23.205 version 9.4.0 Release 9



Figure 6.2/1: Basic Mobile Originating Call, Forward Bearer Establishment (message sequence chart)



Figure 6.2/2: Basic Mobile Originating Call, Forward Bearer Establishment (message sequence chart continue)

6.1.2 Backward bearer establishment

The basic mobile originating call shall be established in accordance with 3GPP TS 23.108 [17]. The following clauses describe the additional requirements for the bearer independent CS core network. If out-of-band transcoder control is applied for a speech call, it shall be performed in accordance with 3GPP TS 23.153 [3].

6.1.2.1 MGW selection

The MSC server shall select an MGW for the bearer connection before it performs the access bearer assignment or the network side bearer establishment. This happens before sending the IAM (bullet 1 or 2 in figure 6.4). For GSM, if performing Service based handover (see 3GPP TS 48.008 [27]) the MSC Server may omit MGW selection at this time.

6.1.2.2 Network side bearer establishment

The MSC server shall select bearer characteristics for the network side bearer connection before sending the IAM. The MSC server requests the MGW to prepare for the network side bearer establishment using the Prepare Bearer procedure. The MSC server requests the MGW to provide a bearer address and a binding reference, and provides the MGW with the bearer characteristics (bullet 3 in figure 6.4). After the MGW has replied with the bearer address and the binding reference, the MSC server sends the IAM to the succeeding node.

6.1.2.3 Initial addressing

The MSC server shall indicate in the IAM that backward bearer establishment is to be used. If access bearer assignment has not been completed, the MSC server shall indicate that the Continuity message will follow. However, if late access bearer assignment (assignment after alerting or answer) is used the MSC server shall not indicate that the Continuity message will follow. The MSC server provides the bearer characteristics, the bearer address and the binding reference to the succeeding node in the IAM. The MSC server may also provide the MGW-id in the IAM (bullet 4 in figure 6.4).

6.1.2.4 Access bearer assignment

The MSC server shall select bearer characteristics for the access bearer.

For UMTS, before the MSC server starts the access bearer assignment, the MSC server requests the MGW to prepare for the access bearer establishment using the Prepare Bearer procedure. The MSC server requests the MGW to provide a bearer address and a binding reference, provides the MGW with the bearer characteristics and requests notification that the bearer can be modified. For speech calls, the MSC server shall provide the MGW with the speech coding information and conditionally GTT related information in accordance with 3GPP TS 23.226 [28] for the bearer. For a non-speech call the MSC server also provides the MGW with a PLMN Bearer Capability [4]. After the MGW has replied with the bearer address and the binding reference the MSC server requests access bearer assignment using the provided bearer address and binding reference (bullet 1 in figure 6.4) in accordance with 3GPP TS 25.413 [26]. The MSC shall only be notified by the MGW using the Bearer Modification Support procedure if the existing link characteristics of the access bearer can be modified at a later stage, see subclause 13.18.1.

For GERAN Iu mode the MSC Server receives the GERAN capabilities within the RANAP INITIAL UE MESSAGE, indicating the services (e.g. for CS speech services the supported codec types and, for an adaptive codec type, the supported codec modes (for definition see [27])), which will be available at the RAB establishment procedure. The MSC server shall take the indicated GERAN capabilities into account as well as the received MS capabilities when negotiating a service. Additionally, when requesting the access bearer assignment the MSC server shall indicate to the GERAN the selected service (e.g. selected codec type). The MSC server shall not set codec information in the NAS Synchronisation Indicator (see [4]). Instead it shall set codec information in the GERAN BSC container.

For GSM, before the MSC server starts the access bearer assignment, the MSC server uses the Reserve Circuit procedure to seize a TDM circuit. For a non-speech call the MSC server also provides the MGW with a PLMN Bearer Capability [4] and a GSM channel coding. After the MGW has replied the TDM circuit seizure the MSC server requests access bearer assignment (bullet 2 in figure 6.4) in accordance with 3GPP TS 48.008 [27]. If performing Service based handover (see 3GPP TS 48.008 [27]) the MSC Server may omit to perform Reserve Circuit procedure.

6.1.2.5 Framing protocol initialisation

In 3GPP CS CN speech and data shall be carried using the Iu/Nb User Plane Protocol. The specification for the Iu UP protocol is defined in [20] and the Nb UP Protocol in [7] and [21]. The Iu/Nb UP Protocol is established through the CN in a forward direction. This is established independently of the bearer establishment direction. The MGW derives the forward direction from information sent by the MSC server within the Establish Bearer and Prepare Bearer procedures [6].

6.1.2.6 Confirmation of bearer establishment

If the IAM was sent to the succeeding node indicating that the Continuity message will follow, the MSC server sends the Continuity message when the access bearer assignment has been completed.

6.1.2.7 Through-Connection

During the Prepare Bearer or Reserve Circuit procedures, the MSC server will use the Change Through-Connection procedure to request the MGW to through-connect the bearer terminations so that the bearer will be backward through-connected (bullet 1 or 2, and bullet 3 in figure 6.4).

For a multimedia call, the MSC may request the MGW to both-way through-connect the bearer using the Change Through-Connection procedure to generate a multimedia CAT (see subclause 14.10.1).

Otherwise when the MSC server receives the answer indication, it requests the MGW to both-way through-connect the bearer using the Change Through-Connection procedure (bullet 5 in figure 6.4).

6.1.2.8 Interworking function

The MGW may use an interworking function that is based on the PLMN Bearer Capability [4] of the bearer termination. The activation of the possible interworking function in both bearer terminations will be requested by the MSC server at reception of the answer indication using the Activate Interworking Function procedure (bullet 5 in figure 6.4).

6.1.2.9 Codec handling

The MGW may include a speech transcoder based upon the speech coding information provided to each bearer termination.

6.1.2.10 Voice Processing function

A voice processing function located on the MGW may be used to achieve desired acoustic quality on the bearer terminations. The MSC server shall request the activation of the voice processing functions in the bearer terminations. For non-speech calls, the MSC server has the ability to instruct the MGW to disable the voice processing functions (bullet 5 in figure 6.4).

6.1.2.11 Failure handling in MSC server

If any procedure between the MSC server and the MGW has not completed successfully or the MSC server receives a Bearer Released procedure from the MGW, the call shall be cleared as described in clause 7.3, (G)MSC server initiated call clearing or in clause 7.4, MGW initiated call clearing. Alternatively, the MSC server may only release the resources in the MGW that caused the failure, possibly select a new MGW for the bearer connection and continue the call establishment using new resources in the selected MGW.

6.1.2.12 Example

Figure 6.3 shows the network model for the mobile originating call. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling (not applicable in A/Gb mode for the A-interface) and the bearer. The MSC server seizes one context with two bearer terminations in the MGW. The bearer termination T1 is used for the bearer towards the RNC/BSC and the bearer termination T2 is used for the bearer towards the succeeding MGW.



Figure 6.3: Basic Mobile Originating Call, Backward Bearer Establishment (network model)

Figure 6.4 shows the message sequence chart example for the mobile originating call. In the example the MSC server requests seizure of the access side bearer termination and network side bearer termination. As the access bearer assignment has been completed before the IAM, no Continuity message will be sent. When the MSC server receives an answer indication, it requests the MGW to both-way through-connect the bearer terminations. The MSC server, shall also request the possible activation of the interworking function in both bearer terminations. The MSC server shall request the possible activation of the voice processing functions for the bearer terminations.



Figure 6.4/1: Basic Mobile Originating Call, Backward Bearer Establishment (message sequence chart)



Figure 6.4/2: Basic Mobile Originating Call, Backward Bearer Establishment (message sequence chart continue)

6.1.3 Originating Call Establishment For Iu Interface on IP

If IuCS on IP is supported by the MSC server, the Core Network side procedures described in 6.1.1 or 6.1.2 shall apply. For the access side termination, the exchange of IP addresses via call control procedures is described in this clause.

Before the MSC server starts the access bearer assignment, the MSC server requests the MGW to prepare for the access bearer using the Prepare_IP_Transport procedure. The MSC server requests the MGW to provide an IP Transport Address and a Iu UDP Port and provides the MGW with the bearer characteristics. For speech calls, the MSC server shall provide the MGW with the speech coding information and conditionally GTT related information in accordance with 3GPP TS 23.226 [28]. For a non-speech call the MSC server also provides the MGW with a PLMN Bearer Capability [4]. After the MGW has replied with the IP address and UDP Port the MSC server requests access bearer assignment using the provided IP address and UDP Port in accordance with 3GPP TS 25.413 [26]. The IP addresses and UDP Ports of the MGW and the RNC are exchanged via the RANAP procedures. If the bearer transport is IP and IuUP mode is Transparent, when the MSC receives the RANAP RAB assignment response it shall send the RNC IP address and UDP Port to the MGW Access bearer termination using the Modify_IP_Transport_Address procedure.

If the bearer transport is IP and IuUP mode is Support, the MGW shall use the source IP address and UDP Port of the IuUP Init packet received from the radio access network as the destination address for subsequent downlink packets.

The sequence is shown in figure 6.1.3/1.

3GPP TS 23.205 version 9.4.0 Release 9

31



Figure 6.1.3/1: Call Establishment for Iu on IP

6.1.4 Forward bearer establishment with Trace Session Activation

The mobile originating call shall be established in accordance with 3GPP TS 23.108 [17]. The following clause describe the additional requirement for the bearer independent CS core network if a trace session is activated from a MSC Server to a MGW.

6.1.4.1 Trace activation

When a Trace Session is activated in the MSC Server and the trace control and configuration parameters requires Trace Session activation to the MGW, the MSC Server activates the Trace Session to the MGW by using the trace activation procedure.

6.1.4.2 Example

Figure 6.1.4.1 shows the network model for the mobile originating call. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling (not applicable in A/Gb mode for the A-interface) and the bearer. The MSC server seizes one context with two bearer terminations in the MGW. The bearer termination T1 is used for the bearer towards the RNC/BSC and the bearer termination T2 is used for the bearer towards the succeeding MGW.



Figure 6.1.4.1: Basic Mobile Originating Call, Forward Bearer Establishment with Trace Session Activation (network model)

Figure 6.1.4.2 shows the message sequence chart example for the mobile originating call. In the example the MSC server activates the Trace Session to MGW when it request the creation of the incoming and outgoing termination with ADD request. MGW sends the Trace Session Activation result with NOTIFY request to MSC Server.

3GPP TS 23.205 version 9.4.0 Release 9



Figure 6.1.4.2/1: Basic Mobile Originating Call, Forward Bearer Establishment with Trace Session Activation (message sequence chart)



Figure 6.1.4.2/2: Basic Mobile Originating Call, Forward Bearer Establishment with Trace Session Activation (message sequence chart continue)

6.1.5 Originating Call Establishment for A interface over IP

If AoIP is supported by the MSC server, the Core Network side procedures described in 6.1.1 or 6.1.2 shall apply. For the access side termination, the exchange of IP addresses and UDP port numbers via call control procedures is described in this clause.

Before the MSC server sends the Assignment Request to the BSC, the MSC server requests the MGW to reserve an RTP bearer termination using the Reserve RTP Connection Point procedure. The MSC server requests the MGW to reserve an IP Address and UDP Port and also may indicate that the IP interface type is for AoIP. The MGW reserves the RTP termination and indicates the IP address and UDP port number to the MSC server. MSC server then requests access bearer assignment using the provided IP address and UDP Port. When MSC server receives the BSSMAP ASSIGNMENT COMPLETE message, it shall send the BSC IP address and UDP Port to the MGW Access bearer termination using the Configure RTP Connection Point procedure.

The sequence is shown in figure 6.1.5.1.



Figure 6.1.5.1: Originating Call Establishment for AoIP

6.2 Basic Mobile Terminating Call

6.2.1 Forward bearer establishment

The basic mobile terminating call shall be established in accordance with 3GPP TS 23.108 [18]. The following clauses describe the additional requirements for the bearer independent CS core network. If out-of-band transcoder control is applied for a speech call, it shall be performed in accordance with 3GPP TS 23.153 [3].

6.2.1.1 GMSC server

6.2.1.1.1 MGW selection

The GMSC server shall select an MGW for the bearer connection before it performs the incoming side bearer establishment or the outgoing side bearer establishment. This may happen either before sending the IAM or after receiving the Bearer Information message. If the GMSC server received an MGW-id from the preceding node and/or from the succeeding node, then it may use one of them for the MGW selection (bullet 1 or bullet 4 in figure 6.6).

NOTE: As an implementation option, if there is no need for the GMSC server to manipulate the bearer, the GMSC server may perform call control signalling without any associated MGW. In that case the bearer related information shall be passed transparently through the GMSC server.
3GPP TS 23.205 version 9.4.0 Release 9

6.2.1.1.2 Initial addressing

The GMSC server shall indicate in the IAM that forward bearer establishment is to be used. The GMSC server shall also indicate in the IAM that the Continuity message will follow if either of the following conditions is satisfied before sending the IAM:

- 1. the incoming IAM indicated that the Continuity message will follow, but no Continuity message has been received;
- 2. the GMSC server selected an MGW, but a notification of successful bearer establishment on the incoming side has not been received from the MGW.

The GMSC server shall provide the bearer characteristics to the succeeding node in the IAM. If the MGW is selected at an early stage the MGW-id may also be provided in the IAM (bullet 1 in figure 6.6).

6.2.1.1.3 Outgoing side bearer establishment

The GMSC server shall select bearer characteristics for the outgoing side bearer connection before it sends the IAM. After the GMSC server has received a bearer address and a binding reference in the Bearer Information message from the succeeding node the GMSC server requests the MGW to establish a bearer to the given destination MGW using the Establish Bearer procedure. The GMSC server shall provide the MGW with the bearer address, the binding reference and the bearer characteristics (bullet 4 in figure 6.6).

6.2.1.1.4 Incoming side bearer establishment

The GMSC server requests the MGW to prepare for the incoming side bearer establishment using the Prepare Bearer procedure. The GMSC server requests the MGW to provide a bearer address, a binding reference and to notify when the bearer is established (bullet 5 in figure 6.6). The GMSC server also provides the MGW with the bearer characteristics that was received from the preceding node in the IAM. After the MGW has replied with the bearer address and the binding reference, the GMSC server sends the Bearer Information message to the preceding node. The GMSC server may also include the MGW-id in the Bearer Information message (bullet 6 in figure 6.6).

NOTE: The incoming side bearer establishment may take place either before or after HLR interrogation.

6.2.1.1.5 Framing protocol initialisation

In 3GPP CS CN speech and data shall be carried using the Iu/Nb User Plane Protocol. The specification for the Iu UP protocol is defined in [20] and the Nb UP Protocol in [7] and [21]. The Iu/Nb UP Protocol is established through the CN in a forward direction. This is established independently of the bearer establishment direction. The MGW derives the forward direction from information sent by the MSC server within the Establish Bearer and Prepare Bearer procedures [6]. The notification of bearer establishment shall not be sent until the Iu/Nb UP has been initialised.

6.2.1.1.6 Through-Connection

During the Prepare Bearer and Establish Bearer procedures, the GMSC server will use the Change Through-Connection procedure to request the MGW to both-way through-connect the bearer termination (bullet 4 and bullet 5 in figure 6.6).

6.2.1.1.7 Confirmation of bearer establishment

If the IAM which was sent to the succeeding node indicated that the Continuity message will follow, the Continuity message shall be sent when both of the following conditions are satisfied:

- 1. Either:
 - a. The incoming IAM indicated that the Continuity message will follow, and a Continuity message has been received from the preceding node (bullet 8 in figure 6.6), or
 - b. The incoming IAM did not indicate that the Continuity message will follow;
- 2. Either:
 - a. The GMSC server has selected an MGW, and a notification of successful bearer establishment in the incoming side has been received from the MGW (bullet 7 in figure 6.6), or

b. MGW selection is not required for this call.

6.2.1.1.8 Voice Processing function

A voice processing function located on the MGW may be used to achieve desired acoustic quality on the bearer terminations. The GMSC server shall request the activation of the voice processing functions in the bearer terminations. For non-speech calls, the GMSC server has the ability to instruct the MGW to disable the voice processing functions (bullet 13 in figure 6.6). The voice activation request from the GMSC server to MGWa may be issued as soon as bullet 8 in figure 6.6, and may be issued as late as bullet 13 in figure 6.6 as illustrated.

6.2.1.1.9 Failure handling in GMSC server

If any procedure between the GMSC server and the MGW has not completed successfully or the GMSC server receives a Bearer Released procedure from the MGW, the call shall be cleared as described in clause 7.3, (G)MSC server initiated call clearing or in clause 7.4, MGW initiated call clearing. Alternatively, the GMSC server may only release the resources in the MGW that caused the failure, possibly select a new MGW for the bearer connection and continue the call establishment using new resources in the selected MGW.

6.2.1.2 MSC server

6.2.1.2.1 Paging

If the network side bearer establishment is delayed whilst the paging procedure is completed, the MSC server starts the Start_Bearer_Establishment timer when the paging procedure is started. The Start_Bearer_Establishment timer is stopped when the paging procedure is completed, or optionally when the Call Confirmed message is received in accordance with 3GPP TS 23.153 [3]. If the Start_Bearer_Establishment timer expires, the MSC server starts the network side bearer establishment.

6.2.1.2.2 Call setup

The MSC server indicates to the UE in the SETUP message that early access bearer assignment is used in order to establish the bearer end-to-end before the UE starts alerting. The MSC server indicates to the UE in SETUP message that early access bearer assignment is used if either of the following conditions is satisfied before sending the SETUP message (bullet 2 in figure 6.6):

- 1. The incoming IAM indicated that the Continuity message will follow, but no Continuity message has been received;
- 2. A notification of successful bearer establishment in the network side has not been received from the MGW.

6.2.1.2.3 MGW selection

The MSC server shall select an MGW for the bearer connection before it performs the network side bearer establishment or the access bearer assignment. This happens at latest after the UE has sent the Call Confirmed message. If the MSC server received an MGW-id from the preceding node, it may use this for the MGW selection (bullet 3 in figure 6.6). For GSM, if performing Service based handover (see 3GPP TS 48.008 [27]) the MSC Server may omit MGW selection at this time.

6.2.1.2.4 Network side bearer establishment

The MSC server requests the MGW to prepare for the network side bearer establishment using the Prepare Bearer procedure. The MSC server requests the MGW to provide a bearer address, a binding reference and to notify when the bearer is established (bullet 3 in figure 6.6). The MSC server also provides the MGW with the bearer characteristics that was received from the preceding node in the IAM. After the MGW has replied with the bearer address and the binding reference, the MSC server provides the Bearer Information message to the preceding node. The MSC server may also provide the MGW-id in the Bearer Information message.

6.2.1.2.5 Access bearer assignment

The access bearer assignment shall be started only when both of the following conditions are satisfied:

- 1. Either:
 - a. The incoming IAM indicated that the Continuity message will follow, and a Continuity message has been received from the preceding node, or
 - b. The incoming IAM did not indicate that the Continuity message will follow;
- 2. A notification of successful bearer establishment in the network side has been received from the MGW (bullet 6 in figure 6.6).

The MSC server shall select bearer characteristics for the access bearer.

For the access bearer assignment in UMTS the MSC server requests the MGW to prepare for the access bearer establishment using the Prepare Bearer procedure. The MSC server requests the MGW to provide a bearer address and a binding reference, provides the MGW with the bearer characteristics and requests notification that the bearer can be modified. For speech calls, the MSC server shall provide the MGW with the speech coding information and conditionally GTT related information in accordance with 3GPP TS 23.226 [28] for the bearer. For a non-speech call the MSC server also provides the MGW with a PLMN Bearer Capability [4]. After the MGW has replied with the bearer address and the binding reference the MSC server requests the access bearer assignment using the provided bearer address and the binding reference (bullet 9 in figure 6.6) in accordance with 3GPP TS 25.413 [26]. The MSC shall only be notified by the MGW using the Bearer Modification Support procedure if the existing link characteristics of the access bearer can be modified at a later stage, see subclause 13.18.1.

For GERAN Iu mode the MSC Server receives the GERAN capabilities within the RANAP INITIAL UE MESSAGE, indicating the services (e.g. for CS speech services the supported codec types and, for an adaptive codec type, the supported codec modes (for definition see [27])), which will be available at the RAB establishment procedure. The MSC server shall take the indicated GERAN capabilities into account as well as the received MS capabilities when negotiating a service. Additionally, when requesting the access bearer assignment the MSC server shall indicate to the GERAN the selected service (e.g. selected codec type). The MSC server shall not set codec information in the NAS Synchronisation Indicator (see [4]). Instead it shall set codec information in the GERAN BSC container.

For GSM, before the MSC server starts the access bearer assignment, the MSC server uses the Reserve Circuit procedure to seize a TDM circuit. For a non-speech call the MSC server also provides the MGW with a PLMN Bearer Capability [4] and a GSM channel coding. After the MGW has replied the TDM circuit seizure the MSC server requests access bearer assignment (bullet 10 in figure 6.6) in accordance with 3GPP TS 48.008 [27]. If performing Service based handover (see 3GPP TS 48.008 [27]) the MSC Server may omit to perform Reserve Circuit procedure.

6.2.1.2.6 Framing protocol initialisation

In 3GPP CS CN speech and data shall be carried using the Iu/Nb User Plane Protocol. The specification for the Iu UP protocol is defined in [20] and the Nb UP Protocol in [7] and [21]. The Iu/Nb UP Protocol is established through the CN in a forward direction. This is established independently of the bearer establishment direction. The MGW derives the forward direction from information sent by the MSC server within the Establish Bearer and Prepare Bearer procedures [6]. The notification of bearer establishment shall not be sent until the Nb UP has been initialised.

6.2.1.2.7 Called party alerting

For a speech call, when the MSC server receives an Alerting message, it requests the MGW to provide a ringing tone to the calling party using the Send Tone procedure (bullet 11 in figure 6.6).

NOTE: Other kind of tones may be provided to the calling party at an earlier stage of the call establishment.

6.2.1.2.8 Called party answer

For a speech call, when the MSC server receives a Connect message, it requests the MGW to stop providing the ringing tone to the calling party using the Stop Tone procedure (bullet 12 in figure 6.6).

6.2.1.2.9 Through-Connection

During the Prepare Bearer and Reserve Circuit procedures, the MSC server will use the Change Through-Connection procedure to request the MGW to through-connect the bearer terminations so that the bearer will be not through-connected (bullet 3, and bullet 9 or 10 in figure 6.6).

When the MSC server receives the Connect message, it requests the MGW to both-way through-connect the bearer using the Change Through-Connection procedure (bullet 12 in figure 6.6).

6.2.1.2.10 Interworking function

The MGW may use an interworking function that is based on the PLMN Bearer Capability [4] of the bearer termination. The activation of the possible interworking function in both bearer terminations will be requested by the MSC server at reception of the Connect message using the Activate Interworking Function procedure (bullet 12 in figure 6.6).

6.2.1.2.11 Codec handling

The MGW may include a speech transcoder based upon the speech coding information provided to each bearer termination.

6.2.1.2.12 Voice Processing function

A voice processing function located on the MGW may be used to achieve desired acoustic quality on the bearer terminations. The MSC server shall request the activation of the voice processing functions in the bearer terminations. For non-speech calls, the MSC server has the ability to instruct the MGW to disable the voice processing functions (bullet 12 in figure 6.6).

6.2.1.2.13 Failure handling in MSC server

If any procedure between the MSC server and the MGW is not completed successfully, the call shall be cleared as described in clause 7.3, (G)MSC server initiated call clearing. Alternatively, the MSC server may only release the resources in the MGW that caused the failure, possibly select a new MGW for the bearer connection and continue the call establishment using new resources in the selected MGW.

6.2.1.2.14 Example

Figure 6.5 shows the network model for the basic mobile terminating call. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling (not applicable in A/Gb mode for the A-interface) and the bearer. The MSC server seizes one context with two bearer terminations in MGWb. The bearer termination T1 is used for the bearer towards the RNC/BSC and the bearer termination T2 is used for the bearer towards the GMSC server seizes one context with two bearer terminations in MGWa. The bearer termination T3 is used for the bearer towards the MSC server selected MGWa and the bearer termination T4 is used for the bearer termination T4 is used for the bearer towards the preceding MGW.



Figure 6.5: Basic Mobile Terminating Call Forward Bearer Establishment (network model)

Figure 6.6 shows the message sequence example for the basic mobile terminating call. In the example the GMSC server requests seizure of the outgoing side bearer termination and establishment of the bearer when the Bearer Information message is received from the MSC server. After the outgoing side bearer termination is seized the GMSC server requests seizure of the incoming side bearer termination. The MGW sends a notification of an established incoming side bearer. The MSC server requests seizure of the network side bearer termination when Call Confirmed message is

received from the UE. The MGW sends a notification of an established network side bearer. When the Continuity message is received from the GMSC server, the MSC server requests seizure of the access side bearer termination. For a speech call the MSC server requests MGW to provide a ringing tone to the calling party at alerting. At answer the MSC server requests MGW to both-way through-connect the bearer. For a speech call the MSC server requests MGW to stop the ringing tone to the calling party at answer. When the MSC server receives an answer indication, it shall request the possible activation of the interworking function in both bearer terminations. The (G)MSC server shall request the possible activation of the voice processing functions for the bearer terminations.



Figure 6.6/1: Basic Mobile Terminating Call, Forward Bearer Establishment (message sequence chart)

41





6.2.2 Backward bearer establishment

The basic mobile terminating call shall be established in accordance with 3GPP TS 23.108 [4]. The following clauses describe the additional requirements for the bearer independent CS core network. If out-of-band transcoder control is applied for a speech call, it shall be performed in accordance with 3GPP TS 23.153 [3].

6.2.2.1 GMSC server

6.2.2.1.1 MGW selection

The GMSC server shall select an MGW for the bearer connection before it performs the incoming side bearer establishment or the outgoing side bearer establishment. This happens before sending the IAM. If the GMSC server received an MGW-id from the preceding node, it may use this for the MGW selection (bullet 1 in figure 6.8).

NOTE 1: As an implementation option, if there is no need for the GMSC server to manipulate the bearer, the GMSC server may perform call control signalling without any associated MGW. In that case the bearer related information shall be provided transparently through the GMSC server.

6.2.2.1.2 Outgoing side bearer establishment

The GMSC server shall select bearer characteristics for the outgoing side bearer connection before it sends the IAM. The GMSC server requests the MGW to prepare for the outgoing side bearer establishment using the Prepare Bearer procedure. The GMSC server requests the MGW to provide a bearer address and a binding reference, and provides the MGW with the bearer characteristics (bullet 3 in figure 6.8). After the MGW has replied with the bearer address and the binding reference, the GMSC server sends the IAM to the succeeding node.

6.2.2.1.3 Initial addressing

The GMSC server shall indicate in the IAM that backward bearer establishment is to be used. The GMSC server shall also indicate in the IAM that the Continuity message will follow if either of the following conditions is satisfied before sending the IAM:

- 1. The incoming IAM indicated that the Continuity message will follow, but no Continuity message has been received, or
- 2. The GMSC server selected an MGW, but a notification of successful bearer establishment on the incoming side has not been received from the MGW.

The GMSC server shall provide the bearer characteristics to the succeeding node in the IAM. The MGW-id may also be provided in the IAM (bullet 4 in figure 6.8).

6.2.2.1.4 Incoming side bearer establishment

The GMSC server requests the MGW to establish a bearer to the given destination MGW and to notify when the bearer is established using the Establish Bearer procedure. The GMSC server provides the MGW with the bearer address, the binding reference and the bearer characteristics that were received from the preceding node in the IAM (bullet 1 in figure 6.8).

NOTE 2: The incoming side bearer establishment may take place either before or after HLR interrogation.

6.2.2.1.5 Framing protocol initialisation

In 3GPP CS CN speech and data shall be carried using the Iu/Nb User Plane Protocol. The specification for the Iu UP protocol is defined in [20] and the Nb UP Protocol in [7] and [21]. The Iu/Nb UP Protocol is established through the CN in a forward direction. This is established independently of the bearer establishment direction. The MGW derives the forward direction from information sent by the MSC server within the Establish Bearer and Prepare Bearer procedures [6]. The notification of bearer establishment shall not be sent until the Iu/Nb UP has been initialised

6.2.2.1.6 Through-Connection

During the Prepare Bearer and Establish Bearer procedures, the GMSC server will use the Change Through-Connection procedure to request the MGW to both-way through-connect the bearer termination (bullet 1 and bullet 3 in figure 6.8).

6.2.2.1.7 Confirmation of bearer establishment

If the IAM which was sent to the succeeding node indicated that the Continuity message will follow, the Continuity message shall be sent when both of the following conditions are satisfied:

- 1. Either:
 - a. The incoming IAM indicated that the Continuity message will follow, and a Continuity message has been received from the preceding node, or
 - b. The incoming IAM did not indicate that the Continuity message will follow;
- 2. Either:
 - a. The GMSC server has selected an MGW, and a notification of successful bearer establishment in the incoming side has been received from the MGW (bullet 2 in figure 6.8), or
 - b. MGW selection is not required for this call.

6.2.2.1.8 Voice Processing function

A voice processing function located on the MGW may be used to achieve desired acoustic quality on the bearer terminations. The (G)MSC server shall request the activation of voice processing functions in the bearer terminations. For non-speech calls, the GMSC server has the ability to instruct the MGW to disable the voice processing functions (bullet 12 in figure 6.8). The voice activation request from the GMSC server to MGWa may be issued as soon as bullet 4 in figure 6.8, and may be issued as late as bullet 12 in figure 6.8 as illustrated.

6.2.2.1.9 Failure handling in GMSC server

If any procedure between the MSC server and the MGW is not completed successfully or the GMSC server receives a Bearer Released procedure from the MGW, the call shall be cleared as described in clause 7.3, (G)MSC server initiated call clearing or in clause 7.4, MGW initiated call clearing. Alternatively, the GMSC server may only release the resources in the MGW that caused the failure, possibly select a new MGW for the bearer connection and continue the call establishment using new resources in the selected MGW.

6.2.2.2 MSC server

6.2.2.2.1 Paging

If the network side bearer establishment is delayed whilst the paging procedure is completed, the MSC server starts the Start_Bearer_Establishment timer when the paging procedure is started. The Start_Bearer_Establishment timer is stopped when the paging procedure is completed, or optionally when the Call Confirmed message is received in accordance with 3GPP TS 23.153 [3]. If the Start_Bearer_Establishment timer expires, the MSC server starts the network side bearer establishment.

6.2.2.2.2 Call setup

The MSC server indicates to the UE in the SETUP message that early access bearer assignment is used in order to establish the bearer end-to-end before the UE starts alerting. The MSC server indicates to the UE in the SETUP message that early access bearer assignment is used, if and only if, either of the following conditions are satisfied before sending the SETUP message (bullet 5 in figure 6.8):

- 1. If the IAM indicated that the Continuity message will follow, but no Continuity message has been received.
- 2. A notification of successful bearer establishment in the network side has not been received from the MGW.

6.2.2.2.3 MGW selection

The MSC server shall select an MGW for the bearer connection before it performs the network side bearer establishment or the access bearer assignment. This happens at latest after the UE has sent the Call Confirmed message. If the MSC server received an MGW-id from the preceding node, it may use this for the MGW selection (bullet 6 in figure 6.8). For GSM, if performing Service based handover (see 3GPP TS 48.008 [27]) the MSC Server may omit MGW selection at this time.

6.2.2.2.3 Network side bearer establishment

The MSC server requests the MGW to establish a bearer to the given destination MGW and to notify when the bearer is established using the Establish Bearer procedure. The MSC server provides the MGW with the bearer address, the binding reference and the bearer characteristics that were received from the preceding node in the IAM (bullet 6 in figure 6.8).

6.2.2.2.4 Access bearer assignment

The access bearer assignment shall be started only when both of the following conditions are satisfied:

- 1. Either:
 - a. The incoming IAM indicated that the Continuity message will follow, and a Continuity message has been received from the preceding node, or
 - b. The incoming IAM did not indicate that the Continuity message will follow;
- 2. A notification of successful bearer establishment in the network side has been received from the MGW (bullet 7 in figure 6.8).

The MSC server shall select bearer characteristics for the access bearer.

For the access bearer assignment in UTRAN the MSC server requests the MGW to prepare for the access bearer establishment using the Prepare Bearer procedure. The MSC server requests the MGW to provide a bearer address and a binding reference, provides the MGW with the bearer characteristics and requests notification that the bearer can be modified. For speech calls, the MSC server shall provide the MGW with the speech coding information and conditionally GTT related information in accordance with 3GPP TS 23.226 [28] for the bearer. For a non-speech call the MSC server also provides the MGW with a PLMN Bearer Capability [4]. After the MGW has replied with the bearer address and the binding reference the MSC server requests the access bearer assignment using the provided bearer address and the binding reference (bullet 8 in figure 6.8) in accordance with 3GPP TS 25.413 [26]. The MSC shall only be notified by the MGW using the Bearer Modification Support procedure if the existing link characteristics of the access bearer can be modified at a later stage, see subclause 13.18.1.

For GERAN Iu mode the MSC Server receives the GERAN capabilities within the RANAP INITIAL UE MESSAGE, indicating the services (e.g. for CS speech services the supported codec types and, for an adaptive codec type, the supported codec modes (for definition see [27])), which will be available at the RAB establishment procedure. The MSC server shall take the indicated GERAN capabilities into account as well as the received MS capabilities when negotiating a service. Additionally, when requesting the access bearer assignment the MSC server shall indicate to the GERAN the selected service (e.g. selected codec type). The MSC server shall not set codec information in the NAS Synchronisation Indicator (see [4]). Instead it shall set codec information in the GERAN BSC container..

For GSM, before the MSC server starts the access bearer assignment, the MSC server uses the Reserve Circuit procedure to seize a TDM circuit. For a non-speech call the MSC server also provides the MGW with a PLMN Bearer Capability [4] and a GSM channel coding. After the MGW has replied the TDM circuit seizure the MSC server requests access bearer assignment (bullet 9 in figure 6.8) in accordance with 3GPP TS 48.008 [27]. If performing Service based handover (see 3GPP TS 48.008 [27]) the MSC Server may omit to perform Reserve Circuit procedure.

6.2.2.2.5 Framing protocol initialisation

In 3GPP CS CN speech and data shall be carried using the Iu/Nb User Plane Protocol. The specification for the Iu UP protocol is defined in [20] and the Nb UP Protocol in [7] and [21]. The Iu/Nb UP Protocol is established through the CN in a forward direction. This is established independently of the bearer establishment direction. The MGW derives the forward direction from information sent by the MSC server within the Establish Bearer and Prepare Bearer procedures [6]. The notification of bearer establishment shall not be sent until the Nb UP has been initialised.

6.2.2.2.6 Called party alerting

For a speech call, when the MSC server receives an Alerting message, it requests the MGW to provide a ringing tone to the calling party using the Send Tone procedure (bullet 10 in figure 6.8).

NOTE: Other kind of tones may be provided to the calling party at an earlier stage of the call establishment.

6.2.2.2.7 Called party answer

For a speech call, when the MSC server receives a Connect message, it requests the MGW to stop providing the ringing tone to the calling party using the Stop Tone procedure (bullet 11 in figure 6.8).

6.2.2.2.8 Through-Connection

During any one of the Prepare Bearer, Reserve Circuit and Establish Bearer procedures, the MSC server will use the Change Through-Connection procedure to request the MGW to through-connect the bearer terminations so that the bearer will be not through-connected (bullet 6, and bullet 8 or 9 in figure 6.8).

When the MSC server receives the Connect message, it requests the MGW to both-way through-connect the bearer using the Change Through-Connection procedure (bullet 11 in figure 6.8).

6.2.2.2.9 Interworking function

The MGW may use an interworking function that is based on the PLMN Bearer Capability [4] of the bearer termination. The activation of the possible interworking function in both bearer terminations will be requested by the MSC server at reception of the Connect message using the Activate Interworking Function procedure (bullet 11 in figure 6.8).

6.2.2.2.10 Codec handling

The MGW may include a speech transcoder based upon the speech coding information provided to each bearer termination.

6.2.2.2.11 Voice Processing function

A voice processing function located on the MGW may be used to achieve desired acoustic quality on the bearer terminations. The MSC server shall request the activation of the voice processing functions in the bearer terminations. For non-speech calls, the MSC server has the ability to instruct the MGW to disable the voice processing functions (bullet 11 in figure 6.8).

6.2.2.2.12 Failure handling in MSC server

If any procedure between the MSC server and the MGW is not completed successfully or the MSC server receives a Bearer Released procedure from the MGW, the call shall be cleared as described in clause 7.3, (G)MSC server initiated call clearing or in clause 7.4, MGW initiated call clearing. Alternatively, the MSC server may only release the resources in the MGW that caused the failure, possibly select a new MGW for the bearer connection and continue the call establishment using new resources in the selected MGW.

6.2.2.2.13 Example

Figure 6.7 shows the network model for the basic mobile terminating call. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling (not applicable in A/Gb mode for the A-interface) and the bearer. The MSC server seizes one context with two bearer terminations in MGWb. The bearer termination T1 is used for the bearer towards the RNC/BSC and the bearer termination T2 is used for the bearer towards the GMSC server seizes one context with two bearer terminations in MGWa. The bearer termination T3 is used for the bearer towards the MSC server selected MGWa and the bearer termination T4 is used for the bearer termination T4 is used for the bearer towards the preceding MGW.



Figure 6.7: Basic Mobile Terminating Call, Backward Bearer Establishment (network model)

Figure 6.8 shows the message sequence example for the basic mobile terminating call. In the example the GMSC server requests seizure of the incoming side bearer termination and establishment of the bearer first. After a notification of incoming side bearer establishment has been received from the MGW, the GMSC server requests seizure of the outgoing side bearer termination. The MSC server requests seizure of the network side bearer termination and establishment of the bearer termination and establishment of the bearer when the Call Confirmed message is received from the UE. After a notification of the network side bearer establishment has been received from the MGW the MSC server requests seizure of the access side bearer termination. For a speech call, When the MSC server receives an alerting message, it requests MGW to provide a ringing tone to the calling party. When the MSC server receives an answer indication, it requests MGW to stop the ringing tone to the calling party and requests the possible activation of the interworking function in both bearer terminations. The (G)MSC server shall request the possible activation of the voice processing functions for the bearer terminations.



Figure 6.8/1: Basic Mobile Terminating Call, Backward Bearer Establishment (message sequence chart)



Figure 6.8/2: Basic Mobile Terminating Call, Backward Bearer Establishment (message sequence chart continue)

6.2.3 Terminating Call Establishment For Iu Interface on IP

If IuCS on IP is supported by the MSC server, the Core Network side procedures described in 6.2.1 or 6.2.2 shall apply. For the access bearer termination, the exchange of IP addresses via call control procedures is described in this clause.

Before the MSC server starts the access bearer assignment, the MSC server requests the MGW to prepare for the access bearer using the Prepare_IP_Transport procedure. The MSC server requests the MGW to provide an IP Transport Address and UDP Port and provides the MGW with the bearer characteristics. For speech calls, the MSC server shall provide the MGW with the speech coding information and conditionally GTT related information in accordance with 3GPP TS 23.226 [28]. For a non-speech call the MSC server also provides the MGW with a PLMN Bearer Capability [4]. After the MGW has replied with the IP address and UDP Port the MSC server requests access bearer assignment using the provided IP address and UDP Port in accordance with 3GPP TS 25.413 [26]. The IP addresses and UDP Ports of the MGW and the RNC are exchanged via the RANAP procedures. If the bearer transport is IP and IuUP mode is Transparent, when the MSC receives the RANAP RAB assignment response it shall send the RNC IP address and UDP Port to the MGW Access bearer termination using the Modify_IP_Transport_Address procedure.

If the bearer transport is IP and IuUP mode is Support, the MGW shall use the source IP address and UDP Port of the IuUP Init packet received from the radio access network as the destination address for subsequent downlink packets.

The sequence is shown in figure 6.2.3/1.



Figure 6.2.3/1 Terminating Call Establishment For Iu Interface on IP

6.2.4 Terminating Call Establishment For A Interface over IP

If AoIP is supported by the MSC server, the Core Network side procedures described in 6.2.1 or 6.2.2 shall apply. For the access side termination, the exchange of IP addresses via call control procedures is described in this clause.

Before the MSC server sends the Assignment Request to the BSC, the MSC server requests the MGW to reserve an RTP bearer termination using the Reserve RTP Connection Point procedure. The MSC server requests the MGW to reserve an IP Address and UDP Port and also may indicate that the IP interface type is for A interface over IP. The MGW reserves the AoIP termination and indicates the IP address and UDP port number to the MSC server. MSC server then requests access bearer assignment using the provided IP address and UDP Port. When MSC server receives the BSSMAP ASSIGNMENT COMPLETE message, it shall send the BSC IP address and UDP Port to the MGW Access bearer termination using the Configure RTP Connection Point procedure.

The sequence is shown in figure 6.2.3/1.



Figure 6.2.3/1 Terminating Call Establishment For AoIP

7 Call Clearing

NOTE: All message sequence charts in this clause are examples. All valid call establishment message sequences can be derived from the example message sequences and associated message pre-conditions.

7.1 Network Initiated

The terms "incoming" and "outgoing" in the following text refers to the direction of propagation of the Release message, not to the direction of establishing the original call.

7.1.1 GMSC server

7.1.1.1 Call clearing from the incoming side

Once the Release message has been received from the preceding node, the GMSC server releases any MGW allocated resources for the incoming side. If any resources were seized in the MGW, the GMSC server shall use the Release Bearer and Release Termination procedures to indicate to the MGW to remove the incoming side bearer termination and that the bearer can be released towards the preceding MGW. After the resources in the MGW are released the GMSC server sends the Release Complete message to the preceding node.

7.1.1.2 Call clearing to the outgoing side

The GMSC server sends the Release message to the succeeding node. Once the succeeding node has sent the Release Complete message, the GMSC server releases any MGW allocated resources for the outgoing side. If any resources were seized in the MGW, the GMSC server shall use the Release Bearer and Release Termination procedures to indicate to the MGW to remove the outgoing side bearer termination and that the bearer can be released towards the succeeding MGW.

7.1.2 MSC server

The network initiated call clearing shall be performed in accordance with 3GPP TS 23.108 [18]. The following clauses describe the additional requirements for the bearer independent CS core network.

7.1.2.1 Call clearing from the network side

Once the Release message has been received from the preceding/succeeding node, the MSC server releases any MGW allocated resources for the network side. If any resources were seized in the MGW, the MSC server shall use the Release Bearer and Release Termination procedures to indicate to the MGW to remove the network side bearer termination and that the bearer can be released towards the preceding/succeeding MGW. After the resources in the MGW are released the MSC server sends the Release Complete message to the preceding/succeeding node (bullet 1 in figure 7.2).

7.1.2.2 Call clearing to the UE

The MSC server initiates call clearing towards the UE and requests release of the associated radio resources as described in 3GPP TS 23.108[18]. Once the call clearing and the release of the associated radio resources have been completed, the MSC server releases any MGW allocated resources for the access side. If any resources were seized in the MGW, the MSC server uses the Release Termination procedure to requests the MGW to remove the access side bearer termination (bullet 2 or bullet 3 in figure 7.2).

53

7.1.2.3 Example

Figure 7.1 shows the network model for a network initiated clearing of the mobile call. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling (not applicable in A/Gb mode for the A-interface) and the bearer. The MSC server seizes one context with two bearer terminations in the MGW. Bearer termination T1 is used for the bearer towards RNC/BSC and bearer termination T2 is used for the bearer towards succeeding MGW.



Figure 7.1: Network Initiated Call Clearing (Network model)

Figure 7.2 shows the message sequence example for the network initiated clearing of a mobile call. In the example the when the call clearing indication is received from the preceding/succeeding node, MSC server indicates that the network bearer can be released and to release the network side bearer termination. After the release of the network side bearer termination the MSC server indicates to the preceding/succeeding node that call clearing has been completed. The MSC server initiates call clearing towards the UE and requests release of the radio resource. After the response of the radio resource release is received then the MSC server requests release of the access side bearer termination.



Figure 7.2: Network Initiated Call Clearing (message sequence chart)

7.2 User Initiated

The user initiated call clearing shall be performed in accordance with 3GPP TS 23.108 [18]. The following clauses describe the additional requirements for the bearer independent CS core network.

55

7.2.1 Void

7.2.2 MSC server

7.2.2.1 all clearing from the UE

The UE initiated call clearing is performed and the release of the associated radio resources is performed as described in 3GPP TS 23.108 [18]. Once the call clearing and the associated radio resources release have been completed, the MSC server releases any MGW allocated resources for the access side. If any resources were seized in the MGW the MSC server uses the Release Termination procedure to requests the MGW to remove the access side bearer termination (bullet 1 or bullet 2 in figure 7.4).

7.2.2.2 Call clearing to the network side

The MSC server sends the Release message to the preceding/succeeding node. Once the preceding/succeeding node has sent the Release Complete, the MSC server releases any MGW allocated resources for the network side. If any resources were seized in the MGW server shall use the Release Bearer and Release Termination procedures to indicate to the MGW to remove the network side bearer termination and that the bearer can be released towards the preceding/succeeding MGW (bullet 3 in figure 7.4).

7.2.2.3 Example

Figure 7.3 shows the network model for a user initiated clearing of a mobile call. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling (not applicable in A/Gb mode for the A-interface) and the bearer. The MSC server seizes one context with two bearer terminations in the MGW. Bearer termination T1 is used for the bearer towards RNC/BSC and bearer termination T2 is used for the bearer towards succeeding MGW.



Figure 7.3: User Initiated Call Clearing (Network model)

Figure 7.4 shows the message sequence example for the user initiated clearing of a mobile call. In the example the UE initiates call clearing towards the MSC server and the MSC server requests release of the radio resource. After the response of the radio resource release is received the MSC server requests the release of the access side bearer termination. The MSC server initiates call clearing towards the preceding/succeeding node. Once the preceding/succeeding node has indicated that call clearing has been completed, the MSC server indicates that the network bearer can be released and to release the network side bearer termination.



Figure 7.4: User Initiated Call Clearing (message sequence chart)

7.3 (G)MSC server Initiated

The following clauses describe the additional requirements for (G)MSC server initiated call clearing in the bearer independent CS core network.

7.3.1 GMSC server

7.3.1.1 Call clearing to the destination side

If the call is already established towards the destination, call clearing is performed as described in clause 7.1.1, call clearing to the outgoing side.

7.3.1.2 Call clearing to the originating side

The call clearing to the originating side is performed as described in clause 7.1.1, call clearing to the outgoing side.

57

7.3.2 MSC server

7.3.2.1 Call clearing to the UE

The call clearing to the UE is performed as described in clause 7.1.2, call clearing to the UE (bullet 1 and bullet 2 in figure 7.6).

7.3.2.2 Call clearing to the network side

If the call is already established towards the network side, the call clearing to the network side is performed as described in clause 7.2.2, call clearing to the network side (bullet 3 in figure 7.6).

7.3.2.3 Example

Figure 7.5 shows the network model for the MSC server initiated clearing of the mobile call. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling (not applicable in A/Gb mode for the A-interface) and the bearer. The MSC server seizes one context with two bearer terminations in the MGW. Bearer termination T1 is used for the bearer towards RNC/BSC and bearer termination T2 is used for the bearer towards succeeding MGW.



Figure 7.5: MSC Server Initiated Call Clearing (Network model)

Figure 7.6 shows the message sequence example for the MSC server initiated clearing of a mobile call. In the example the MSC server initiates call clearing of the network side and the access side. After the call clearing towards the UE and the release of the radio resource have been completed the MSC server requests release of the access side bearer termination. Once the preceding/succeeding node has indicated that call clearing has been completed, the MSC server indicates that the network bearer can be released and to release the network side bearer termination.



Figure 7.6: MSC Server Initiated Call Clearing (message sequence chart)

7.4 MGW Initiated

The following clauses describe the additional requirements for MGW initiated call clearing in the bearer independent CS core network.

7.4.1 GMSC server

7.4.1.1 Bearer released on the destination side

After the GMSC server has received the Bearer Released procedure from the MGW, it shall send the Release message to the succeeding node. Once the succeeding node has sent the Release Complete message, the GMSC server releases any MGW allocated resources for the destination side. The GMSC server uses the Release Termination procedure to request the MGW to remove the destination side bearer termination.

The call clearing to the incoming side is performed as described in clause 7.1.1.2, call clearing to the outgoing side.

7.4.1.2 Bearer released on the originating side

After the GMSC server has received the Bearer Released procedure from the MGW, the GMSC server sends the Release message to the preceding node. Once the preceding node has sent the Release Complete message, the GMSC server releases any MGW allocated resources for the originating side. The GMSC server uses the Release Termination procedure to request the MGW to remove the originating side bearer termination.

If the call is already established towards the destination side, call clearing to the destination side is performed as described in clause 7.1.1.2, call clearing to the outgoing side.

7.4.2 MSC server

7.4.2.1 Bearer released on the access side

After the MSC server has received the Bearer Released procedure from the MGW, the MSC server initiates the call clearing towards the UE and requests release of the allocated radio resources as described in 3GPP TS 23.108 [18]. Once the call clearing and the radio resources release have been completed, the MSC server releases any MGW allocated resources for the access side. The MSC server uses the Release Termination procedure to request the MGW to remove the access side bearer termination.

If the call is already established towards the network side, call clearing to the network side is performed as described in clause 7.2, call clearing to the network side.

7.4.2.2 Bearer released on the network side

After the MSC server has received the Bearer Released procedure from the MGW, the MSC server sends the Release message to the preceding/succeeding node. Once the preceding/succeeding node has sent the Release Complete message, the MSC server releases any MGW allocated resources for the network side. The MSC server uses the Release Termination procedure to request the MGW to remove the network side bearer termination (bullet 1 and bullet 2 in figure 7.8).

Call clearing to the UE is performed as described in clause 7.1.2.2, call clearing to the UE (bullet 3 in figure 7.8).

7.4.2.3 Example

Figure 7.7 shows the network model for an MGW initiated clearing of a mobile call. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling (not applicable in A/Gb mode for the A-interface) and the bearer. The MSC server seizes one context with two bearer terminations in the MGW. Bearer termination T1 is used for the bearer towards RNC/BSC and bearer termination T2 is used for the bearer towards succeeding MGW.



Figure 7.7: MGW Initiated Call Clearing (Network model)

Figure 7.8 shows the message sequence example for the MGW initiated clearing of a mobile call. After the MSC server is notified that the MGW has released the network side bearer, the MSC server initiates call clearing of the network side and the access side. After the call clearing towards the UE and the radio resource release have been completed the MSC server requests release of the access side bearer termination. Once the preceding/succeeding node has indicated that call clearing has been completed, the MSC server requests the release of the network side bearer termination.



Figure 7.8: MGW Initiated Call Clearing (message sequence chart)

7.5 Call Clearing for Iu Interface on IP

Procedures for Call Clearing where the Iu Interface is on IP are as described in sections 7.1 to 7.4, with the exception that only the Access side procedures apply and the Release Bearer procedure is not sent. For Iu Interface on IP, the Release Termination procedures for IP are used to clear the MGW termination from the (G)MSC.

Example

Figure 7.9 shows the network model for a network initiated clearing of the mobile call when IP transport is used on the Iu interface. The 'squared' line represents the call control signalling. The 'dotted' line represents the bearer. The MSC server releases one context with two bearer terminations in the MGW. Bearer termination T1 is used for the bearer towards RNC/BSC and bearer termination T2 is used for the bearer towards succeeding MGW.





Figure 7.10 shows the message sequence example for the network initiated clearing of a mobile call when IP transport is used on the Iu Interface. In the example when the call clearing indication is received from the preceding/succeeding node, MSC server indicates that network bearer can be released and to release the network side bearer termination. After the release of the network side bearer termination the MSC server indicates to the preceding/succeeding node that call clearing has been completed. The MSC server initiates call clearing towards the UE and requests release of the radio resource. After the response of the radio resource release is received then the MSC server requests release of the access side bearer termination.



Figure 7.10 Network Initiated Call Clearing (message sequence chart)

8 Handover/Relocation

NOTE: All message sequence charts in this clause are examples. All valid handover/relocation message sequences can be derived from the example message sequences and associated message pre-conditions.

8.1 UMTS to UMTS

In the context of the following clauses, the terms RNS or RNC refer also to a GERAN BSS or BSC (respectively) when serving an UE in Iu mode.

8.1.1 Intra-MSC SRNS/SBSS Relocation

The procedures specified in 3GPP TS 23.009 [8] for "Intra-3G_MSC SRNS Relocation" shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network.

8.1.1.1 Relocation Required

When the Relocation Required message is received, the MSC server requests the MGW to provide a binding reference and a bearer address, using the Prepare Bearer procedure. For speech calls, the MSC server shall provide the MGW with the speech coding information for the bearer. For non-speech calls the MSC server also provides the MGW with the same PLMN Bearer Capability [4] as was provided at the last access bearer assignment. The MSC server uses the Change Flow Direction Procedure to request the MGW to set the Handover Device to initial state. The MSC server sends the Relocation Request message, containing the bearer address and the binding reference, to RNC-B (bullet 1 in figure 8.2/1).

For Relocation towards GERAN Iu mode the GERAN capabilities of the target cell will be indicated to the MSC-Server within the RANAP RELOCATION REQUIRED message if the target cell provides different capabilities than the current cell. If no information about the GERAN capabilities of the target cell are received within this message, the MSC-Server shall assume that the GERAN target cell will provide the same capabilities as the current cell (for details see [29]). The MSC server shall indicate to GERAN the selected services within the RANAP RELOCATION REQUEST message. The MSC server shall not set codec information in the NAS Synchronisation Indicator (see [4]). Instead it shall set codec information in the GERAN BSC container.

8.1.1.2 Relocation Command/Relocation Detect

When the MSC server sends the Relocation Command message or alternatively if it receives the Relocation Detect message, the MSC server uses the Change Flow Direction procedure to request the MGW to set the Handover Device to intermediate state (bullet 2 in figure 8.2/1).

8.1.1.3 Relocation Complete

When the MSC server receives the Relocation Complete message, it requests RNC-A to release the IU. The MSC server also requests the MGW to set the Handover Device to its final state by removing the bearer termination towards RNC-A, using the Release Termination procedure (bullet 3 in figure 8.2/2).

8.1.1.4 Interworking function

The interworking function used by the MGW before relocation will also be used after relocation.

8.1.1.5 Codec handling

The MGW may include a speech transcoder based upon the speech coding information provided to each bearer termination.

8.1.1.6 Voice Processing function

After relocation, the MGW may continue or modify voice-processing function(s) provided to each bearer termination.

8.1.1.7 Handling of multiple bearers (multicall)

If the UE is engaged with multiple bearers all procedures related to the handling of bearers and terminations described for the relocation of a single bearer shall be repeated for each bearer.

8.1.1.8 Failure Handling in MSC server

When a procedure between the MSC server and the MGW fails the MSC server shall handle the failure as an internal error in accordance with 3GPP TS 23.009 [8] and 3GPP TS 29.010 [23]. If MGW resources have been already seized at the target access side then the resources shall be released using the Release Termination procedure. If the call is to be cleared, then it shall be handled as described in clause 7.3.

8.1.1.9 Example

Figure 8.1 shows the network model for the Intra-MSC SRNS Relocation. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling and the bearer. The bearer termination T1 is used

for the bearer towards RNC-A, bearer termination T3 is used for the bearer towards RNC-B and the bearer termination T2 is used for the bearer towards the succeeding/preceding MGW.



Before Relocation:



During Relocation:



After Relocation:



Figure 8.2 shows the message sequence example for the Intra-MSC SRNS Relocation.

It is assumed that the Handover Device is located in the MGW, which has been selected for the call establishment by the MSC server. The MSC server controls the call and the mobility management. It is also assumed that only one bearer has been established towards RNC-A.

In the example the MSC server requests seizure of RNC-B side bearer termination with specific flow directions. The MSC server orders the establishment of the bearer by sending Relocation Request towards RNC-B. When the relocation is detected in RNC-B the MSC server requests to change the flow directions between the terminations within the context. When the MSC server receives a Relocation Complete indication from RNC-B it orders RNC-A to release the

IU. This action causes release of the bearer between the RNC and the MGW. Finally the MSC server requests the MGW to release RNC-A side bearer termination.



Figure 8.2/1: Intra-MSC SRNS Relocation (message sequence chart)



Figure 8.2/2: Intra-MSC SRNS Relocation (message sequence chart)

8.1.2 Basic Inter-MSC SRNS/SBSS Relocation

The procedures specified in 3GPP TS 23.009 [8] for "Basic Relocation Procedure Requiring a Circuit Connection between 3G_MSC-A and 3G_MSC-B" shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network.

8.1.2.1 MSC-A/MGW-A

8.1.2.1.1 Relocation Required

For Relocation towards GERAN Iu mode the GERAN capabilities of the target cell will be indicated to the MSC-A server within the RANAP RELOCATION REQUIRED message if the target cell provides different capabilities than the current cell. If no information about the GERAN capabilities of the target cell are received within this message, the MSC-A server shall assume that the GERAN target cell will provide the same capabilities as the current cell (for details see [29]).

The MSC-A server shall indicate to the MSC-B server the GERAN capabilities of the target cell, if available, with the MAP Prepare Handover request. For speech calls, the MSC-A server shall additionally indicate to the MSC-B server the selected codec, the list of supported codecs, and the currently used codec.

Furthermore, the MSC-A server shall indicate to the GERAN the selected service within the RANAP RELOCATION REQUEST message and shall set the RAB parameters within the RANAP RELOCATION REQUEST message according to the selected service. The MSC server shall not set codec information in the NAS Synchronisation Indicator (see [4]). Instead it shall set codec information in the GERAN BSC container..

8.1.2.1.2 Bearer establishment between MGW-A and MGW-B

The handling of the bearer establishment is as described for a Basic Mobile Originating Call, using either forward or backward bearer establishment. For speech calls, the MSC-A server shall provide the MGW-A with the speech coding information for the bearer. For non-speech calls, the MSC-A server shall provide MGW-A with the same PLMN Bearer Capability [4] as was provided at the last access bearer assignment. The MSC-A server also uses the Change Flow Direction procedure to request MGW-A to set the Handover Device to initial state (bullet 3 in figure 8.4/1).

8.1.2.1.3 Relocation Command/Relocation Detect

When the MSC-A server sends the Relocation Command message or alternatively if it receives the Relocation Detect message, the MSC-A server uses the Change Flow Direction procedure to requests MGW-A to set the Handover Device to intermediate state (bullet 4 in figure 8.4/2).

8.1.2.1.4 Relocation Complete

When the MSC-A server receives the Relocation Complete message, it requests RNC-A to release the IU. The MSC-A server also requests MGW-A to set the Handover Device to its final state by removing the bearer termination towards RNC-A, using the Release Termination procedure (bullet 5 in figure 8.4/2).

8.1.2.1.5 Interworking function

The interworking function used by MGW-A before relocation will also be used after relocation.

8.1.2.1.6 Codec handling

The MGW may include a speech transcoder based upon the speech coding information provided to each bearer termination.

8.1.2.1.7 Voice Processing function

Voice processing function(s) provided by MGW-A before relocation, may be modified or disabled by MGW-A after relocation.

8.1.2.1.8 Handling of multiple bearers (multicall)

If the UE is engaged with multiple bearers all procedures related to the handling of bearers and terminations described for the relocation of a single bearer shall be repeated for each bearer.

8.1.2.1.9 Failure Handling in MSC server

When a procedure between the MSC-A server and MGW-A fails the MSC-A server shall handle the failure as an internal error in accordance with 3GPP TS 23.009 [8] and 3GPP TS 29.010 [23]. If call establishment towards MSC-B has already started then the call towards MSC-B server shall be cleared as described in clause 7.3. If the original call is to be cleared, then it shall be handled as described in clause 7.3.

8.1.2.2 MSC-B/MGW-B

8.1.2.2.1 MGW selection

The MSC-B server selects an MGW when it receives Prepare Handover Request message (bullet 1 in figure 8.4/1).

8.1.2.2.2 Bearer establishment towards RNC-B

When the MSC-B server has selected MGW-B it requests MGW-B to provide a binding reference and a bearer address, using the Prepare Bearer procedure. For speech calls, the MSC-B server shall provide the MGW-B with the speech coding information for the bearer. The MSC-B server sends the Relocation Request message to the RNC-B containing the bearer addresses and binding references (bullet 2 in figure 8.4/1).

For Relocation towards GERAN Iu mode, if the selected service is speech and the MSC-B server cannot provide the codec requested by the MSC-A server, the MSC-B server shall select another codec according to the received GERAN capabilities of the target cell and the received list of supported codecs, and shall set the RAB parameters within the RANAP RELOCATION REQUEST message according to the new selected codec. Furthermore, the MSC-B server shall report the chosen codec and codec modes back to the MSC-A server with MAP Prepare Handover response.

8.1.2.2.3 Bearer establishment between MGW-A and MGW-B

The handling of the bearer establishment is as described at Basic Mobile Terminating Call, using either forward or backward bearer establishment.

8.1.2.2.4 Codec handling

The MGW may include a speech transcoder based upon the speech coding information provided to each bearer termination.

8.1.2.2.5 Voice Processing function

Voice processing function(s) provided by MGW-A before relocation, may be continued or modified by MGW-B after relocation.

8.1.2.2.6 Handling of multiple bearers (multicall)

If the UE is engaged with multiple bearers all procedures related to the handling of bearers and terminations described for the relocation of a single bearer shall be repeated for each bearer.

8.1.2.2.7 Failure Handling in MSC server

When a procedure between the MSC-B server and MGW-B fails the MSC-B server shall handle the failure as an internal error in accordance with 3GPP TS 23.009 [8] and 3GPP TS 29.010 [23]. If MGW-B resources have been already seized at the target access side then the resources shall be released using the Release Termination procedure. The call from MSC-A server shall be released as described at clause 7.1.

69

8.1.2.2.8 Example

Figure 8.3 shows the network model for the Basic Inter-MSC SRNS Relocation. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling and the bearer. In MGW-A the bearer termination T1 is used for the bearer towards RNC-A, bearer termination T3 is used for the bearer towards MGW-B, and the bearer termination T2 is used for the bearer towards the succeeding/preceding MGW. In MGW-B the bearer termination T4 is used for the bearer towards RNC-B, bearer termination T5 is used for the bearer towards MGW-A.



Before Relocation:



During Relocation:



After Relocation:

Figure 8.3: Basic Inter-MSC SRNS Relocation (network model)

Figure 8.4 shows the message sequence example for the Basic Inter-MSC SRNS Relocation. It is assumed that the Handover Device is located in the MGW (MGW-A) selected for the call establishment by the MSC server (MSC-A server) which controls the call and the mobility management. It is also assumed that only one bearer has been established towards RNC-A. In the example the MSC-B server requests MGW-B to seize an RNC-B side bearer. The MSC-B server orders the establishment of the bearer towards RNC-B by sending Relocation Request. The call is established between MSC-A and MSC-B servers, and the bearer is established between MGW-A and MGW-B. When the relocation is detected in RNC-B the MSC-A server requests to change the flow directions between the terminations within the context in MGW-A. When MSC-A server receives Relocation Complete indication from MSC-B server it orders RNC-A to release the IU. This action causes release of the bearer between RNC-A and MGW-A. Finally MSC-A server requests MGW-A to remove RNC-A side bearer termination.



Figure 8.4/1: Basic Inter-MSC SRNS Relocation (message sequence chart)



Figure 8.4/2: Basic Inter-MSC SRNS Relocation (message sequence chart)

8.1.3 Subsequent Inter-MSC SRNS/SBSS Relocation back to the Anchor MSC

The procedures specified in 3GPP TS 23.009 [8] for "Subsequent Relocation from 3G_MSC-B to 3G_MSC-A requiring a Circuit Connection between 3G_MSC-A and 3G_MSC-B" shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network.

8.1.3.1 MSC-A/MGW-A

8.1.3.1.1 Relocation Request

When the MSC-A server receives the MAP Prepare Subsequent Handover request containing a Relocation Request message, it requests MGW-A to provide a binding reference and a bearer address using the Prepare Bearer procedure. For speech calls, the MSC-A server shall provide the MGW-A with the speech coding information for the bearer. For non-speech calls the MSC-A server shall provide MGW-A with the same PLMN Bearer Capability [4] as was provided at the last access bearer assignment. The MSC-A server uses the Change Flow Direction procedure to request the MGW-A to set the Handover Device to initial state. The MSC-A server sends the Relocation Request message, containing the bearer address and the binding reference, to RNC-B (bullet 1 in figure 8.6/1).

For Relocation towards GERAN Iu mode, if the selected service is speech and the MSC-A server cannot provide the codec requested by the MSC-B server, the MSC-A server shall select another codec according to the received GERAN capabilities of the target cell and the list of supported codecs.

8.1.3.1.2 Relocation Command/Relocation Detect

When the MSC-A server sends the MAP Prepare Subsequent Handover response message or alternatively if it receives the Relocation Detect message, the MSC-A server uses the Change Flow Direction procedure to requests MGW-A to set the Handover Device to intermediate state (bullet 2 in figure 8.6/1).
8.1.3.1.3 Relocation Complete

When the MSC-A server receives the Relocation Complete message, it informs the MSC-B server about reception of this message. The MSC-A server then initiates call clearing towards the MSC-B server as described in clause 7.3.

8.1.3.1.4 Interworking function

The interworking function used by MGW-A before relocation will also be used after relocation.

8.1.3.1.5 Codec handling

The MGW may include a speech transcoder based upon the speech coding information provided to each bearer termination.

8.1.3.1.6 Voice Processing function

Voice processing function(s) provided by MGW-A and MGW-B before relocation, may be continued or modified by MGW-A after relocation.

8.1.3.1.7 Handling of multiple bearers (multicall)

If the UE is engaged with multiple bearers all procedures related to the handling of bearers and terminations described for the relocation of a single bearer shall be repeated for each bearer.

8.1.3.1.8 Failure Handling in MSC server

When a procedure between the MSC-A server and the MGW fails the MSC-A server shall handle the failure as an internal error in accordance with 3GPP TS 23.009 [8] and 3GPP TS 29.010 [23]. If MGW resources have already been seized at the target access side then the resources shall be released using the Release Termination procedure. If the call is to be cleared, then it shall be handled as described in clause 7.3.

8.1.3.2 MSC-B/MGW-B

8.1.3.2.1 Relocation Required

For Relocation towards GERAN Iu mode the GERAN capabilities of the target cell will be indicated to the MSC-B server within the RANAP RELOCATION REQUIRED message if the target cell provides different capabilities than the current cell. If no information about the GERAN capabilities of the target cell are received within this message, the MSC-B server shall assume that the GERAN target cell will provide the same capabilities as the current cell (for details see [29]). The different interworking scenarios (e.g. interworking to pre-Rel5 UTRAN) are described in [29].

The MSC-B server shall indicate to the MSC-A server the GERAN capabilities of the target cell, if available, with the MAP Prepare Subsequent Handover request. For speech calls, the MSC-B server shall additionally indicate to the MSC-A server the selected codec and the currently used codec.

Furthermore, the MSC-B server shall indicate to the GERAN the selected service within the RANAP RELOCATION REQUEST message and shall set the RAB parameters within the RANAP RELOCATION REQUEST message according to the selected service. The MSC server shall not set codec information in the NAS Synchronisation Indicator (see [4]). Instead it shall set codec information in the GERAN BSC container...

8.1.3.2.2 Relocation Complete

When the MSC-B server receives the Relocation Complete message, it requests RNC-A to release the IU. The MSC-B server requests MGW-B to remove the bearer termination towards RNC-A using the Release Termination procedure (bullet 3 in figure 8.6/2).

8.1.3.2.3 Release of bearer towards MGW-A

When the MSC-B server receives a call clearing indication from the MSC-A server, the MSC-B server handles it as described in clause 7.2.

8.1.3.2.4 Handling of multiple bearers (multicall)

If the UE is engaged with multiple bearers all procedures related to the handling of bearers and terminations described for the relocation of a single bearer shall be repeated for each bearer.

8.1.3.2.5 Example

Figure 8.5 shows the network model for the Subsequent Inter-MSC SRNS Relocation back to the Anchor MSC. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling and the bearer. In MGW-A the bearer termination T6 is used for the bearer towards RNC-B, bearer termination T3 is used for the bearer towards MGW-B, and the bearer termination T2 is used for the bearer towards the succeeding/preceding MGW. In MGW-B the bearer termination T4 is used for the bearer towards RNC-A, bearer termination T5 is used for the bearer towards MGW-A.



Before Relocation:



During Relocation:



After Relocation:

Figure 8.5: Subsequent Inter-MSC SRNS Relocation back to the Anchor MSC (network model)

Figure 8.6 shows the message sequence example for the Subsequent Inter-MSC SRNS Relocation back to the Anchor MSC. It is assumed that the Handover Device is located in the MGW (MGW-A) selected for the call establishment by the MSC server (MSC-A server) which controls the call and the mobility management. Also assumed that only one bearer has been established towards RNC-A. In the example the MSC-A server requests MGW-A to seize RNC-B side bearer termination with specific flow directions. The MSC server orders the establishment of the bearer towards RNC-B by sending Relocation Request. When the relocation is detected in RNC-B the MSC-A server requests to change the flow directions between the terminations within the context in MGW-A. When the MSC-A server receives a Relocation Complete indication from RNC-B it transfers this indication to MSC-B server. MSC-B server orders RNC-A to release the IU. This action causes release of the bearer between RNC-A and the MGW-B. MSC-A server initiates call clearing towards MSC-B server.



Figure 8.6/1: Subsequent Inter-MSC SRNS Relocation back to the Anchor MSC (message sequence chart)



Figure 8.6/2: Subsequent Inter-MSC SRNS Relocation back to the Anchor MSC (message sequence chart)

8.1.4 Subsequent Inter-MSC SRNS/SBSS Relocation to a third MSC

The relocation to a third MSC server (from MSC-B server to MSC-B server) is the combination of the two previous inter-MSC handover cases:

- for MSC-B server a subsequent relocation from MSC-B server back to MSC-A server as described in clause 8.1.3; and
- for MSC-B server a basic relocation from MSC-A server to MSC-B server as described in clause 8.1.2.

MSC-A server implements the corresponding parts of each handover case; i.e. access handling in MSC-A server is not included.

8.1.5 SRNS/SBSS Relocation with lu on IP

If IuCS on IP is supported by the MSC server, the Core Network side procedures described in 8.1.1, 8.1.2, 8.1.3 and 8.1.4 shall apply. For the access bearer termination, the exchange of IP addresses via call control procedures is described in this clause.

Before the MSC server starts the access bearer assignment, the MSC server requests the MGW to prepare for the access bearer using the Prepare_IP_Transport procedure. The MSC server requests the MGW to provide an IP Transport address and UDP Port and provides the MGW with the bearer characteristics. For speech calls, the MSC server shall provide the MGW with the speech coding information and conditionally GTT related information in accordance with 3GPP TS 23.226 [28] for the bearer. For a non-speech call the MSC server also provides the MGW with a PLMN Bearer Capability [4]. After the MGW has replied with the IP address and UDP Port the MSC server requests access bearer assignment using the provided IP address and UDP Port in accordance with 3GPP TS 25.413 [26]. The IP addresses and UDP Ports of the MGW and the RNC are exchanged via the RANAP procedures. If the bearer transport is IP and IuUP mode is Transparent, when the MSC receives the RANAP Iu Relocation Request response, it shall send the RNC IP address and UDP Port to the MGW Access bearer termination using the Modify_IP_Transport_Address procedure.

If the bearer transport is IP and IuUP mode is Support, the MGW shall use the source IP address and UDP Port of the IuUP Init packet received from the radio access network as the destination address for subsequent downlink packets.

3GPP TS 23.205 version 9.4.0 Release 9

76

The sequence is shown in figure 8.1.5/1.



Figure 8.1.5/1 SRNS Relocation with lu on IP

8.1.6 Intra-MSC enhanced SRNS Relocation

8.1.6.1 Introduction

The procedures specified in 3GPP TS 23.009 [8] for "Intra-3G_MSC SRNS Relocation" shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network. The support of the enhanced SRNS Relocation is optional.

8.1.6.2 Enhanced Relocation Complete Request

When the Enhanced Relocation Complete Request message is received, the MSC server requests the MGW to provide a binding reference and a bearer address, using the Prepare Bearer procedure. For speech calls, the MSC server shall provide the MGW with the speech coding information for the bearer. For non-speech calls the MSC server also provides the MGW with the same PLMN Bearer Capability [4] as was provided at the last access bearer assignment. The MSC server uses the Change Flow Direction Procedure to request the MGW to set the Handover Device to initial state. If required the MSC Server shall set-up an interconnection between the "anchor" and the "target" MGW. The MSC server then sends the Enhanced Relocation Complete Response message, containing the bearer address and the binding reference, to RNC-B (bullet 1 in figure 8.1.6/2).

8.1.6.3 Enhanced Relocation Complete Confirm

When the MSC server receives the Enhanced Relocation Complete Confirm message, the MSC server uses the Change Flow Direction procedure to request the MGW to set the Handover Device to intermediate state (bullet 2 in figure 8.1.6/1).

After that the MSC Server requests RNC-A to release the IU. The MSC server also requests the MGW to set the Handover Device to its final state by removing the bearer termination towards RNC-A, using the Release Termination procedure (bullet 3 in figure 8.1.6/2).

8.1.6.4 Interworking function

The interworking function used by the MGW before relocation will also be used after relocation.

8.1.6.5 Codec handling

The MGW may include a speech transcoder based upon the speech coding information provided to each bearer termination.

8.1.6.6 Voice Processing function

After relocation, the MGW may continue or modify voice-processing function(s) provided to each bearer termination.

8.1.6.7 Handling of multiple bearers (multicall)

If the UE is engaged with multiple bearers all procedures related to the handling of bearers and terminations described for the relocation of a single bearer shall be repeated for each bearer.

8.1.6.8 Failure Handling in MSC server

When a procedure between the MSC server and the MGW fails the MSC server shall send an IU-ENHANCED-RELOCATION-COMPLETE-FAILURE message to RNS-B in accordance with 3GPP TS 23.009 [8] and 3GPP TS 29.010 [23]. If MGW resources have been already seized at the target access side then the resources shall be released using the Release Termination procedure. If the call is to be cleared, then it shall be handled as described in clause 7.3.

8.1.6.9 Example

Figure 8.1.6/1 shows the network model for the Intra-MSC enhanced SRNS Relocation. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling and the bearer. The bearer termination T1 is used for the bearer towards RNC-A, bearer termination T3 is used for the bearer towards RNC-B and the bearer termination T2 is used for the bearer towards the succeeding/preceding MGW.



Before Relocation:



After Relocation:





Figure 8.1.6/1: Intra-MSC Enhanced SRNS Relocation (network model)

Figure 8.1.6/2 shows the message sequence example for the Intra-MSC enhanced SRNS Relocation. It is assumed that the Handover Device is located in the MGW, which has been selected for the call establishment by the MSC server. The MSC server controls the call and the mobility management. It is also assumed that only one bearer has been established towards RNC-A.

In the example the MSC server requests seizure of RNC-B side bearer termination with specific flow directions. The MSC server orders the establishment of the bearer by sending Enhanced Relocation Complete Response towards RNC-B. When the MSC server receives an Enhanced Relocation Complete Confirm indication from RNC-B the MSC server requests to change the flow directions between the terminations within the context. MSC Server also orders RNC-A to release the IU. This action causes release of the bearer between the RNC-A and the MGW. Finally the MSC server requests the MGW to release RNC-A side bearer termination.



Figure 8.1.6/2: Intra-MSC Enhanced SRNS Relocation (message sequence chart)

8.1.7 Intra-MSC enhanced SRNS Relocation with lu on IP

If IuCS on IP is supported by the MSC server, the Core Network side procedures described in 8.1.x shall apply. For the access bearer termination, the exchange of IP addresses via call control procedures as described in 8.1.5 applies with the following exception. The Prepare_IP_Transport procedure is used after the MSC Server receives the RANAP Enhanced Relocation Complete Request. If the bearer transport is IP and IuUP mode is Transparent, when the MSC Server receives the RANAP Enhanced Relocation Complete Response, it shall send the RNC IP address and UDP Port to the MGW Access bearer termination using the Modify_IP_Transport_Address procedure.

The sequence is shown in figure 8.7/1.



Figure 8.1.7/1 Enhanced SRNS Relocation with lu on IP

8.2 UMTS to GSM

In the context of the following clauses, the terms RNS or RNC refer also to a GERAN BSS or BSC (respectively) when serving an UE in Iu mode.

8.2.1 Intra-MSC UMTS to GSM Handover

The procedures specified in 3GPP TS 23.009 [8] for "Intra-3G_MSC Handover from UMTS to GSM" shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network.

8.2.1.1 Relocation Required

When the MSC server receives the Relocation Required message, it requests the MGW to seize a TDM circuit, using the Reserve Circuit procedure. For non-speech calls the MSC server shall provide the MGW with the GSM Channel coding properties and the same PLMN Bearer Capability [4] as was provided at the last access bearer assignment. The MSC server uses the Change Flow Direction procedure to request the MGW to set the Handover Device to initial state. The MSC server sends the Handover Request message, containing the CIC, to BSC-B (bullet 1 in figure 8.8/1).

8.2.1.2 Handover Request Acknowledge

For non-speech calls after receiving the Handover Request Acknowledge message if the assigned GSM Channel coding properties differ from the previously provided ones the MSC server provides the MGW with the assigned GSM Channel coding properties using the Modify Bearer Characteristics procedure (bullet 2 in figure 8.8/1).

8.2.1.3 Relocation Command/Handover Detect

When the MSC server sends the Relocation Command message or alternatively if it receives the Handover Detect message, the MSC server uses the Change Flow Direction procedure to requests the MGW to set the Handover Device to intermediate state (bullet 3 in figure 8.8/1).

8.2.1.4 Handover Complete

When the MSC server receives the Handover Complete message, it requests RNC-A to release the IU. The MSC server also requests the MGW to set the Handover Device to its final state by removing the bearer termination towards RNC-A, using the Release Termination procedure (bullet 4 in figure 8.8/2).

8.2.1.5 Interworking function

The interworking function used by the MGW before handover will also be used after handover.

8.2.1.6 Voice Processing function

After handover, the MGW may continue or modify voice processing function(s) provided to each bearer termination.

8.2.1.7 Handling of multiple bearers (multicall)

If the UE is engaged with multiple bearers then one bearer is selected to be handed over according to 3GPP TS 23.009 [8]. The calls carried by the bearers that have not been selected will be cleared after the reception of the Handover Complete message, as described in clause 7.3.

8.2.1.8 Failure Handling in MSC server

When a procedure between the MSC server and the MGW fails the MSC server shall handle the failure as an internal error in accordance with 3GPP TS 23.009 [8] and 3GPP TS 29.010 [23]. If MGW resources have been already seized at the target access side then the resources shall be released using the Release Termination procedure. If the call is to be cleared, then it shall be handled as described in clause 7.3.

8.2.1.9 Example

Figure 8.7 shows the network model for the Intra-MSC UMTS to GSM Handover. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling (not applicable in case of GSM access) and the bearer. The bearer termination T1 is used for the bearer towards RNC-A, bearer termination T3 is used for the bearer towards BSC-B and the bearer termination T2 is used for the bearer towards the succeeding/preceding MGW.



Before Handover



During Handover



After Handover

Figure 8.7: Intra-MSC UMTS to GSM Handover (network model)

Figure 8.8 shows the message sequence example for the Intra-MSC UMTS to GSM Handover. It is assumed that the Handover Device is located in the MGW selected for the call establishment by the MSC server, which controls the call and the mobility management. It is also assumed that only one bearer has been established towards RNC-A and that MGW-A is capable of handling GSM access.

In the example the MSC server requests seizure of BSC-B side bearer termination with specific flow directions. The MSC server starts handover execution by sending Handover Request towards BSC-B. When the handover is detected in BSC-B the MSC server requests to change the flow directions between the terminations within the context. When MSC server receives Handover Complete indication from BSC-B it orders RNC-A to release the IU. Finally the MSC server requests the MGW to release RNC-A side bearer termination.



Figure 8.8/1: Intra-MSC UMTS to GSM Handover (message sequence chart)



Figure 8.8/2: Intra-MSC UMTS to GSM Handover (message sequence chart)

8.2.1.10 Intra-MSC UMTS to GSM Handover for A interface over IP

When the MSC server receives the RANAP Relocation Required message, it requests the MGW to reserve an RTP bearer termination using the Reserve RTP Connection Point procedure. The MSC server requests the MGW to reserve an IP address and UDP port. The MSC server uses the Change Flow Direction procedure to request the MGW to set the Handover Device to initial state.

The MGW reserves the RTP termination and indicates the IP address and UDP port number to the MSC server. The IP addresses and UDP ports of the MGW is sent to the BSC in the BSSMAP Handover Request message. When the MSC server receives the BSSMAP Handover Request-Ack message, it shall send the BSC IP address and UDP Port number to the MGW using the Configure RTP Connection Point procedure.

The sequence is shown in Figure 8.2.1.10/1.



Figure 8.2.1.10/1: Intra-MSC UMTS to GSM Handover for AoIP (message sequence chart)

8.2.2 Basic Inter-MSC UMTS to GSM Handover

The procedures specified in 3GPP TS 23.009 [8] for "Basic Handover Procedure Requiring a Circuit Connection between 3G_MSC-A and MSC-B" shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network.

8.2.2.1 MSC-A/ MGW-A

8.2.2.1.1 Bearer establishment between MGW-A and MGW-B

The handling of the bearer establishment between MGW-A and MGW-B is as described for a Basic Mobile Originating Call, using either forward or backward bearer establishment. For non-speech calls the MSC-A server shall provide MGW-A with the GSM Channel coding properties and the same PLMN Bearer Capability [4] as was provided at the last access bearer assignment. The MSC-A server also uses the Change Flow Direction procedure to request MGW-A to set the Handover Device to initial state (bullet 3 in figure 8.10/1).

8.2.2.1.2 Relocation Command/Handover Detect

When the MSC-A server sends the Relocation Command message or alternatively if it receives the Handover Detect message, the MSC-A server uses the Change Flow Direction procedure to requests the MGW to set the Handover Device to intermediate state (bullet 2 in figure 8.10/1).

8.2.2.1.3 Handover Complete

When the MSC-A server receives the Relocation Complete message, it requests RNC-A to release the IU. The MSC-A also requests the MGW to set the Handover Device to its final state by removing the bearer termination towards RNC-A, using the Release Termination procedure (bullet 3 in figure 8.10/1).

8.2.2.1.4 Interworking function

The interworking function used by MGW-A before handover will also be used after handover.

8.2.2.1.5 Voice Processing function

Voice processing function(s) provided by MGW-A before handover, may be continued or modified by MGW-A and/or MGW-B after handover.

8.2.2.1.6 Handling of multiple bearers (multicall)

If the UE is engaged with multiple bearers then one bearer is selected to be handed over according to 3GPP TS 23.009 [8]. The calls carried by bearers that have not been selected will be cleared after the reception of Handover Complete message, as described in clause 7.3.

8.2.2.1.7 Failure Handling in MSC server

When a procedure between the MSC-A server and MGW-A fails the MSC-A server shall handle the failure as an internal error in accordance with 3GPP TS 23.009 [8] and 3GPP TS 29.010 [23]. If MGW-A resources have been already seized for the bearer towards MGW-B then the resources shall be released using the Release Termination procedure. The call towards MSC-B server shall be cleared as described in clause 7.3. If the original call is to be cleared, then it shall be handled as described in clause 7.3.

8.2.2.2 MSC-B/MGW-B

8.2.2.2.1 MGW selection

The MSC-B server selects an MGW when it receives Prepare Handover Request message (bullet 1 in figure 8.10/1).

8.2.2.2.2 Bearer establishment towards BSC-B

When the MSC-B server has selected MGW-B it requests MGW-B to seize a TDM circuit, using the Reserve Circuit procedure. The MSC-B server sends the Handover Request message to the BSC-B containing the CIC (bullet 2 in figure 8.10/1).

8.2.2.2.3 Bearer establishment between MGW-A and MGW-B

The handling of the bearer establishment between MGW-A and MGW-B is as described for a Basic Mobile Terminating Call, using either forward or backward bearer establishment.

8.2.2.2.4 Voice Processing function

Voice processing function(s) provided by MGW-A before handover, may be continued or modified by MGW-B after handover.

8.2.2.2.5 Failure Handling in MSC server

When a procedure between the MSC-B server and MGW-B fails the MSC-B server shall handle the failure as an internal error in accordance with 3GPP TS 23.009 [8] and 3GPP TS 29.010 [23]. If MGW-B resources have already been seized at the target access side then the resources shall be released using the Release Termination procedure. The call from MSC-A server shall be released as described at subclause 7.1.

8.2.2.2.6 Example

Figure 8.9 shows the network model for the Basic Inter-MSC UMTS to GSM Handover. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling (not applicable in case of GSM access) and the bearer. In MGW-A the bearer termination T1 is used for the bearer towards RNC-A, bearer termination T3 is used for the bearer towards MGW-B, and the bearer termination T2 is used for the bearer towards the succeeding/preceding MGW. In MGW-B the bearer termination T4 is used for the bearer towards BSC-B, bearer termination T5 is used for the bearer towards MGW-A.





After Handover

Figure 8.9: Basic Inter-MSC UMTS to GSM Handover (network model)

Figure 8.10 shows the message sequence example for the Basic Inter-MSC UMTS to GSM Handover. It is assumed that the Handover Device is located in the MGW (MGW-A) which has been selected for the call establishment by the MSC server (MSC-A server). The MSC server controls the call and the mobility management. It is also assumed that only one bearer has been established towards RNC-A.

In the example the MSC-B server requests MGW-B to seize BSC-B side bearer termination. The call is established between MSC-A server and MSC-B server, and the bearer is established between MGW-A and MGW-B. When the handover is detected in BSC-B the MSC-A server requests to change the flow directions between the terminations within the context in MGW-A. When MSC-A server receives Handover Complete indication from MSC-B server it orders RNC-A to release the IU. Finally MSC-A server requests MGW-A to remove RNC-A side bearer termination.



Figure 8.10/1: Basic Inter-MSC UMTS to GSM Handover (message sequence chart)



Figure 8.10/2: Basic Inter-MSC UMTS to GSM Handover (message sequence chart)

8.2.2.3 Basic Inter-MSC UMTS to GSM Handover for A Interface over IP

If AoIP is supported by the MSC server, the Core Network side procedures described earlier in 8.2.2 shall apply. For the access side termination, the exchange of IP addresses via call control procedures is described in this clause.

When the MSC-B server receives the MAP Prepare Handover Request., it requests the MGW-B to reserve an RTP bearer termination using the Reserve RTP Connection Point procedure. The MSC-B server requests the MGW-B to reserve an IP address and UDP port.

The MGW-B reserves the RTP termination and indicates the IP address and UDP port number to the MSC-B server. The IP addresses and UDP ports of the MGW is sent to the BSC-B in the BSSMAP Handover Request message. When the MSC-B server receives the BSSMAP Handover Request-Ack message, it shall send the BSC-B IP address and UDP Port number to the MGW-B using the Configure RTP Connection Point procedure.

3GPP TS 23.205 version 9.4.0 Release 9

90

The sequence is shown in Figure 8.2.2.3/1.





8.2.3 Subsequent Inter-MSC UMTS to GSM Handover back to the Anchor MSC

The following handling shall be applied for a call that started as UMTS call. The procedures specified in 3GPP TS 23.009 [8] for "Subsequent UMTS to GSM handover requiring a Circuit Connection between 3G_MSC-A and 3G_MSC-B, 3G_MSC-B to MSC-A" shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network.

8.2.3.1 MSC-A

8.2.3.1.1 Handover Request

When the MSC-A server receives the MAP Prepare Subsequent Handover request containing a Handover Request message, it requests MGW-A to seize a TDM circuit, using the Reserve Circuit procedure. For non-speech calls the MSC-A server shall provide MGW-A with the GSM Channel coding properties and the same PLMN Bearer Capability

[4] as was provided at the first access bearer assignment. The MSC-A server uses the Change Flow Direction procedure to request MGW-A to set the Handover Device to initial state. The MSC-A server sends the Handover Request message, containing the CIC, to BSC-B (bullet 1 in figure 8.12/1).

8.2.3.1.2 Handover Request Acknowledge

For non-speech calls after receiving the Handover Request Acknowledge message if the assigned GSM Channel coding properties differ from the previously provided ones the MSC-A server shall provide MGW-A with the assigned GSM Channel coding properties using the Modify Bearer Characteristics procedure (bullet 2 in figure 8.12/1).

8.2.3.1.3 Relocation Command/Handover Detect

When the MSC-A server sends the MAP Prepare Subsequent Handover response message or alternatively if it receives the Handover Detect message, the MSC-A server uses the Change Flow Direction procedure to requests MGW-A to set the Handover Device to intermediate state (bullet 3 in figure 8.12/2).

8.2.3.1.4 Handover Complete

When the MSC-A server receives the Handover Complete message, it informs the MSC-B server about reception of this message (bullet 3 in figure 8.12/2). The MSC-A server then initiates call clearing towards the MSC-B server as described at 7.3.

8.2.3.1.5 Interworking function

The interworking function used by MGW-A before handover will also be used after handover.

8.2.3.1.6 Voice Processing function

Voice processing function(s) provided by MGW-A and MGW-B before handover, may be continued or modified by MGW-A after handover.

8.2.3.1.7 Handling of multiple bearers (multicall)

If the UE is engaged with multiple bearers the selected bearer to be handed over is received from MSC-B server in the Handover Request message according to 3GPP TS 23.009 [8]. The calls carried by the bearers that have not been selected will be cleared by MSC-A server after the reception of the Handover Complete message, as described in clause 7.3.

8.2.3.1.8 Failure Handling in MSC server

When a procedure between the MSC-A server and the MGW fails the MSC-A server shall handle the failure as an internal error in accordance with 3GPP TS 23.009 [8] and 3GPP TS 29.010 [23]. If MGW resources have already been seized at the target access side then the resources shall be released using the Release Termination procedure. If the call is to be cleared, then it shall be handled as described in clause 7.3.

8.2.3.2 MSC-B

8.2.3.2.1 Handover Complete

When the MSC-B server receives the Handover Complete message, it requests RNC-A to release the IU. The MSC-B server also requests MGW-B to remove the bearer termination towards RNC-A using the Release Termination procedure (bullet 4 in figure 8.12/2).

8.2.3.2.2 Release of bearer towards MGW-A

When the MSC-B server receives a call clearing indication from the MSC-A server, the MSC-B server handles it as described in clause 7.2.

8.2.3.2.3 Handling of multiple bearers (multicall)

If the UE is engaged with multiple bearers then one bearer is selected by MSC-B server to be handed over according to 3GPP TS 23.009 [8].

8.2.3.2.4 Example

Figure 8.11 shows the network model for the Subsequent Inter-MSC UMTS to GSM handover back to the Anchor MSC. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling (not applicable in case of GSM access) and the bearer. In MGW-A the bearer termination T6 is used for the bearer towards BSC-B, bearer termination T3 is used for the bearer towards MGW-B, and the bearer termination T2 is used for the bearer towards the succeeding/preceding MGW. In MGW-B the bearer termination T4 is used for the bearer towards RNC-A, bearer termination T5 is used for the bearer towards MGW-A.



During Handover



After Handover

Figure 8.11: Subsequent Inter-MSC UMTS to GSM Handover back to the Anchor MSC (network model)

Figure 8.12 shows the message sequence example for the Subsequent Inter-MSC UMTS to GSM Handover back to the Anchor MSC. It is assumed that the Handover Device is located in the MGW (MGW-A) selected for the call establishment by the MSC server (MSC-A server) which controls the call and the mobility management. Also assumed that only one bearer has been established towards RNC-A and that MGW-A is capable to handle GSM access. In the example the MSC-A server requests MGW-A to seize BSC-B side bearer termination with specific flow directions. When the relocation is detected in BSC-B the MSC-A server requests to change the flow directions between the terminations within the context in MGW-A. When MSC-A server receives Handover Complete indication from BSC-B it transfers this indication to MSC-B server. MSC-B server orders RNC-A to release the IU. MSC-A server initiates call clearing towards MSC-B server.



Figure 8.12/1: Subsequent Inter-MSC UMTS to GSM Handover back to the Anchor MSC (message sequence chart)



Figure 8.12/2: Subsequent Inter-MSC UMTS to GSM Handover back to the Anchor MSC (message sequence chart)

8.2.3.3 Subsequent Inter-MSC UMTS to GSM Handover back to the anchor MSC for A Interface over IP

When the MSC-A server receives the MAP Prepare Subsequent Handover Request. message, it requests the MGW-A to reserve an RTP bearer termination using the Reserve RTP Connection Point procedure. The MSC-A server requests the MGW-A to reserve an IP address and UDP port. The MSC-A server uses the Change Flow Direction procedure to request the MGW-A to set the Handover Device to initial state.

The MGW-A reserves the RTP termination and indicates the IP address and UDP port number to the MSC-A server. The IP addresses and UDP ports of the MGW-A is sent to the BSC-B in the BSSMAP Handover Request message. When the MSC-A server receives the BSSMAP Handover Request-Ack message, it shall send the BSC-A IP address and UDP Port number to the MGW-B using the Configure RTP Connection Point procedure.

The sequence is shown in Figure 8.2.3.3/1.



Figure 8.2.3.c.1: Subsequent Inter-MSC UMTS to GSM Handover back to the anchor MSC for AoIP (message sequence chart)

8.2.4 Subsequent Inter-MSC UMTS to GSM Handover to a third MSC

The UMTS to GSM handover to a third MSC server (from MSC-B server to MSC-B server) is the combination of the two previous inter-MSC handover cases:

- for MSC-B server a subsequent UMTS to GSM handover from MSC-B server back to MSC-A server as described in clause 8.2.3; and
- for MSC-B server a basic UMTS to GSM handover from MSC-A server to MSC-B server as described in clause 8.2.2.

MSC-A server implements the corresponding parts of each handover case; i.e. access handling in MSC-A server is not included.

8.3 GSM to UMTS

In the context of the following clauses, the terms RNS or RNC refer also to a GERAN BSS or BSC (respectively) when serving an UE in Iu mode.

8.3.1 Intra-MSC GSM to UMTS Handover

The procedures specified in 3GPP TS 23.009 [8] for "Intra-3G_MSC GSM to UMTS Handover" shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network.

8.3.1.1 Handover Required

When the MSC server receives the Handover Required message, it requests the MGW to provide a binding reference and a bearer address using the Prepare Bearer procedure. For speech calls, the MSC server shall provide the MGW with the speech coding information for the bearer. For non-speech calls the MSC server shall provide the MGW with the same PLMN Bearer Capability [4] as was provided at the last channel assignment. The MSC server uses the Change Flow Direction procedure to request the MGW to set the Handover Device to initial state. The MSC server sends the Relocation Request message to the RNC-B containing the bearer address and binding reference (bullet 1 in figure 8.14).

For Relocation towards GERAN Iu mode the GERAN capabilities of the target cell will be indicated to the MSC-Server within the Handover Required message if the target cell provides different capabilities than the current cell. If no information about the GERAN capabilities of the target cell are received within this message, the MSC-Server shall assume that the GERAN target cell will provide the same capabilities as the current cell (for details see [29]). The MSC server shall indicate to GERAN the selected services within the RANAP RELOCATION REQUEST message. The MSC server shall not set codec information in the NAS Synchronisation Indicator (see [4]). Instead it shall set codec information in the GERAN BSC container.

8.3.1.2 Handover Command/Relocation Detect

When the MSC server sends the Handover Command message or alternatively if it receives a Relocation Detect message, the MSC server uses the Change Flow Direction procedure to requests the MGW to set the Handover Device to intermediate state (bullet 2 in figure 8.14).

8.3.1.3 Relocation Complete

When the MSC server receives the Relocation Complete message, it releases the A-interface line towards BSC-A and requests the MGW to set the Handover Device to its final state removing the bearer termination towards BSC-A, using Release Termination procedure (bullet 3 in figure 8.14).

8.3.1.4 Interworking function

The interworking function used by the MGW before handover will also be used after handover.

8.3.1.5 Codec handling

The MGW may include a speech transcoder based upon the speech coding information provided to each bearer termination.

8.3.1.6 Voice Processing function

After handover, the MGW may continue or modify voice processing function(s) provided to each bearer termination.

8.3.1.7 Failure Handling in MSC server

When a procedure between the MSC server and the MGW fails the MSC server shall handle the failure as an internal error in accordance with 3GPP TS 23.009 [8] and 3GPP TS 29.010 [23]. If MGW resources have already been seized at the target access side then the resources shall be released using the Release Termination procedure. If the call is to be cleared, then it shall be handled as described in clause 7.3.

8.3.1.8 Example

Figure 8.13 shows the network model for the Intra-3G_MSC GSM to UMTS Handover. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling and the bearer. The bearer termination T1 is used for the bearer towards the BSC-A (connected through the MSC server), the bearer termination

T3 is used for the bearer towards the RNC-B and the bearer termination T2 is used for the bearer towards the succeeding/preceding MGW.



Before Handover



During Handover



After Handover

Figure 8.13: Intra-3G_MSC GSM to UMTS Handover (network model)

Figure 8.14 shows the message sequence example for the Intra-MSC GSM to UMTS Handover. It is assumed that the Handover Device is located in the MGW selected for the call establishment by the MSC server, which controls the call and the mobility management. In the example the MSC server requests seizure of RNC-B side bearer termination with specific flow directions. The MSC server starts handover execution by sending Relocation Request towards RNC-B. When the relocation is detected in RNC-B the MSC server requests to change the flow directions between the terminations within the context. When MSC server receives Relocation Complete indication from RNC-B it releases the A-interface line towards the BSC-A. Finally the MSC server requests the MGW to release BSC-A side bearer termination.





8.3.2 Basic Inter-MSC GSM to UMTS Handover

The following handling shall be applied for a call that started as UMTS call. The procedures specified in 3GPP TS 23.009 [8] for "Basic Handover Procedure Requiring a Circuit Connection between MSC-A and 3G_MSC-B" shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network.

8.3.2.1 MSC-A

8.3.2.1.1 Handover Required

For Handover towards GERAN Iu mode the GERAN capabilities of the target cell will be indicated to the MSC-A server within the Handover Required message if the target cell provides different capabilities than the current cell. If no information about the GERAN capabilities of the target cell are received within this message, the MSC-A server shall assume that the GERAN target cell will provide the same capabilities as the current cell (for details see [29]).

The MSC-A server shall indicate to the MSC-B server the GERAN capabilities of the target cell, if available, with the MAP Prepare Handover request. For speech calls, the MSC-A server shall additionally indicate to the MSC-B server the selected codec, the list of supported codecs, and the currently used codec.

8.3.2.1.2 Bearer establishment between MGW-A and MGW-B

The handling of the bearer establishment between MGW-A and MGW-B is as described for a Basic Mobile Originating Call, using either forward or backward bearer establishment. For speech calls, the MSC-A server shall provide the MGW-A with the speech coding information for the bearer. For non-speech calls the MSC-A server shall provide MGW-A with the same PLMN Bearer Capabilities [4] as were provided at the last access bearer assignment. The MSC-A server also uses the Change Flow Direction procedure to request MGW-A to set the Handover Device to initial state (bullet 3 in figure 8.16/1).

8.3.2.1.3 Handover Command/Handover Detect

When the MSC-A server sends the Handover Command message or alternatively if it receives the Handover Detect message, the MSC-A server uses the Change Flow Direction procedure to requests MGW-A to set the Handover Device to intermediate state (bullet 4 in figure 8.16/2).

8.3.2.1.4 Handover Complete

When the MSC-A server receives the Handover Complete message, it releases the A-interface line towards BSC-A and requests MGW-A to set the Handover Device to its final state by removing the bearer termination towards BSC-A, using Release Termination procedure (bullet 5 in figure 8.16/2).

8.3.2.1.5 Interworking function

The interworking function used by MGW-A before handover will also be used after handover.

8.3.2.1.6 Codec handling

The MGW may include a speech transcoder based upon the speech coding information provided to each bearer termination.

8.3.2.1.7 Voice Processing function

Voice processing function(s) provided by MGW-A before handover, may be modified or disabled by MGW-A after handover.

8.3.2.1.8 Failure Handling in MSC server

When a procedure between the MSC-A server and MGW-A fails the MSC-A server shall handle the failure as an internal error in accordance with 3GPP TS 23.009 [8] and 3GPP TS 29.010 [23]. If the call establishment towards MSC-B has already started then the call towards MSC-B server shall be cleared as described in clause 7.3. If the original call is to be cleared, then it shall be handled as described in clause 7.3.

8.3.2.2 MSC-B

8.3.2.2.1 MGW selection

The MSC-B server selects an MGW when it receives Prepare Handover Request message (bullet 1 in figure 8.16).

8.3.2.2.2 Bearer establishment towards RNC-B

When the MSC-B server has selected MGW-B it requests MGW-B to provide a binding reference and a bearer address using the Prepare Bearer procedure. For speech calls, the MSC server shall provide the MGW with the speech coding information for the bearer. The MSC-B server sends the Relocation Request message to the RNC-B containing the bearer address and binding reference (bullet 2 in figure 8.16).

For Handover towards GERAN Iu-mode, the MSC-B Server shall select a service according to the Channel Type received with the Handover Request message and the capabilities of the GERAN target cell, if the GERAN classmark was received. For speech calls, the MSC-B server shall additionally take into account the selected codec, the list of supported codecs and the currently used codec received with MAP Prepare Handover request. The list of permitted speech versions received with the Channel Type in the Handover Request message is applicable to GERAN A/Gb mode only.

The MSC-B server shall indicate to the GERAN the selected service within the RANAP RELOCATION REQUEST message and shall set the RAB parameters within the RANAP RELOCATION REQUEST message according to the selected service. The MSC server shall not set codec information in the NAS Synchronisation Indicator (see [4]). Instead it shall set codec information in the GERAN BSC container..

For speech calls, the MSC-B server shall report the chosen codec and codec modes back to the MSC-A server with MAP Prepare Handover response.

8.3.2.2.3 Bearer establishment between MGW-A and MGW-B

The handling of the bearer establishment is as described at Basic Mobile Terminating Call, using either forward or backward bearer establishment.

8.3.2.2.4 Codec handling

The MGW may include a speech transcoder based upon the speech coding information provided to each bearer termination.

8.3.2.2.5 Voice Processing function

Voice processing function(s) provided by MGW-A before handover, may be continued or modified by MGW-B after handover.

8.3.2.2.6 Failure Handling in MSC server

When a procedure between the MSC-B server and MGW-B fails the MSC-B server shall handle the failure as an internal error in accordance with 3GPP TS 23.009 [8] and 3GPP TS 29.010 [23]. If MGW-B resources have already been seized at the target access side then the resources shall be released using the Release Termination procedure. The call from MSC-A server shall be released as described at clause 7.1.

8.3.2.2.7 Example

Figure 8.15 shows the network model for the Basic Inter-MSC GSM to UMTS Handover. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling (not applicable in case of GSM access) and the bearer. In MGW-A the bearer termination T1 is used for the bearer towards BSC-A, bearer termination T3 is used for the bearer towards MGW-B, and the bearer termination T2 is used for the bearer towards the succeeding/preceding MGW. In MGW-B the bearer termination T4 is used for the bearer towards RNC-B, bearer termination T5 is used for the bearer towards MGW-A.



After Handover

Figure 8.15: Basic Inter-MSC GSM to UMTS Handover (network model)

Figure 8.16 shows the message sequence example for the Basic Inter-MSC GSM to UMTS Handover. It is assumed that the Handover Device is located in the MGW (MGW-A) selected for the call establishment by the MSC server (MSC-A server) which controls the call and the mobility management.

In the example the MSC-B server requests MGW-B to seize RNC-B side bearer termination. The call is established between MSC-A server and MSC-B server, and the bearer is established between MGW-A and MGW-B. When the relocation is detected in RNC-B the MSC-A server requests to change the flow directions between the terminations within the context in MGW-A. When MSC-A server receives Handover Complete indication from MSC-B server it releases the A-interface line towards the BSC-A. Finally MSC-A server requests MGW-A to remove BSC-A side bearer termination.



Figure 8.16/1: Basic Inter-MSC GSM to UMTS Handover (message sequence chart)





8.3.3 Subsequent Inter-MSC GSM to UMTS Handover back to the Anchor MSC

The following handling shall be applied for a call that started as UMTS call. The procedures specified in 3GPP TS 23.009 [8] for "Subsequent Inter-MSC GSM to UMTS Handover back to the Anchor MSC" shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network.

8.3.3.1 MSC-A

8.3.3.1.1 Handover Request

When the MSC-A server receives a MAP Prepare Subsequent Handover request containing a Handover Request message, it requests the MGW-A to provide a binding reference and a bearer address using the Prepare Bearer procedure. For speech calls, the MSC-A server shall provide the MGW-A with the speech coding information for the bearer. For non-speech calls the MSC-A server shall provide MGW-A with the same PLMN Bearer Capability [4] as

was provided at the last channel assignment. The MSC-A server uses the Change Flow Direction Procedure to request the MGW-A to set the Handover Device to initial state. The MSC-A server sends the Relocation Request message to the RNC-B containing the bearer address and binding reference (bullet 1 in figure 8.18/1).

For Handover towards GERAN Iu-mode, the MSC-A Server shall select a service according to the Channel Type received with the Handover Request message and the capabilities of the GERAN target cell, if the GERAN classmark was received. For speech calls, the MSC-A server shall additionally take into account the selected codec and the currently used codec received with MAP Prepare Subsequent Handover request, and the list of supported codecs.

The MSC-A server shall indicate to the GERAN the selected service within the RANAP RELOCATION REQUEST message and shall set the RAB parameters within the RANAP RELOCATION REQUEST message according to the selected service. The MSC server shall not set codec information in the NAS Synchronisation Indicator (see [4]). Instead it shall set codec information in the GERAN BSC container.

8.3.3.1.2 Handover Command/Relocation Detect

When the MSC-A server sends the MAP Prepare Subsequent Handover response message or alternatively if it receives a Relocation Detect message, the MSC-A server uses the Change Flow Direction procedure to requests the MGW-A to set the Handover Device to intermediate state (bullet 2 in figure 8.18/2).

8.3.3.1.3 Relocation Complete

When the MSC-A server receives a Relocation Complete message, it informs the MSC-B server about reception of this message. MSC-A server then initiates call clearing towards the MSC-B server as described in clause 7.3.

8.3.3.1.4 Interworking function

The interworking function used by MGW-A before handover will also be used after handover.

8.3.3.1.5 Codec handling

The MGW may include a speech transcoder based upon the speech coding information provided to each bearer termination.

8.3.3.1.6 Voice Processing function

Voice processing function(s) provided by MGW-A and MGW-B before handover, may be continued or modified by MGW-A after handover.

8.3.3.1.7 Failure Handling in MSC server

When a procedure between the MSC-A server and the MGW fails the MSC-A server shall handle the failure as an internal error in accordance with 3GPP TS 23.009 [8] and 3GPP TS 29.010 [23]. If MGW resources have been already seized at the target access side then the resources shall be released using the Release Termination procedure. If the call is to be cleared, then it shall be handled as described in clause 7.3.

8.3.3.2 MSC-B / MGW-B

8.3.3.2.1 Handover Required

For Handover towards GERAN Iu mode the GERAN capabilities of the target cell will be indicated to the MSC-B server within the Handover Required message if the target cell provides different capabilities than the current cell. If no information about the GERAN capabilities of the target cell are received within this message, the MSC-B server shall assume that the GERAN target cell will provide the same capabilities as the current cell (for details see [29]).

The MSC-B server shall indicate to the MSC-A server the GERAN capabilities of the target cell, if available, with the MAP Prepare Subsequent Handover request. For speech calls, the MSC-B server shall additionally indicate to the MSC-A server the selected codec and the currently used codec.

8.3.3.2.2 Handover Complete

When the MSC-B server receives the Handover Complete message, it releases the A-interface line towards the BSC-A and requests the MGW-B to remove the bearer termination towards the BSC-A using the Release Termination procedure (bullet 3 in figure 8.18/2).

8.3.3.2.3 Release of bearer towards MGW-A

When the MSC-B server receives a call clearing indication from the MSC-A server, the MSC-B server handles it as described in subclause 7.2.

8.3.3.2.4 Example

Figure 8.17 shows the network model for Subsequent Inter-MSC GSM to UMTS Handover back to the Anchor MSC. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling (not applicable in case of GSM access) and the bearer. In the MGW the bearer termination T1 is used for the bearer towards RNC-B, the bearer termination T3 is used for the bearer towards MSC-A server, and the bearer termination T2 is used for the bearer towards the succeeding/preceding MGW. In MGW-B the bearer termination T4 is used for the bearer towards BSC-A, bearer termination T5 is used for the bearer towards MGW-A.



Before Handover



During Handover



After Handover

Figure 8.17: Subsequent Inter-MSC GSM to UMTS Handover back to the Anchor MSC (network model)

Figure 8.18 shows the message sequence example for the Subsequent Inter-MSC GSM to UMTS Handover back to the Anchor MSC. It is assumed that the Handover Device is located in the MGW (MGW-A) selected for the call establishment by the MSC server (MSC-A server) which controls the call and the mobility management.

In the example the MSC-A server requests MGW-A to seize RNC-B side bearer termination with specific flow directions. When the relocation is detected in RNC-B the MSC-A server requests to change the flow directions between the terminations within the context in MGW-A. When MSC-A server receives Handover Complete indication from RNC-B it transfers this indication to MSC-B server. MSC-B server releases the A-interface line towards the BSC-A. MSC-A server initiates call clearing towards MSC-B server.



Figure 8.18/1: Subsequent Inter-MSC GSM to UMTS Handover back to the Anchor MSC (message sequence chart)



Figure 8.18/2: Subsequent Inter-MSC GSM to UMTS Handover back to the Anchor MSC (message sequence chart)

8.3.4 Subsequent Inter-MSC GSM to UMTS Handover to a third MSC

The GSM to UMTS handover to a third MSC server (from MSC-B server to MSC-B server) is the combination of the two previous inter-MSC handover cases:

- for MSC-B server a subsequent GSM to UMTS handover from MSC-B server back to MSC-A server as described in clause 8.3.3; and
- for MSC-B server a basic GSM to UMTS handover from MSC-A server to MSC-B server as described in clause 8.3.2.
MSC-A server implements the corresponding parts of each handover case; i.e. access handling in MSC-A server is not included.

8.3.5 GSM to UMTS Handover with lu on IP

If IuCS on IP is supported by the MSC server, the Core Network side procedures described in clauses 8.3.1, 8.3.2, 8.3.3, 8.3.4 shall apply. For the access bearer termination, the exchange of IP addresses via call control procedures is described in this clause.

Before the MSC server starts the access bearer assignment, the MSC server requests the MGW to prepare for the access bearer using the Prepare_IP_Transport procedure. The MSC server requests the MGW to provide an IP Transport address and UDP Port and provides the MGW with the bearer characteristics. For speech calls, the MSC server shall provide the MGW with the speech coding information and conditionally GTT related information in accordance with 3GPP TS 23.226 [28] for the bearer. For a non-speech call the MSC server also provides the MGW with a PLMN Bearer Capability [4]. After the MGW has replied with the IP address and UDP Port the MSC server requests access bearer assignment using the provided IP address and UDP Port in accordance with 3GPP TS 25.413 [26]. The IP addresses and UDP Ports of the MGW and the RNC are exchanged via the RANAP procedures. If the bearer transport is IP and IuUP mode is Transparent, when the MSC receives the RANAP Iu Relocation Request response, it shall send the RNC IP address and UDP Port to the MGW Access bearer termination using the Modify_IP_Transport_Address procedure.

If the bearer transport is IP and IuUP mode is Support, the MGW shall use the source IP address and UDP Port of the IuUP Init packet received from the radio access network as the destination address for subsequent downlink packets.

The sequence is shown in figure 8.3.5/1.



Figure 8.3.5/1 GSM to UMTS Handover with lu on IP

8.4 GSM to GSM

8.4.1 Intra-MSC Inter-BSS GSM to GSM Handover

The procedures specified in 3GPP TS 23.009 [8] for "Intra-MSC Handover" shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network.

8.4.1.1 Handover Required

When the MSC server receives a Handover Required message, it requests the MGW to seize a TDM circuit, using the Reserve Circuit procedure. For non-speech calls the MSC server shall provide the MGW with the GSM Channel coding properties and the same PLMN Bearer Capability [4] as was provided at the last access bearer assignment. The MSC server uses the Change Flow Direction procedure to request the MGW to set the Handover Device to initial state. The MSC server sends the Handover Request message to the BSC-B containing the CIC (bullet 1 in figure 8.20/1).

8.4.1.2 Handover Request Acknowledge

For non-speech calls after receiving Handover Request Acknowledge message if the assigned GSM Channel coding properties differ from the previously provided ones the MSC server provides the MGW-A with the assigned GSM Channel coding properties using the Modify Bearer Characteristics procedure (bullet 2 in figure 8.20/1).

8.4.1.3 Handover Command/Handover Detect

When the MSC server sends the Handover Command message or alternatively if it receives the Handover Detect message, the MSC server uses the Change Flow Direction procedure to requests the MGW to set the Handover Device to intermediate state (bullet 3 in figure 8.20/1).

8.4.1.4 Handover Complete

When the MSC server receives the Handover Complete message, it releases the A-interface line towards BSC-A. The MSC server also requests the MGW to set the Handover Device to its final state by removing the bearer termination towards the BSC-A, using the Release Termination procedure (bullet 4 in figure 8.20/2).

8.4.1.5 Interworking function

The interworking function used by the MGW before handover will also be used after handover.

8.4.1.6 Voice Processing function

After handover, the MGW may continue or modify voice processing function(s) provided to each bearer termination.

8.4.1.7 Failure Handling in MSC server

When a procedure between the MSC server and the MGW fails the MSC server shall handle the failure as an internal error in accordance with 3GPP TS 23.009 [8] and 3GPP TS 29.010 [23]. If MGW resources have already been seized at the target access side then the resources shall be released using the Release Termination procedure. If the call is to be cleared, then it shall be handled as described in clause 7.3.

8.4.1.8 Example

Figure 8.19 shows the network model for the Intra-MSC GSM to GSM Handover. The "squared" line represents the call control signalling. The "dotted" line represents the bearer. The bearer termination T1 is used for the bearer towards BSC-A, bearer termination T3 is used for the bearer towards BSC-B and the bearer termination T2 is used for the bearer towards the succeeding/preceding MGW.





After Handover

Figure 8.19: Intra-MSC GSM to GSM Handover (network model)

Figure 8.20 shows the message sequence example for the Intra-MSC GSM to GSM Handover. It is assumed that the Handover Device is located in the MGW selected for the call establishment by the MSC server, which controls the call and the mobility management.

In the example the MSC server requests seizure of BSC-B side bearer termination with specific flow directions. The MSC server starts handover execution by sending Handover Request towards BSC-B. When the handover is detected in BSC-B the MSC server requests to change the flow directions between the terminations within the context. When MSC server receives Handover Complete indication from BSC-B it releases the A-interface line towards the BSC-A. Finally the MSC server requests the MGW to release BSC-A side bearer termination.



Figure 8.20/1: Intra-MSC GSM to GSM Handover (message sequence chart)



Figure 8.20/2: Intra-MSC GSM to GSM Handover (message sequence chart)

8.4.1.9 Intra-MSC GSM to GSM Handover for A interface over IP

When the MSC server receives the BSSAP Handover Required message, it requests the MGW to reserve an RTP bearer termination using the Reserve RTP Connection Point procedure. The MSC server requests the MGW to reserve an IP address and UDP port. The MSC server uses the Change Flow Direction procedure to request the MGW to set the Handover Device to initial state.

The MGW reserves the RTP termination and indicates the IP address and UDP port number to the MSC server. The IP addresses and UDP ports of the MGW is sent to the BSC in the BSSMAP Handover Request message. When the MSC server receives the BSSMAP Handover Request-Ack message, it shall send the BSC IP address and UDP Port number to the MGW using the Configure RTP Connection Point procedure.

The sequence is shown in Figure 8.4.1.9/1.



Figure 8.4.1.9/1: Intra-MSC GSM to GSM Handover for AoIP (message sequence chart)

8.4.2 Basic Inter-MSC GSM to GSM Handover

The procedures specified in 3GPP TS 23.009 [8] for "Basic Handover Procedure Requiring a Circuit Connection between MSC-A and MSC-B" shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network.

8.4.2.1 MSC-A / MGW-A

8.4.2.1.1 Bearer establishment between MGW-A and MGW-B

The handling of the bearer establishment between MGW-A and MGW-B is as described for a Basic Mobile Originating Call, using either forward or backward bearer establishment. For non-speech calls the MSC-A server shall provide MGW-A with the GSM Channel coding properties and the same PLMN Bearer Capability [4] as was provided at the last access bearer assignment. The MSC-A server also uses the Change Flow Direction procedure to request MGW-A to set the Handover Device to initial state (bullet 3 in figure 8.22/2).

8.4.2.1.2 Handover Command/Handover Detect

When the MSC-A server sends the Handover Command message or alternatively if it receives the Handover Detect message, the MSC-A server uses the Change Flow Direction procedure to requests MGW-A to set the Handover Device to intermediate state (bullet 4 in figure 8.22/2).

8.4.2.1.3 Handover Complete

When the MSC-A server receives the Handover Complete message, it releases the A-interface line towards the BSC-A. The MSC-A server also requests MGW-A to set the Handover Device to its final state by removing the bearer termination towards the BSC-A, using the Release Termination procedure (bullet 5 in figure 8.22/2).

8.4.2.1.4 Interworking function

The interworking function used by MGW-A before handover will also be used after handover.

8.4.2.1.5 Voice Processing function

Voice processing function(s) provided by MGW-A before handover, may be modified or disabled by MGW-A after handover.

8.4.2.1.6 Failure Handling in MSC server

When a procedure between the MSC-A server and MGW-A fails the MSC-A server shall handle the failure as an internal error in accordance with 3GPP TS 23.009 [8] and 3GPP TS 29.010 [23]. If call establishment towards the MSC-B server has already started then the call towards MSC-B server shall be cleared as described in clause 7.3. If the original call is to be cleared, then it shall be handled as described in clause 7.3.

8.4.2.2 MSC-B / MGW-B

8.4.2.2.1 MGW selection

The MSC-B server selects an MGW when it receives Prepare Handover Request message (bullet 1 in figure 8.22/1).

8.4.2.2.2 Bearer establishment towards BSC-B

When the MSC-B server has selected MGW-B it requests MGW-B to seize a TDM circuit, using the Reserve Circuit procedure. The MSC-B server sends the Handover Request message to the BSC-B containing the CIC (bullet 2 in figure 8.22/1).

8.4.2.2.3 Bearer establishment between MGW-A and MGW-B

The handling of the bearer establishment between MGW-A and MGW-B is as described for a Basic Mobile Terminating Call, using either forward or backward bearer establishment.

8.4.2.2.4 Voice Processing function

Voice processing function(s) provided by MGW-A before handover, may be continued or modified by MGW-B after handover.

8.4.2.2.5 Failure Handling in MSC server

When a procedure between the MSC-B server and MGW-B fails the MSC-B server shall handle the failure as an internal error in accordance with 3GPP TS 23.009 [8] and 3GPP TS 29.010 [23]. If MGW-B resources have already been seized at the target access side then the resources shall be released using the Release Termination procedure. The call from MSC-A server shall be released as described at clause 7.1.

8.4.2.2.6 Example

Figure 8.21 shows the network model for the Basic Inter-MSC GSM to GSM. The "squared" line represents the call control signalling. The "dotted" line represents the bearer. In MGW-A the bearer termination T1 is used for the bearer towards BSC-A, bearer termination T3 is used for the bearer towards MGW-B, and the bearer termination T2 is used for the bearer towards the succeeding/preceding MGW. In MGW-B the bearer termination T4 is used for the bearer towards BSC-B, bearer termination T5 is used for the bearer towards MGW-A.



Before Handover



During Handover





Figure 8.21: Basic Inter-MSC GSM to GSM Handover (network model)

Figure 8.22 shows the message sequence example for the Basic Inter-MSC GSM to GSM Handover.

It is assumed that the Handover Device is located in the MGW (MGW-A) selected for the call establishment by the MSC server (MSC-A server) which controls the call and the mobility management.

In the example the MSC-B server requests MGW-B to seize BSC-B side bearer termination. The call is established between MSC-A server and MSC-B server, and the bearer is established between MGW-A and MGW-B. When the handover is detected in BSC-B the MSC-A server requests to change the flow directions between the terminations within the context in MGW-A. When MSC-A server receives Relocation Complete indication from MSC-B server it releases the A-interface line towards the BSC-A. Finally MSC-A server requests MGW-A to remove BSC-A side bearer termination.



Figure 8.22/1: Basic Inter-MSC GSM to GSM Handover (message sequence chart)



Figure 8.22/2: Basic Inter-MSC GSM to GSM Handover (message sequence chart)

8.4.2.3 Basic Inter-MSC GSM to GSM Handover for A Interface over IP

If AoIP is supported by the MSC server, the Core Network side procedures described earlier in 8.4.2 shall apply. For the access side termination, the exchange of IP addresses via call control procedures is described in this clause.

When the MSC-B server receives the MAP Prepare Handover Request., it requests the MGW-B to reserve an RTP bearer termination using the Reserve RTP Connection Point procedure. The MSC-B server requests the MGW-B to reserve an IP address and UDP port.

The MGW-B reserves the RTP termination and indicates the IP address and UDP port number to the MSC-B server. The IP addresses and UDP ports of the MGW is sent to the BSC-B in the BSSMAP Handover Request message. When the MSC-B server receives the BSSMAP Handover Request-Ack message, it shall send the BSC-B IP address and UDP Port number to the MGW-B using the Configure RTP Connection Point procedure. The sequence is shown in Figure 8.4.2.3/1.



Figure 8.4.2.3/1: Basic Inter-MSC GSM to GSM Handover for AoIP (message sequence chart)

8.4.3 Subsequent Inter-MSC GSM to GSM Handover back to the Anchor MSC

The procedures specified in 3GPP TS 23.009 [8] for "Subsequent Handover from MSC-B to MSC-A requiring a Circuit Connection between 3G_MSC-A and 3G_MSC-B" shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network.

8.4.3.1 MSC-A / MGW-A

When the MSC-A server receives a MAP Prepare Subsequent Handover request containing a Handover Request message, it requests MGW-A to seize a TDM circuit, using the Reserve Circuit procedure. For non-speech calls the MSC-A server shall provide MGW-A with the GSM Channel coding properties and the same PLMN Bearer Capability [4] as was provided at the first access bearer assignment The MSC-A server uses the Change Flow Direction Procedure to request MGW-A to set the Handover Device to initial state. The MSC-A server sends the Handover Request message to the BSC-B containing the CIC (bullet 1 in figure 8.24/1).

8.4.3.1.1 Handover Request Acknowledge

For non-speech calls after receiving Handover Request Acknowledge message if the assigned GSM Channel coding properties differ from the previously provided ones the MSC-A server provides the MGW-A with the assigned GSM Channel coding properties using the Modify Bearer Characteristics procedure (bullet 2 in figure 8.24/2).

8.4.3.1.2 Handover Command/Handover Detect

When the MSC-A server sends the MAP Prepare Subsequent Handover response message or alternatively if it receives the Handover Detect message, the MSC-A server uses the Change Flow Direction procedure to request MGW-A to set the Handover Device to intermediate state (bullet 3 in figure 8.24/2).

8.4.3.1.3 Handover Complete

When the MSC-A server receives the Handover Complete message, it informs the MSC-B server about reception of this message. The MSC-A server then initiates call clearing towards the MSC-B server as described in clause 7.3.

8.4.3.1.4 Interworking function

The interworking function used by MGW-A before handover will also be used after handover.

8.4.3.1.5 Voice Processing function

Voice processing function(s) provided by MGW-A and MGW-B before handover, may be continued or modified by MGW-A after handover.

8.4.3.1.6 Failure Handling in MSC server

When a procedure between the MSC-A server and MGW-A fails the MSC-A server shall handle the failure as an internal error in accordance with 3GPP TS 23.009 [8] and 3GPP TS 29.010 [23]. If MGW-A resources have already been seized at the target access side then the resources shall be released using the Release Termination procedure. If the call is to be cleared, then it shall be handled as described in clause 7.3.

8.4.3.2 MSC-B/MGW-B

8.4.3.2.1 Handover Complete

When the MSC-B server receives the Handover Complete message, it releases the A-interface line towards the BSC-A and requests the MGW-B to remove the bearer termination towards the BSC-A using the Release Termination procedure (bullet 4 in figure 8.24/2).

8.4.3.2.2 Release of bearer towards MGW-A

When the MSC-B server receives a call clearing indication from the MSC-A server, the MSC-B server handles it as described in clause 7.2.

8.4.3.2.3 Example

Figure 8.24 shows the network model for the Subsequent Inter-MSC GSM to GSM Handover back to the Anchor MSC. The "squared" line represents the call control signalling. The "dotted" line represents the bearer. In MGW-A the bearer termination T6 is used for the bearer towards BSC-B, bearer termination T3 is used for the bearer towards MGW-B, and the bearer termination T2 is used for the bearer towards the succeeding/preceding MGW. In MGW-B the bearer termination T4 is used for the bearer towards BSC-A, bearer termination T5 is used for the bearer towards MGW-A.



Before Handover



During Handover



After Handover

Figure 8.23: Subsequent Inter-MSC GSM to GSM Handover back to the Anchor MSC (network model)

Figure 8.24 shows the message sequence example for the Subsequent Inter-MSC GSM to GSM Handover back to the Anchor MSC. It is assumed that the Handover Device is located in the MGW (MGW-A) selected for the call establishment by the MSC server (MSC-A server) which controls the call and the mobility management.

In the example the MSC-A server requests MGW-A to seize BSC-B side bearer termination with specific flow directions. When the handover is detected in BSC-B the MSC-A server requests to change the flow directions between the terminations within the context in MGW-A. When MSC-A server receives Relocation Complete indication from BSC-B it transfers this indication to MSC-B server. MSC-B server releases the A-interface line towards the BSC-A. MSC-A server initiates call clearing towards MSC-B server.



Figure 8.24/1: Subsequent Inter-MSC GSM to GSM Handover back to the Anchor MSC (message sequence chart)



Figure 8.24/2: Subsequent Inter-MSC GSM to GSM Handover back to the Anchor MSC (message sequence chart)

Subsequent Inter-MSC GSM to GSM Handover back to the anchor MSC for 8.4.3.3 A Interface over IP

124

When the MSC-A server receives the MAP Prepare Subsequent Handover Request message, it requests the MGW-A to reserve an RTP bearer termination using the Reserve RTP Connection Point procedure. The MSC-A server requests the MGW-A to reserve an IP address and UDP port. The MSC-A server uses the Change Flow Direction procedure to request the MGW-A to set the Handover Device to initial state.

The MGW-A reserves the RTP termination and indicates the IP address and UDP port number to the MSC-A server. The IP addresses and UDP ports of the MGW-A is sent to the BSC-B in the BSSMAP Handover Request message. When the MSC-A server receives the BSSMAP Handover Request-Ack message, it shall send the BSC-A IP address and UDP Port number to the MGW-B using the Configure RTP Connection Point procedure.

The sequence is shown in Figure 8.4.3.3/1.



Figure 8.4.3.3/1: Subsequent Inter-MSC GSM to GSM Handover back to the anchor MSC for AoIP (message sequence chart)

8.4.4 Subsequent GSM to GSM Handover to a third MSC

The GSM to GSM handover to a third MSC server (from MSC-B server to MSC-B server) is the combination of the two previous inter-MSC handover cases:

- for MSC-B server a subsequent GSM to GSM handover from MSC-B server back to MSC-A server as described in clause 8.4.3; and
- for MSC-B server a basic GSM to GSM handover from MSC-A server to MSC-B server as described in clause 8.4.2.

MSC-A server implements the corresponding parts of each handover case, i.e. access handling in MSC-A server is not included.

8.4.5 BSS Internal Handover

8.4.5.1 Introduction

The procedures specified in 3GPP TS 23.009 [8] for "BSS Internal Handover" shall be followed. The protocol messages are defined in 3GPP TS 48.008 [27]. The following clauses describe the additional requirements for the bearer independent CS core network due to BSS internal handover for A-Interface User Plane over IP (AoIP).

A BSS Internal Handover may be initiated by the BSS on its own, or triggered by the MSC server with the transmission of an Internal Handover Enquiry message to the BSS.

8.4.5.2 Internal Handover Required

When the MSC server receives an Internal Handover Required message, it requests the MGW to seize a TDM circuit using the Reserve Circuit procedure or an RTP termination using the Reserve and Configure RTP Connection Point procedure. In the latter case, the MSC server sends the BSC IP address and UDP port and the determined A-Interface Codec Type/Configuration to the MGW and requests the MGW to reserve a local IP address and UDP port.

NOTE: During a BSS Internal handover procedure involving an IP-based target A-Interface Type before and after the handover, the BSC IP addresses and UDP ports used before and after the handover may be identical or different. This is transparent to the MSC Server. In the former case, the BSC can receive on its BSC IP address and UDP port identical RTP flows received from the source and target access terminations reserved in the MGW.

For non-speech calls the MSC server shall provide the MGW with the GSM Channel coding properties and the same PLMN Bearer Capability as was provided at the last access bearer assignment. The MSC server uses the Change Flow Direction procedure to request the MGW to set the Handover Device to initial state.

The MGW reserves the TDM circuit or RTP termination and indicates in the latter case its own IP address and UDP port to the MSC server. The MSC server then sends the Internal Handover Command message to the BSC containing the speech codec type/configuration chosen and either the CIC or the MGW IP address and UDP port.

If the resources requested by the BSS in the Internal Handover Required message (e.g. codec type or codec configuration, interface type, etc.) are not available, an Internal Handover Required Reject message shall be returned to the BSS with the appropriate failure cause, and the procedure shall be terminated.

8.4.5.3 Internal Handover Command/Handover Detect

When the MSC server sends the Internal Handover Command message or alternatively if it receives the Handover Detect message, the MSC server uses the Change Flow Direction procedure to requests the MGW to set the Handover Device to intermediate state (bullet 2 in figure 8.4.5.8.2).

8.4.5.4 Handover Complete

When the MSC server receives the Handover Complete message, it requests the MGW to set the Handover Device to its final state by removing the bearer termination towards the source BSC IP address and UDP port number, using the Release Termination procedure (bullet 3 in figure 8.4.5.8.2).

8.4.5.5 Interworking function

The interworking function used by the MGW before handover will also be used after handover.

8.4.5.6 Voice Processing function

After handover, the MGW may continue or modify voice processing function(s) provided to each bearer termination.

8.4.5.7 Failure Handling in MSC server

When a procedure between the MSC server and the MGW fails the MSC server shall handle the failure as an internal error in accordance with 3GPP TS 23.009 [8] and 3GPP TS 29.010 [23]. If MGW resources have already been seized

for BSS internal handover then the resources shall be released using the Release Termination procedure. If the call is to be cleared, then it shall be handled as described in clause 7.3.

If the MSC server receives the Handover Failure message during BSS internal handover procedure, it shall release MGW resources if they have been seized for BSS internal handover using the Release Termination procedure.

8.4.5.8 Example for a successful BSS Internal Handover

Figure 8.4.5.8.1 shows the network model for the BSS Internal Handover, where IP transport is used on the A interface before and after the BSS Internal Handover procedure. The "squared" line represents the call control signalling. The "dotted" line represents the bearer. The bearer termination T1 is used for the bearer towards the source BSC IP address and UDP port number Ta, bearer termination T3 is used for the bearer towards the target BSC IP address and UDP port number Tb and the bearer termination T2 is used for the bearer towards the succeeding/preceding MGW.







During BSS Internal Handover



After BSS Internal Handover

Figure 8.4.5.8.1: Intra-MSC BSS Internal Handover (network model)

Figure 8.4.5.8.2 shows the message sequence example for the Intra-MSC BSS Internal Handover.

In the example, the MSC server receives the Internal Handover Required message and requests the MGW to reserve an RTP bearer termination (T3) using the Reserve and Configure RTP Connection Point procedure with specific flow directions.

The MSC server starts the Internal Handover execution by sending the Internal Handover Command towards the BSC in which it provides the IP address and UDP port of the target MGW RTP termination (T3). When the handover is detected in the BSC, the MSC server requests the MGW to change the flow directions between the terminations within the context. When the MSC server receives the Handover Complete indication from the BSC it requests the MGW to release the bearer termination T1 towards the source BSC IP address and UDP port number Ta that was used before the BSS Internal Handover was initiated. The sequence is shown in Figure 8.4.5.8.2.



Figure 8.4.5.8.2: Intra-MSC BSS Internal Handover for AoIP (message sequence chart)

8.5 Handling of GSM Services after UMTS to GSM Handover

The handling of GSM services after handover in the Bearer Independent CS Core Network architecture is as for the corresponding UMTS services, if not stated differently.

9 Compatibility Issues

A Release 4 (or later) node, according to 3GPP TS 23.205, is backward compatible with a Release 99 (or earlier) node.

9.1 Interworking with GERAN (A i/f)

9.1.1 Introduction

The A-interface signalling terminates in the MSC server and the user plane terminates in the MGW. TDM and/or IP transports may be supported on the A-interface. The MSC server uses the Mc interface to remotely control the TDM or IP resources in the MGW.

For intra-MSC handover, the target A-interface is handled as described in clause 8. If the target A-interface user plane terminates in a different MGW from the MGW that terminates the serving A-interface user plane, a bearer has to be established between the two MGWs using Prepare Bearer and Establish Bearer procedures. Because the same MSC server controls both MGWs, no external call control signalling is involved.

It is important to note that the separation between the user and control planes remains the same before and after interaction with services in the 3G BICC CSCN.

9.1.2 A interface over TDM

Only one MSC server may control the TDM circuits connected to one GERAN node.

For each TDM circuit a physical termination is provisioned in the MGW. The TDM circuit is identified by the termination Id in the Mc interface. Since TDM circuits are also grouped together, the physical termination Ids are structured in accordance with the grouping of TDM circuits. The MSC server also knows the termination Ids and the grouping of termination Ids. The physical termination exists as long as the TDM circuit(s) exists in the MGW.

Figure 9.1 shows the network model for the A-interface. The "squared" line represents the call control signalling and the "dotted" line represents the TDM circuits. The terminations T1-Tn represent the TDM circuits in the MGW. The MSC server has a mapping table between circuits CIC1-CICn and the terminations T1-Tn.



Figure 9.1 TDM circuits used for A-interface (network model)

For call-independent transactions the general (G)MSC server-MGW procedures, as described in clause 10, apply to the physical terminations in the same way as to any other terminations.

For call related transactions the handling as described in the clauses 6, 7 and 8 apply to physical terminations in the same way as any other terminations. All call related procedures, except Prepare Bearer, Establish Bearer, Release Bearer and Tunnel Information Up/Down, as described in clause 16, apply to the physical terminations in the same way as any other terminations.

10 General (G)MSC server-MGW Procedures

10.1 MGW Unavailable

The (G)MSC server recognises that the MGW is unavailable in the following 4 cases:

1. The signalling connection is unavailable



Figure 10.1: Signalling connection failure

2. The MGW indicates the failure condition to all connected (G)MSC servers



Figure 10.2: MGW indicates the Failure/Maintenance locking

The failure indication indicates that the MGW will soon go out of service and that no new connections should be established using this MGW. The MGW can choose between the "graceful" and the "forced" method. In the graceful method the connections are cleared when the corresponding calls are disconnected. In the forced method all connection are cleared immediately.

- 3. The (G)MSC server recognises that the MGW is not functioning correctly, e.g. because there is no reply on periodic sending of Audits. The periodic sending of Audits by (G)MSC server should go on.
- 4. The MGW indicates the maintenance locking condition to all concerned (G)MSC servers.

The maintenance locking indication indicates that the MGW is locked for new calls and that no new connections shall be established using this MGW. The MGW can choose between the "graceful" and the "forced" method. In the graceful method the connections are cleared when the corresponding calls are disconnected. In the forced method all connection are cleared immediately

In all of the above cases the (G)MSC server shall prevent the usage of the MGW until the MGW has recovered or the communication with the MGW is restored. The (G)MSC server shall prohibit the surrounding network from seizing circuits connected to the unavailable TDM access by sending blocking messages.

10.2 MGW Available

The (G)MSC server discovers that the MGW is available when it receives an MGW Communication Up message or an MGW Restoration message. If the (G)MSC does not wish to sustain an association with the MGW, the response sent to the MGW may indicate an alternative MGCId signalling address, in which case the MGW shall not consider itself registered and should preferably attempt to re-register with this alternative MGC before any further alternate MGCs. Otherwise, the response shall not indicate any alternative MGCId signalling address.

When the (G)MSC server discovers that the MGW is available the following shall occur:

1. Signalling recovery

The MGW indicates to all connected (G)MSC servers that the signalling connection is restored. Any changes to the physical termination state during the loss of communication shall be reported by the MGW using Termination Out Of Service (10.7) or Termination Restoration (10.8), otherwise the (G)MSC Server can assume that Terminations service state has not changed. To avoid unsuccessful calls for physical terminations which went out of service during the loss of communication but have not yet been reported by the MGW the MSC Server may Audit the physical terminations before it uses them.

NOTE: Auditing in this case will cause duplicate signalling.



Figure 10.3: Communication goes up

2. MGW restoration/maintenance unlocking indication.

The MGW indicates to all connected (G)MSC servers that normal operation has resumed. Changes of the physical termination state during the connection break shall be reported by the MGW using Termination Out Of Service (10.7) and Termination restoration (10.8) procedures, otherwise the (G)MSC Server can assume that Terminations service state has not changed. To avoid unsuccessful calls for physical terminations which went out of service during the loss of communication but have not yet been reported by the MGW the (G)MSC Server may Audit the physical terminations before it uses them.

NOTE: Auditing in this case will cause duplicate signalling.



Figure 10.4: MGW indicates recovery from a failure/or maintenance unlocking

NOTE: This procedure may be used after recovery from a signalling failure.

3. The (G)MSC server recognises that the MGW is now functioning correctly, e.g. because there is a reply on periodic sending of Audits.

After this the (G)MSC server can use the MGW. If the corresponding devices of the surrounding network are blocked, unblocked messages shall be sent to the nodes concerned for devices that are in service: the (G)MSC Server has determined the actual state of the physical devices either by receiving Termination Restoration procedure (10.8) or by performing an Audit.

If none of 1,2, and 3 happens the (G)MSC server can initiate the (G)MSC Server Ordered Re-register procedure.

10.3 MGW Recovery

If the MGW recovers from a failure, is maintenance unlocked, or it has been restarted, it registers to its known (G)MSC servers using the MGW Restoration procedure or the MGW Registration procedure. The MGW can indicate whether the Service has been restored or whether it has restarted with a cold or warm boot. If the (G)MSC does not wish to sustain an association with the MGW, the response sent to the MGW may indicate an alternative MGCId signalling addres, in which case the MGW shall not consider itself registered and should preferably attempt to re-register with this alternative MGC before any further alternate MGCs. Otherwise, the response shall not indicate any alternative MGCId signalling address.



Figure 10.5/1: MGW Restoration





After the recovery the (G)MSC server can use the MGW. After a MGW warm boot or service restored, the (G)MSC Server can assume the physical terminations state has not changed. Changes of the physical termination state during the connection break shall be reported by the MGW using Termination Out Of Service (10.7) and Termination restoration (10.8) procedures. To avoid unsuccessful calls for physical terminations which went out of service during the loss of communication but have not yet been reported by the MGW the MSC Server may Audit the physical terminations before it uses them.

NOTE: Auditing in this case will cause duplicate signalling.

After receiving a MGW Register (with cold boot) the (G)MSC Server shall audit the termination state of all physical terminations (10.9) if it has not been specifically informed of the actual service state since receiving the MGW service change on ROOT (Register) via independent service change commands (10.7 or 10.8). The (G)MSC Server shall thereby determine the actual Service State of devices. If the corresponding devices of the surrounding network are blocked, unblocked messages are sent to the nodes concerned for all devices that are in service.

10.4 (G)MSC server Recovery

10.4.1 General

If an MGW-unavailable condition is provoked by a failure/recovery action, the (G)MSC server recovery sequence will, from an information flow point of view, look like MGW unavailable and then MGW available. If an MGW-unavailable condition is not provoked, the (G)MSC server recovery sequence will look like MGW available.

After the information flow, the terminations affected by the recovery action are released.

10.4.2 (G)MSC Server Restoration



Figure 10.6: (G)MSC Server Restoration

NOTE: Normal release procedure may also be initiated.

After the recovery action is complete and it is possible to signal to the MGW the (G)MSC server starts a timer Tw. If recovery indications are not received (MGW Communication Up or MGW Restoration) from the MGW during Tw an Audit is sent. If the (G)MSC server receives a recovery indication or MGW communication up indication, it shall acknowledge the indication before the (G)MSC Server Restoration may be sent or the release procedure is initiated.

10.5 MGW Re-register

When the (G)MSC requests an MGW to perform a registration (see clause 10.2, 10.3 and 10.6), the MGW performs a re-registration to the (G)MSC which is defined in the (G)MSC address. If the (G)MSC Server is uncertain of the termination service state of physical terminations after the Re-Register it shall use the Audit Procedure (10.9). Changes of the physical termination state during the connection break shall be reported by the MGW using Termination Out Of Service (10.7) and Termination restoration (10.8) procedures.



Figure 10.7: Re-registration of an MGW

10.6 MGW Re-registration Ordered by (G)MSC server

If the (G)MSC server knows that communication is possible, but the MGW has not registered, the (G)MSC server can order re-registration of the MGW.



Figure 10.8: Re-registration ordered by the (G)MSC server

If the re-registration request is accepted the MGW uses the MGW Re-register procedure to register with the (G)MSC server.

10.7 Removal from Service of a Physical Termination

The MGW indicates the removal from service of a physical termination using the Termination Out-of-Service procedure. In this procedure the MGW indicates which termination is to be removed from service and whether the "graceful" or "forced" method will be used. In the graceful method a possible connection is cleared when the corresponding call is disconnected. In the forced method the possible connection is cleared immediately.



Figure 10.9: Removal from service of a Physical Termination

The (G)MSC server shall prevent the use of the Termination(s) concerned until the physical termination is restored to service.

10.8 Restoration to Service of a Physical Termination

If the physical termination is restored to service, the MGW shall report it to the (G)MSC server(s) using the Termination Restoration procedure.



Figure 10.10: Restoration to service of a Physical Termination

The (G)MSC server can use the physical termination when the termination has been restored to service. If the corresponding devices of the surrounding network are blocked, the (G)MSC server sends an unblocked message to each node concerned.

10.9 Audit of MGW

10.9.1 Audit of Value

The (G)MSC server may request the MGW to report the current values assigned to distinct objects in the MGW. Objects, which can be addressed, are listed in 3GPP TS 29.232 [6]. This procedure shall be used when a change has occurred in the (G)MSC server such that the server is unsure of the current Service State of Terminations, such as restart or new trunks configured in the server. For any situation where a change to the Service State occurs in the MGW the (G)MSC server shall expect the Service State to be reported by Termination Restoration (10.8) or Termination Out Of Service (10.7).



Figure 10.11: Audit Value

10.9.2 Audit of Capability

The (G)MSC server may request the MGW to report the capabilities of distinct objects in the MGW. Objects, which can be addressed, are listed in 3GPP TS 29.232 [6].



Figure 10.12: Audit Capability

10.10 MGW Capability Change

The MGW reports a change of capability of distinct objects in the MGW. Objects, which can be addressed, are listed in 3GPP TS 29.232 [6].



Figure 10.13: Capability Update

The (G)MSC server can use the Audit Value and/or Audit Capability procedures to obtain further information, about the objects whose capabilities have changed.

10.11 Void



Figure 10.14: (G)MSC Server Out of Service

If a (G)MSC server discovers that it wants to go out of service it starts a (G)MSC Server Out of Service procedure. The (G)MSC server can indicate whether it requires the context to be cleared immediately (forced) or cleared as the bearer control protocol clears the bearer (Graceful). Physical terminations are always cleared when the (G)MSC Server Out of Service indication reaches the MGW.

10.13 MGW Resource Congestion Handling - Activate

When the (G)MSC server requires that a MGW congestion notification mechanism be applied in the MGW, the (G)MSC server shall use the MGW Resource Congestion Handling - Activate procedure towards the MGW.



Figure 10.15: MGW Resource Congestion Handling - Activate

10.14 MGW Resource Congestion Handling -Indication

When the (G)MSC server receives a load reduction notification from the MGW via the MGW Resource Congestion Handling - Indication procedure, the (G)MSC server tries to reduce the processing load that the (G)MSC server creates on the MGW. The MGW shall decide the actual level of traffic reduction.



Figure 10-17: MGW Resource Congestion Handling – Indication

10.15 Control association monitoring

Monitoring of the H.248 control association may be performed by monitoring the status of the transport link association where the transport protocol provides sufficient coupling to the H.248.1 protocol, i.e. if the transport link association is disconnected when no local H.248.1 protocol connection exists.

An alternative method for the MGW to detect loss of the MGC may be achieved by requesting the MGW to poll the (G)MSC periodically

Upon registration of a MGW, the (G)MSC server may use the Inactivity Timeout - Activate procedure towards the MGW to request the MGW to monitor incoming messages for periods of silence exceeding the maximum inactivity timer value.



Figure 10.15.1: Inactivity Timeout - Activate

Upon receipt of an inactivity timeout notification from the MGW via the Inactivity Timeout -Indication procedure, the (G)MSC server shall send a reply to the MGW. If the MGC has failed, the MGW will not receive a reply.



Figure 10.15.2: Inactivity Timeout - Indication

If no Inactivity Timeout – Indication Ack reply is received, the MGW shall consider the (G)MSC to have failed. The MGW may then attempt to recontact its controlling (G)MSC by performing MGW Communication UP. If not successful, the MGW may then attempt to register to a new (G)MSC.

10.16 Hanging termination detection

Whenever requesting new ephemeral bearer terminations, the (G)MSC shall request the MGW to periodically report termination heartbeat indications to detect hanging context and termination in the MGW that may result e.g. from a loss of communication between the MSC-S and the MGW. This may also be done when requesting TDM terminations, though alternative means exist for the (G)MSC to detect hanging TDM terminations, e.g. via receipt of the error #433 when seizing the termination.

When the (G)MSC server receives a termination heartbeat notification from the MGW via the Termination heartbeat - Indication procedure, the (G)MSC shall return a Termination heartbeat –Indication Ack (without an error) if the context id / termination identity combination exists in the (G)MSC. If it does not exist, the (G)MSC shall return an error and

shall correct the mismatch, e.g. by requesting the MGW to subtract the indicated termination and to clear any associated context.



Figure 10.16.1: Termination heartbeat – Indication

11 Identities

11.1 Bearer Address and Binding Reference

The Bearer Address is exchanged on the Nc and Mc interfaces to identify the termination point of the bearer control signalling within the peer Media Gateway.

A Binding Reference is an identity, unique within the scope of one bearer control function, which identifies a bearer network connection. This information is exchanged on the Nc and Mc interfaces. The bearer control function is identified by the Bearer Address.

11.2 MGW-ld

The Media Gateway Identity (MGW-Id) is information sent on the Nc interface to aid Media Gateway selection by the succeeding/preceding node.

The MGW-Id is bearer independent and it can be translated into a signalling address towards the appropriate MGW.

11.3 (G)MSC server Address

The (G)MSC server Address defines the signalling address associated with the (G)MSC server that is used to interact with the Media Gateway over the Mc interface. This is a unique address in the network service supplier domain.

12 Operational Aspects

12.1 Charging

No impacts.

13 Interactions with Other Services

NOTE 1: All message sequence charts in this clause are informative examples.

NOTE 2: The continuity indication in the IAM is not used to indicate that a continuity check will be performed on the current leg of the call, but it is used to indicate that a Continuity message can be expected as a result of a continuity check on a preceding ISUP circuit, or establishment of a preceding bearer connection.

13.1 Enhanced Multi-Level Precedence and Pre-emption service (eMLPP)

No impact.

13.2 Call Deflection Service

The procedures specified in 3GPP TS 23.072 [9] for the Call Deflection supplementary service shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network.

13.2.1 MGW selection and incoming side bearer establishment

The MGW selected for the mobile terminating call is used. The incoming side bearer has already been established by the mobile terminating call procedures.

13.2.2 IU/A-interface release

If the call deflection request from a served subscriber is accepted the call towards the served mobile subscriber shall be released as described in the clause for call clearing.

13.2.3 Notification to the Calling Subscriber

If the served mobile subscriber has requested that the calling subscriber shall receive a notification about the call forwarding, a notification is sent to the calling party. If the notification is implemented using intermediate tones or announcements the MSC server requests the MGW to play an announcement/tone to the calling party, as described in clause 14.6, before establishing the call to the forwarded-to subscriber.

13.2.4 Initial addressing

The call towards the deflected-to subscriber is established as for basic call. If out-of-band transcoder control is applied for a speech call, it shall be performed in accordance with 3GPP TS 23.153 [3]. After the possible generation of in-band information has been completed the MSC server shall indicate in the IAM that forward or backward bearer establishment is to be used. The MSC server shall indicate in the IAM that no Continuity message will follow since the incoming bearer has already been established.

The MGW-id can be provided to the succeeding node in the IAM.

13.2.5 Establishment of bearer towards the forwarded-to subscriber

The bearer establishment towards the forwarded-to subscriber is performed as described for the mobile originating call, network side bearer establishment, using either forward or backward bearer establishment. The MSC server also requests the MGW to both-way through-connect the bearer.

13.2.6 Failure handling in MSC server

The failures are handled as described for the basic mobile originating call.

13.2.7 Example

Figure 13.1 shows the network model for call deflection. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling and the bearer. Note that for a TDM access case there is no separation of call and bearer control signalling. The MSC server replaces the bearer termination for the served mobile

subscriber (T_B) with the bearer termination for the deflected-to subscriber (T_C) in an existing context in the MGW. The bearer termination T_A is used for the bearer towards the preceding MGW (calling subscriber).



Figure 13.1: Call deflection (Network model)

Figure 13.2 shows the message sequence example for the call deflection with a possible notification to the calling party using an announcement. In the example, after the call and the bearer towards the access side have been released the MSC server requests the MGW to remove the bearer termination for the served mobile subscriber, and optionally requests the MGW to play an announcement and to notify the announcement completion. The MSC server shall request the establishment of the call towards the deflected-to subscriber after the possible announcement has completed.



NOTE: OR1: Notification to calling subscriber required (Y:yes N:no)

Figure 13.2: Call deflection (message sequence chart)

13.3 Line identification Services

13.3.1 Calling Line Identification Presentation (CLIP)

No impact.

13.3.2 Calling Line Identification Restriction (CLIR)

No impact.

13.3.3 Connected Line Identification Presentation (COLP)

No impact.

13.3.4 Connected Line Identification Restriction (COLR)

No impact.

13.4 Call Forwarding Services

13.4.1 Call Forwarding Unconditional (CFU)

The procedures specified in 3GPP TS 23.082 [12] for the Call Forwarding Unconditional supplementary service shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network.

13.4.1.1 MGW selection

If in-band information is to be provided to the calling subscriber the GMSC server shall select the MGW before providing the in-band information. The MGW selection can be based on a possibly received MGW-Id from the preceding node.

If in-band information is to not to be provided to the calling subscriber the GMSC server shall select the MGW for the bearer as described for the basic mobile terminating call.

13.4.1.2 Incoming side bearer establishment

The incoming side bearer establishment is handled in the GMSC server as described for the mobile terminating call using either forward or backward bearer establishment.

13.4.1.3 Notification to the Calling Subscriber

If the served mobile subscriber has requested that the calling subscriber shall receive a notification about the call forwarding, a notification is sent to the calling party. If the notification is implemented using intermediate tones or announcements the GMSC server requests the MGW to play an announcement/tone to the calling party, as described in clause 14.6, before establishing the call to the forwarded-to subscriber.

13.4.1.4 Initial addressing

If the incoming call is to be forwarded without being offered to the served mobile subscriber the call towards the forwarded-to subscriber is established as for a basic call. If out-of-band transcoder control is applied for a speech call, it shall be performed in accordance with 3GPP TS 23.153 [3]. After the possible generation of in-band information has been completed the initial addressing towards the forwarded-to subscriber is performed as described for the basic mobile terminating call indicating either forward or backward bearer establishment.

13.4.1.5 Establishment of bearer towards the forwarded-to subscriber

The bearer establishment towards the forwarded-to subscriber is performed as described for the mobile originating call, network side bearer establishment, using either forward or backward bearer establishment. The GMSC server also requests the MGW to both-way through-connect the bearer.

13.4.1.6 Confirmation of bearer establishment

The confirmation of the bearer establishment is handled as described for the basic mobile terminating call.

13.4.1.7 Failure handling in GMSC server

The failures are handled as described for the basic mobile terminating call.

13.4.1.8 Example

Figure 13.3 shows the network model for call forwarding unconditional. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling and the bearer. The GMSC server seizes one context with two bearer terminations in the MGW. The bearer termination T_A is used for the bearer towards the preceding MGW (calling subscriber) and the bearer termination T_C is used for the bearer towards the succeeding MGW (forwarded-to subscriber).



Figure 13.3: CFU (Network model)

Figure 13.4 shows the message sequence example for the call forwarding unconditional with a possible notification to the calling party using an announcement. In the example the GMSC server optionally requests the MGW to play an announcement and to notify the announcement completion, after the bearer to the incoming side has been established. When the possible announcement has completed the GMSC server requests the establishment of the call and the bearer towards the forward-to subscriber.


Figure 13.4: CFU (message sequence chart)

13.4.2 Call Forwarding on mobile subscriber Busy (CFB)

The procedures specified in 3GPP TS 23.082 [12] for the Call Forwarding on Busy supplementary service shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network.

13.4.2.1 Network Determined User Busy (NDUB)

If the mobile is Network Determined User Busy the incoming call for the specified basic service(s) shall be forwarded without being offered to the served mobile subscriber.

13.4.2.1.1 MGW selection

The MSC server shall select an MGW for the bearer connection either before sending the IAM or after receiving the Bearer Information message. If the MSC server received an MGW-id from the preceding node and/or from the succeeding node, then it may use one of them for the MGW selection.

If in-band information is to be provided to the calling subscriber the MSC server shall select the MGW before providing the in-band information. The MGW selection can be based on a possibly received MGW-Id from the preceding node.

NOTE: As an implementation option, if there is no need for the MSC server to manipulate the bearer, the MSC server may only perform call control signalling without any associated MGW. In that case the bearer related information shall be provided transparently through the MSC server.

13.4.2.1.2 Incoming side bearer establishment

The incoming side bearer establishment is handled in the MSC server as described for the mobile terminating call using either forward or backward bearer establishment. The incoming side bearer establishment can take place either before or after the detection of NDUB condition.

13.4.2.1.3 Notification to the calling subscriber

If the served mobile subscriber has requested that the calling subscriber shall receive a notification about the call forwarding, a notification is sent to the calling party. If the notification is implemented using intermediate tones or announcements the MSC server requests the MGW to play an announcement/tone to the calling party, as described in clause 14.6, before establishing the call to the forwarded-to subscriber.

13.4.2.1.4 Initial addressing

The call towards the forwarded-to subscriber is established as for basic call. If out-of-band transcoder control is applied for a speech call, it shall be performed in accordance with 3GPP TS 23.153 [3]. In-band information may be played to the calling subscriber.

If no in-band information is to be played to the calling subscriber, in order to withhold the call completion until the establishment of the bearer is complete.

The MSC server shall indicate in the IAM that the Continuity message will follow from the preceding node if either of the following conditions is satisfied before sending the IAM:

- 1. The incoming IAM indicated that the Continuity message will follow, but no Continuity message has been received, or
- 2. The incoming side bearer has not been established.

If the MGW is selected at an early stage the MGW-id can be provided to the succeeding node in the IAM.

13.4.2.1.5 Establishment of bearer towards the forwarded-to subscriber

The bearer establishment towards the forwarded-to subscriber is performed as described for mobile originating call, network side bearer establishment, using either forward or backward bearer establishment. The MSC server also requests the MGW to both-way through-connect the bearer.

13.4.2.1.5 Confirmation of bearer establishment

If the outgoing IAM indicated that the Continuity message will follow, the Continuity message is sent when both of the following conditions are satisfied:

- 1. Either:
 - a. The incoming IAM indicated that the Continuity message will follow, and a Continuity message has been received, or
 - b. The incoming IAM indicated that no Continuity message will follow.
- 2. Either:
 - a. The MSC server has selected an MGW, and a notification indicating successful completion of the incoming side bearer set-up has been received from the MGW using the Bearer Established procedure, or
 - b. MGW selection is not required for this call.

13.4.2.1.6 Failure handling in MSC server

The failures are handled as described for the basic mobile originating call.

13.4.2.1.7 Example

Figure 13.5 shows the network model for call forwarding busy (network determined user busy). The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling and the bearer. The MSC server seizes one context with two bearer terminations in the MGW. The bearer termination T_A is used for the bearer towards the preceding MGW (calling subscriber) and the bearer termination T_C is used for the bearer towards the succeeding MGW (forwarded-to subscriber).





Figure 13.6 shows the message sequence example for the call forwarding busy (network determined user busy) using a possible announcement. In the example the MSC server optionally requests the MGW to play an announcement and to notify the announcement completion, after the bearer to the incoming side has been established. When the possible announcement has been completed the MSC server requests the establishment of the call towards the forward-to subscriber.



NDUB: Network Determined User Busy

Figure 13.6: CFB (NDUB)

13.4.2.2 User Determined User Busy (UDUB)

13.4.2.2.1 MGW selection

The MGW selected for the mobile terminating call is used, if already selected by the mobile terminating call procedures.

The MSC server selects an MGW for the bearer either before sending the IAM of after receiving the Bearer Information message. If the MSC server received an MGW-id from the preceding node and/or from the succeeding node, then it may use one of them for the MGW selection.

If in-band information is to be provided to the calling subscriber the MSC server shall select the MGW before providing the in-band information. The MGW selection can be based on a possibly received MGW-Id from the preceding node.

NOTE: As an implementation option, if there is no need for the MSC server to manipulate the bearer, the MSC server may only perform call control signalling without any associated MGW. In that case the bearer related information shall be provided transparently through the MSC server.

13.4.2.2.2 Incoming side bearer establishment

For bearer establishment, the sending of bearer information is handled in the MSC server as described for the basic mobile terminating call indicating either forward or backward bearer establishment. The incoming side bearer establishment can take place either before or after the detection of UDUB condition.

13.4.2.2.3 IU/A-interface release

If the mobile is not Network Determined User Busy (NDUB as defined in GSM TS 02.01) the incoming call is offered (as a normal or waiting call) to the served mobile subscriber. If the mobile indicating "User Busy" subsequently releases the call, the call towards the served mobile subscriber is released as described in the clause for call clearing.

NOTE: The MSC server orders the MGW to remove the bearer termination towards the served mobile subscriber only in the case where the radio resources had already been allocated in the MGW (bullet 1 in figure 13.8).

13.4.2.2.4 Notification to the Calling Subscriber

If the served mobile subscriber has requested that the calling subscriber shall receive a notification about the call forwarding, a notification is sent to the calling party. If the notification is implemented using intermediate tones or announcements the MSC server requests the MGW to play an announcement/tone to the calling party as described in clause 14.6 before establishing the call to the forwarded-to subscriber.

13.4.2.2.5 Initial addressing

The call towards the forwarded-to subscriber is established as basic call. If out-of-band transcoder control is applied for a speech call, it shall be performed in accordance with 3GPP TS 23.153 [3]. In-band information may be played to the calling subscriber.

If no in-band information is to be played to the calling subscriber, in order to withhold the call completion until the establishment of the bearer is complete. The MSC server shall indicate in the IAM that the Continuity message will follow from the preceding node if either of the following conditions is satisfied before sending the IAM:

- 1. The incoming IAM indicated that the Continuity message will follow, but no Continuity message has been received, or
- 2. The incoming side bearer has not been established.

If the MGW is selected at an early stage the MGW-id can be provided to the succeeding node in the IAM.

13.4.2.2.6 Establishment of bearer towards the forwarded-to subscriber

The bearer establishment towards the forwarded-to subscriber is performed as described for the mobile originating call, network side bearer establishment, using either forward or backward bearer establishment. The MSC server also requests the MGW to both-way through-connect the bearer.

13.4.2.2.7 Confirmation of bearer establishment

If the outgoing IAM indicated that the Continuity message will follow, the Continuity message is sent when both of the following conditions are satisfied:

1. Either:

- a. The incoming IAM indicated that the Continuity message will follow, and a Continuity message has been received from the preceding node (bullet 8 in figure 6.6), or
- b. The incoming IAM indicated that no Continuity message will follow;
- 2. Either:
 - a. The MSC server has selected an MGW, and a notification of successful bearer establishment in the incoming side has been received from the MGW (bullet 7 in figure 6.6), or
 - b. MGW selection is not required for this call.

13.4.2.2.8 Failure handling in MSC server

The failures are handled as described for the basic mobile originating call.

13.4.2.2.9 Example

Figure 13.7 shows the network model for call forwarding busy (user determined user busy). The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling and the bearer. Note that for a TDM access case there is no separation of call and bearer signalling. The MSC server replaces the bearer termination for the served mobile subscriber (T_B) with the bearer termination for the forwarded-to subscriber (T_C) in an existing context in the MGW. The bearer termination T_A is used for the bearer towards the preceding MGW (calling subscriber).



Figure 13.7: CFB; UDUB (Network model)

Figure 13.8 shows the message sequence example for the call forwarding busy (user determined user busy) with a possible notification to the calling party using an announcement. In the example, after the call and the bearer towards the access have been released the MSC server requests the MGW to remove the bearer termination for the served mobile subscriber, optionally requests the MGW to play an announcement and to notify the announcement completion. After the possible announcement has been completed the MSC server requests the establishment of the call towards the forward-to subscriber.



Figure 13.8: CFB (UDUB)

13.4.3 Call Forwarding on No Reply (CFNRy)

The procedures specified in 3GPP TS 23.082 [12] for the Call Forwarding on No Reply supplementary service shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network.

13.4.3.1 MGW selection and incoming side bearer establishment

The MGW selected for the mobile terminating call is used. The incoming side bearer has already been established by the mobile terminating call procedures.

13.4.3.2 IU/A-interface release

If the call is not answered within the period defined by the no reply condition timer the call towards the served mobile subscriber will be released as described in the clause for call clearing.

13.4.3.3 Notification to the Calling Subscriber

If the served mobile subscriber has requested that the calling subscriber shall receive a notification about the call forwarding, a notification is sent to the calling party. If the notification is implemented using intermediate tones or announcements the MSC server requests the MGW to play an announcement/tone to the calling party, as described in clause 14.6, before establishing the call to the forwarded-to subscriber.

13.4.3.4 Initial addressing

The call towards the forwarded-to subscriber is established as a basic call. If out-of-band transcoder control is applied for a speech call, it shall be performed in accordance with 3GPP TS 23.153 [3]. After the possible generation of in-band information has been completed (bullet 2 in figure 13.10) the MSC server shall indicate in the IAM that no Continuity message will follow from the preceding node because the incoming side bearer has already been established.

The MGW-id can be provided to the succeeding node in the IAM.

13.4.3.5 Establishment of bearer towards the forwarded-to subscriber

The bearer establishment towards the forwarded-to subscriber is performed as described for the mobile originating call, network side bearer establishment, using either forward or backward bearer establishment. The MSC server also requests the MGW to both-way through-connect the bearer.

13.4.3.6 Failure handling in MSC server

The failures are handled as described for the basic mobile originating call.

13.4.3.7 Example

Figure 13.9 shows the network model for call forwarding no reply. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling and the bearer. Note that for a TDM access case there is no separation of call and bearer control signalling. The MSC server replaces the bearer termination for the served mobile subscriber (T_B) with the bearer termination for the forwarded-to subscriber (T_C) in an existing context in the MGW. The bearer termination T_A is used for the bearer towards the preceding MGW (calling subscriber).



Figure 13.9: CFNRy (Network model)

Figure 13.10 shows the message sequence example for the call forwarding on no reply with a possible announcement. In the example, after the call and the bearer towards the access have been released the MSC server requests the MGW to remove the bearer termination for the served mobile subscriber, and optionally requests the MGW to play an announcement and to notify the announcement completion. When the possible announcement has been completed the MSC server requests the establishment of the call towards the forward-to subscriber.



NOTE: OR1: Notification to calling subscriber required (Y:yes N:no)

Figure 13.10: CFNRy (message sequence chart)

13.4.4 Call Forwarding on mobile subscriber Not Reachable (CFNRc)

The procedures specified in 3GPP TS 23.082 [12] for the Call Forwarding on Not Reachable supplementary service shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network.

13.4.4.1 Rerouting by HLR

The same handling as for Call Forwarding Unconditional applies.

13.4.4.2 Rerouting by VLR

If the mobile is not reachable the incoming call for the specified basic service(s) will be forwarded without being offered to the served mobile subscriber.

13.4.4.2.1 MGW selection

If in-band information is to be provided to the calling subscriber the MSC server shall select the MGW before providing the in-band information. The MGW selection can be based on a possibly received MGW-Id from the preceding node.

NOTE: As an implementation option, if in-band information is not to be provided to the calling subscriber the MSC server may either perform call control without any associated MGW, or reserve resources from an MGW and request bearer establishment through that MGW. In the latter case the MSC server selects an MGW for the bearer either before sending the IAM or after receiving the Bearer Information message. If the MSC server received an MGW-Id from the preceding node and/or from the succeeding node, those can be used for the MGW selection.

13.4.4.2.2 Incoming side bearer establishment

The incoming side bearer establishment is handled in the MSC server as described for the mobile terminating call, using either forward or backward bearer establishment. The incoming side bearer establishment can take place either before or after the detection of the not reachable condition.

13.4.4.2.3 Notification to the calling subscriber

If the served mobile subscriber has requested that the calling subscriber shall receive a notification about the call forwarding, a notification is sent to the calling party. If the notification is implemented using intermediate tones or announcements the MSC server requests the MGW to play an announcement/tone to the calling party, as described in clause 14.6, before establishing the call towards the forwarded-to party.

13.4.4.2.4 Initial addressing

The call towards the forwarded-to subscriber is established as a basic call. If out-of-band transcoder control is applied for a speech call, it shall be performed in accordance with 3GPP TS 23.153 [3]. In-band information may be played to the calling subscriber.

If no in-band information is to be played to the calling subscriber, in order to withhold the call completion until the establishment of the bearer is complete the MSC server shall indicate in the IAM that the Continuity message will follow from the preceding node, if either of the following conditions is satisfied before sending the IAM:

- 1. The incoming IAM indicated that the Continuity message will follow, but no Continuity message has been received, or
- 2. The incoming side bearer has not been established.

If the MGW is selected at an early stage the MGW-id can be provided to the succeeding node in the IAM.

13.4.4.2.5 Establishment of bearer towards the forwarded-to subscriber

The bearer establishment towards the forwarded-to subscriber is performed as described for mobile originating call, network side bearer establishment, using either forward or backward bearer establishment. The MSC server also requests the MGW to both-way through-connect the bearer.

13.4.4.2.6 Confirmation of bearer establishment

If the outgoing IAM indicated that a Continuity message will follow, the Continuity message shall be sent when both of the following conditions are satisfied:

- 1. Either:
 - a. The incoming IAM indicated that the Continuity message will follow, and a Continuity message has been received, or
 - b. The incoming IAM indicated that no Continuity message will follow.
- 2. Either:
 - a. The MSC server has selected an MGW, and a notification indicating successful completion of the incoming side bearer set-up has been received from the MGW using the Bearer Established procedure, or
 - b. MGW selection is not required for this call.

13.4.4.2.7 Failure handling in MSC server

The failures are handled as described for the basic mobile originating call.

13.4.4.2.8 Example

Figure 13.11 shows the network model for call forwarding on not reachable. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling and the bearer. The MSC server seizes one context with two bearer terminations in the MGW. The bearer termination T_A is used for the bearer towards the preceding MGW (calling subscriber) and the bearer termination T_C is used for the bearer towards the succeeding MGW (forwarded-to subscriber).



Figure 13.11: CFNRc; Rerouting by VLR (Network model)

Figure 13.12 shows the message sequence example for the call forwarding on not reachable with a possible announcement. In the example the MSC server optionally requests the MGW to play an announcement and to notify the announcement completion, after the bearer to the incoming side has been established. When the possible announcement has been completed the MSC server requests the establishment of the call towards the forward-to subscriber.



Figure 13.12: CFNRc (Rerouting by VLR) (message sequence chart)

13.5 Call Waiting (CW)

The procedures specified in 3GPP TS 23.083 [13] for the Call Waiting supplementary service shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network. The following clauses assume subscriber A to be the served subscriber with the call waiting supplementary service, subscriber B to be the one who is engaged in a call with user A, subscriber C to be the one who has originated a call to subscriber A which causes the call waiting supplementary service to be invoked.

Call confirmation to the waiting call

The MSC server shall, on reception of the call confirmation, select the MGW that will be used for the waiting call. The MSC server should select the MGW which is already in use for the active call. If out-of-band transcoder control is applied for the waiting speech call, it shall be performed in accordance with 3GPP TS 23.153 [3].

Existing call on hold

The clause "Hold request" in clause 13.6 applies.

Existing call released

If the active call is disconnected while another call is waiting, the bearer termination towards the waiting party (C) as well as to the called party (A) is not removed.

Acceptance of waiting call

If the mobile subscriber decides to accept the waiting call, it handles (according to 3GPP TS 23.083 [13]) the existing call as described in clause 13.5 (i.e. it either puts the call on hold or the call is released). When the MSC server receives the connect indication from subscriber A, if required the MSC server shall modify the access bearer as described in subclause 13.18.1. Finally, the MSC server shall connect the access side bearer termination to the previously created bearer termination of the remote party in the waiting call and modify the waiting call's bearer termination so that it is both-way through-connected.

If a different MGW is selected for the incoming call, then a bearer from the new MGW (MGW2) shall be connected towards the old MGW (MGW1) before offering the call to the subscriber A.

If out-of-band transcoder control is applied for the waiting speech call, it shall be performed in accordance with 3GPP TS 23.153[3].

Waiting call released by calling subscriber (subscriber C)

The respective resources already allocated at the selected MGW for the waiting call shall be released.

Example

Figure 13.13 shows the network model for a waiting call at the serving MSC server/MGW. The "thick, squared" line represents the call control signalling for the existing call and, on the Iu interface, the already existing control plane toward the serving RNC. The "thin, squared" line represents the call control signalling for the waiting call. The "thick, dotted" line represents the bearer control signalling and the bearer for the existing call, whereas the "thin, dotted" line represents the ones for the waiting call. Note that for a TDM access there is no separation of call and bearer control signalling.

NOTE: There shall be only one instance of bearer resource/bearer control signalling on the radio side.

If the CW condition applies, the MSC server seizes a new context with one bearer termination, T_c , in the MGW. T_A and T_B are the terminations of the already existing call.



Figure 13.13: Call Wait (network model)

13.5.1 Call Confirmation of the waiting call

Figure 13.14 shows the sequence chart for the actions necessary within the bearer independent CS core network during call confirmation of the waiting call. Call and bearer establishment shall be as described for the mobile terminating call. When the MSC server receives the Alerting indication from the called subscriber (subscriber A), it shall apply the ringing tone to the waiting termination (T_c).



Figure 13.14: Call Confirmation of the Waiting Call.

13.5.2 Acceptance of the Waiting Call

Figure 13.15 shows the sequence chart for the actions necessary within the bearer independent CS core network for the acceptance of a waiting call. When the MSC server receives the Connect indication from the UE (subscriber A) (bullet 1 in figure 13.15), it shall request the MGW to disconnect subscriber C from the applied ringing tone (bullet 2 in figure 13.15) and move T_A to the context of the waiting call (bullet 3 in figure 13.15). The MSC server then requests the MGW to change the through-connection of the termination T_C so that it will be both way through-connected (bullet 4 in figure 13.15).



Figure 13.15: Acceptance of the Waiting Call.

13.6 Call Hold (CH)

The procedures specified in 3GPP TS 23.083 [13] for the Call Hold supplementary service shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network.

13.6.1 Hold request

When the UE makes a request for the hold function the MSC server requests the MGW to interrupt the communication on the bearer by changing the through-connection of the bearer termination towards the served mobile subscriber to "not through-connected" or by using the Isolate Bearer Termination Procedure. Announcements may be applied to the held party as described in clause 14.6.

13.6.2 Retrieval request

When the UE makes a request to retrieve a held call the MSC server requests the MGW to re-establish communication to the held party by changing the through-connection of the bearer termination towards the served mobile subscriber to be both-way through-connected or by using the Join Bearer Termination Procedure.

13.6.3 Setting up another call

The call towards the C party is established as described for the mobile originating call. A new MGW may be selected in the course of setting up the new call. If out-of-band transcoder control is applied for a speech call, it shall be performed in accordance with 3GPP TS 23.153 [3]. If required, the MSC server shall modify the access bearer for the new call as described in subclause 13.18.1. The MSC server will request the MGW to connect the access side bearer termination to the bearer termination of the remote party.

13.6.4 Alternate from one call to the other

When the hold request for the active call is immediately followed by a retrieve request for the held call the MSC server shall request the MGW to connect the bearer termination of the served mobile subscriber to the bearer termination of the held party. The MSC server also requests the MGW to both-way through-connect the bearer for the previously held call.

13.6.5 Disconnect

If the active call is disconnected while another call is on hold, the bearer termination towards the served mobile subscriber is not removed but the call towards the active party is disconnected as described in the clause for call clearing.

If the held call is disconnected while the served mobile subscriber is connected to an active call the bearer termination towards the served mobile subscriber shall not be removed but the call towards the held party is disconnected as described in the clause for call clearing.

13.6.6 Failure handling in MSC server

If any procedure between the MSC server and the MGW has not completed successfully, the MSC server shall reject the hold/retrieve request.

13.6.7 Example 1

Figure 13.16 shows the network model for the call hold with an establishment of a new call. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling and the bearer. The MSC server isolates the flow connection between the served mobile and the termination in the MGW that is used for the held call. If an announcement is to be played to the held party the MSC Server may seize a new context in the MGW. If a new call is made to C subscriber the MSC server seizes a new context in the MGW if not previously done. The new context is used for the new call and the old context is used for the held call. The bearer termination T_A is moved with the bearer towards the RNC (served mobile subscriber) and the bearer termination T_C is used for the bearer towards the succeeding MGW (C-party).

NOTE: The main advantage with example 1 is that if B party is connected to a radio access and a handover occurs then the announcement termination is unaffected and therefore there is no break in the announcement. It is also an option to put the B-party on hold using the isolate bearer procedure immediately but this is not shown in this example.

3GPP TS 23.205 version 9.4.0 Release 9

159



Figure 13.16: Call hold and establishment of a new call (Network model) example 1

Figure 13.17 shows the message sequence for example 1 for the call hold procedure. In the example the MSC server requests the MGW to change the through-connection of the bearer so that it will not be through-connected when the hold request is received from the served mobile subscriber (bullet 2 in figure 13.17). Subsequently if an announcement is to be applied then Termination T_A may be isolated from the held party (bullet 3 in figure 13.17) and a new announcement termination T_{A2} is added to context CTX_{AB} and announcement applied to termination T_{A2} (bullet 3a in figure 13.17). For setting up a new call to the C-party the termination T_A is moved to a new context if not done previously (bullet 4 in figure 13.17) and a new termination T_C is reserved (bullet 5 in figure 13.17) and the call shall continue with the appropriate stream mode settings as for a normal call establishment in accordance with Clause 6.1, see figure 6.2.



Figure 13.17: Hold request (message sequence chart), example 1

Figure 13.18 shows the message sequence for example 1 for the retrieval procedure. In the example the MSC server requests the MGW to move the connection of the bearer back to be both-way through-connected (bullet 3 and 4 in figure 13.18) to the held party, after the held party has been disconnected from an optionally applied announcement (bullet 2 in figure 13.18). The MOVE command (Join Bearer termination procedure) is not required (bullet 3 in figure 13.18) if T_A has not previously been moved into the CTX_{AC}. The Change Through-Connection is only required if the retrieval occurs prior to any connection to C-party.



Figure 13.18: Retrieval request (message sequence chart), example 1

13.6.8 Example 2

Figure 13.18a shows the network model for the call hold with an establishment of a new call. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling and the bearer. The MSC server isolates the flow connection between the served mobile and the termination in the MGW that is used for the held call. After the request to establish a new call is made the MSC server seizes a new context in the MGW, the new context is used for the new call and the old context is used for the held call. The bearer termination T_A for the bearer towards the RNC (served mobile subscriber) is moved to this context and the bearer termination T_C is used for the bearer towards the succeeding MGW (C-party). In this example the announcement is played directly to Termination T_B .

NOTE: if B party is connected to a radio access and a handover occurs during the announcement then the announcement will be interrupted. It is also possible to put B-party on hold using the isolate bearer termination but if the announcement is still played directly to termination T_B as proposed by this example and a handover occurs then the problem still arises.

3GPP TS 23.205 version 9.4.0 Release 9

162



Figure 13.18a: Call hold and establishment of a new call (Network model) example 2

Figure 13.18b shows the message sequence for example 2 for the call hold procedure. In the example the MSC server requests the MGW to change the through-connection of the bearer to inactive when the hold request is received from the served mobile subscriber (bullet 2 in figure 13.18b). Subsequently an announcement may be applied to termination T_B (bullet 3 in figure 13.18b). Then the new call is established via a new context with served mobile subscriber termination T_A being moved to this new context (bullet 4 in figure 13.18b) and a new termination T_C is reserved (bullet 5 in figure 13.18b) and the call shall continue with the appropriate stream mode settings as for a normal call establishment in accordance with Clause 6.1, see figure 6.2.



Figure 13.18b: Hold request (message sequence chart), example 2

Figure 13.18c shows the message sequence example for the retrieval procedure. In the example the MSC server moves the served mobile subscriber back to the context with termination T_B and requests the MGW to change the through-connection of the bearer to be both-way through-connected (bullet 3 and 4 in figure 13.18c) after the held party has been disconnected from an optionally applied announcement (bullet 2 in figure 13.18c). The MOVE command (Join Bearer Termination procedure) is not required (bullet 3 in figure 13.18c) if T_A has not been in the CTX_{AC}. The Change Through-Connection is only required if the retrieval occurs prior to any connection to C-party.



Figure 13.18c: Retrieval request (message sequence chart), example 2

13.7 Multiparty (MPTY)

The procedures specified in 3GPP TS 23.084 [14] for the Multi Party supplementary service shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network. If out-of-band transcoder control is applied for the call, it shall be performed in accordance with 3GPP TS 23.153 [3].

13.7.1 Beginning the Multi Party call

When the served mobile subscriber invokes a Multi Party service the MSC server selects an MGW that provides the Multi Party bridge capabilities. If the selected MGW is different from the MGW that is used for the active call, the MSC server requests the MGW(s) to connect the bearer terminations of the participants to the selected MGW. The bearer terminations are connected together.

13.7.2 Managing an active Multi Party call

When the served mobile subscriber puts the Multi Party call on hold the MSC server requests the MGW to interrupt the connection between the served mobile subscriber and the Multi Party bridge.

When the served mobile subscriber retrieves a held Multi Party call the MSC server requests the MGW to re-establish the connection between the served mobile subscriber and the Multi Party bridge.

When the served mobile subscriber requests private communication with one of the remote parties (e.g. B-party), the MSC server shall request the MGW to interrupt the connection between the served mobile subscriber and the Multi Party bridge, and connection between the remote B party and the Multi Party bridge. The MSC server requests the

MGW to connect the bearer termination of the served mobile subscriber to the bearer termination of the remote party (or vice versa) within a same context.

13.7.3 Disconnect

If a remote party is disconnected while other parties still remain the call towards the remote party is disconnected as described in the clause for call clearing.

13.7.4 Failure handling in MSC server

If resources for the Multi Party service cannot be allocated in any of the MGW resources assigned to the MSC server, then the MSC server shall reject the MPTY request.

13.7.5 Example 1 – Multiparty call establishment

Figure 13.19 shows the network model for multi party call. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling and the bearer. Note that for a TDM access there is no separation between the call and bearer control signalling. In the following example it is assumed that each party participating in the Multi Party conference is handled in a separate context representing the call leg between the Multi Party bridge and the Multi Party participant. The Multi Party bridge itself is handled in a separate context. This separation to several contexts is done in order to simplify interactions with other functionality, such as handover, even though other implementation options are not excluded.



Figure 13.19: Multi Party call (Network model)

For the purposes of the information flow diagrams it is assumed that there are only two remote parties. Party A is the subscriber controlling the Multi Party service (served mobile subscriber). Party B is the held party and party C is the active party.

It is assumed that the Multi Party bridge is located in the MGW that has been selected for the served mobile subscriber.

Figure 13.20 shows the message sequence example for the beginning of multi party call. When the served mobile subscriber invokes a Multi Party service the MSC server requests the MGW to create a separate context for the Multi Party bridge. The MSC server seizes a bearer termination for each party in that context. In addition, each call leg is represented by a separate context. Therefore the parties in the active call will be split in separate contexts. The MSC server requests the MGW to create a new context and to move the bearer termination for the served mobile subscriber from the active call context to the new context. To connect the parties to the Multi Party bridge the MSC server requests the MGW to establish internal Nb connections between the bearer terminations in the Multi Party bridge context and the call leg contexts, using the standard external bearer setup procedures. The held party is informed about the retrieval of the held call, and the both remote parties are informed about the multi party call establishment.



Figure 13.20: Information flow for multi party call, internal ATM bearer (message sequence chart)

UE	2	RNC/BSC	Ĩ	MSC	-S	MG	W
	 Multi P	artyreq					
			Context (\$)		MOV.request (T_A)	
			Context (C _A)		MOV.reply (T_A)		
			Context (\$)		ADD.request (\$ ADD.request (\$ ADD.request (\$	\$) 5) 5)	Establish bearer Establish bearer Establish bearer
			Context (C _{MF}	TY)	ADD.reply (T _{A0} ADD.reply (T _{B0} ADD.reply (T _{C0})))	
			Context (C _{MP}	ry)]]]	NOTIFY.request (NOTIFY.request (NOTIFY.request ($ \begin{array}{c} \mathbf{T}_{A0} \\ \mathbf{T}_{B0} \\ \mathbf{T}_{C0} \end{array} \right) \\ \mathbf{T}_{C0} \end{array} $	Tunnel Information Up Tunnel Information Up Tunnel Information Up
			Context (C _{MP}	_{TY})	NOTIFY.reply (1 NOTIFY.reply (1 NOTIFY.reply (1	$ \begin{bmatrix} A_0 \end{bmatrix} $ $ \begin{bmatrix} A_0 \end{bmatrix} $ $ \begin{bmatrix} B_0 \end{bmatrix} $ $ \begin{bmatrix} C_0 \end{bmatrix} $	
			Context (C _A)	-	ADD.request (\$	5)	Prepare bearer + Tunnel
			Context (C _A)	-	ADD.reply (T_{A1})	Information Down
			Context (C _A)	I	NOTIFY.request ((T_{A1})	Tunnel Information Up
			Context (C _A)	-	NOTIFY.reply (7	(A_{A1})	
			Context (CAB)	_	ADD.request (\$	5)	Prepare bearer + Tunnel
			Context (C _{AB})		ADD.reply (T_{B1})	Information Down
			Context (C _{AB})	I	NOTIFY.request ((T_{B1})	Tunnel Information Up
			Context (C _{AB})	-	NOTIFY.reply (7	B1)	
			Context (C _{AC})	, -	ADD.request (S	5)	Prepare bearer + Tunnel
			Context (C_{AC})	-	ADD.reply (T _C)	Information Down
			Context (CAC)]	NOTIFY.request ((\mathbf{T}_{c_1})	Tunnel Information Up
			Context (C_{AC})	-	NOTIFY.reply (7	[_{C1})	- union morning of
			Context (C _{MP}	TY)	MOD.request (T MOD.request (T MOD.request (T	A0) B0) C0)	Tunnel Information Down Tunnel Information Down Tunnel Information Down
			Context (C _{MP}	FY)	MOD.reply $(T_{A0}$ MOD.reply $(T_{B0}$ MOD.reply $(T_{C0}$)))	
					Call Progre	ss (to B)	
				-	Call Progre	ss (to B)	
	- Mult	i Party acknowle	dge	-	Call Progre	<u>ss (to C)</u>	

Figure 13.20a: Information flow for multi party call, internal IP bearer (message sequence chart)

13.7.6 Example 2 – Private call establishment during multiparty call

Figure 13.20b shows the network model for private call. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling and the bearer. Note that for a TDM access there is no separation between the call and bearer control signalling.



Figure 13.20b: Private call (Network model)



Figure 13.20c: Information flow for private call (message sequence chart)

NOTE: The two MOVE commands may be sent in different transactions.

For the purposes of the information flow diagrams it is assumed that there are only two remote parties and the private call is established between the served mobile subscriber (Party A) and the remote C-party.

Figure 13.20c shows the message sequence example for the beginning of private communication between the served mobile subscriber and the remote parties (e.g. C-party).

13.8 Closed User Group (CUG)

No impact.

13.9 Advice of Charge (AoC)

No impact.

13.10 User-to-User Signalling (UUS)

No impact.

13.11 Call Barring Services

13.11.1 Barring of outgoing calls

No impact.

13.11.2 Barring of incoming calls

No impact.

13.12 Explicit Call Transfer (ECT)

The procedures specified in 3GPP TS 23.091 [15] for the Explicit Call Transfer supplementary service shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network.

Party A is the subscriber controlling the Explicit Call Transfer Call (served mobile subscriber). Party B is the first remote called party (held party). Party C is the second remote called party.

13.12.1 Connection of remote parties

If the result of the ECT checks is successful the MSC server will order the MGW to connect the bearer termination of the C-party to the bearer termination of the B-party (bullet 1 in figure 13.24 or in figure 13.21). As a result of this action the held party will be retrieved.

If the call towards the C-party has not been answered, the MSC server requests the MGW to both-way through-connect the bearer termination towards the C-party.

13.12.2 IU/A-interface release

The served party is disconnected after a successful transfer request. The call towards the served mobile subscriber shall be released as described in the clause for call clearing.

13.12.3 Failure handling in MSC server

If the bearer terminations for the remote parties can not be connected successfully, the MSC server shall reject the ECT request.

13.12.4 Example

Figure 13.23 shows the network model for call explicit call transfer. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling and the bearer. Note that for a TDM access case there is no separation of call and bearer control signalling. The MSC server moves the bearer terminations of the remote parties in the same context and removes the bearer termination for the served mobile subscriber. The bearer termination T_A is used for the bearer towards the served mobile subscriber, the bearer termination T_B is used for the bearer towards the served mobile subscriber, the bearer termination T_B is used for the bearer termination T_C is used for the bearer towards the C-party.



After ECT:



Figure 13.23: ECT (Network model)

Figure 13.24 shows the message sequence example for the explicit call transfer when both calls have been answered. In the example the MSC server requests the MGW to move the bearer termination for the C-party in the active call to the same context which contains the bearer termination for the B-party. The held party is informed about the retrieval of the held call. Both the remote parties are informed about the call transfer. After the move the MSC server releases the call and the bearer connection towards the served mobile subscriber and requests the MGW to remove the bearer termination for the served mobile subscriber.



Figure 13.24: Explicit call transfer; both calls answered (message sequence chart)

Figure 13.25 shows the message sequence example for the explicit call transfer when one call is answered and the other call has been delivered. In the example the MSC server requests the MGW to move the bearer termination for the C-party in the active call to the same context which contains the bearer termination for the B-party. The held party is informed about the retrieval of the held call. Both the remote parties are informed about the call transfer. After the move the MSC server releases the call and the bearer connection towards the served mobile subscriber and requests the MGW to remove the bearer termination for the served mobile subscriber. The B-party is informed about the active call when the C-party sends the Answer indication.



Figure 13.25: Explicit call transfer; other call delivered (message sequence chart)

13.13 Completion of Calls to Busy Subscriber (CCBS)

The procedures specified in 3GPP TS 23.093 [16] for the Completion of Calls to Busy Subscriber supplementary service and 3GPP TS 23.108 [18] shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network.

13.13.1 Clearing when tones/announcements are provided to the calling subscriber

If an announcement is to be provided for the purpose of notifying the subscriber that CCBS activation is possible, the MSC server shall select an MGW. The MGW performs the traffic channel assignment if the bearer termination has not been through-connected (as described in clause 6.1 for the basic mobile originating call). The MGW through-connects the bearer before providing the in-band information. The MSC server requests the MGW to play an announcement/tone using the Play Announcement or Send Tone procedure. When the announcement has completed the MGW notifies the MSC server (using the Announcement Completed procedure as described in clause 14.6) that the announcement is complete.

Otherwise the MSC server handles the call clearing as described in clause 7.

13.13.2 Network initiated mobile originated call

The call is established as described in clause 6.1 for basic mobile originating call.

13.13.2.1 Early Traffic Channel Assignment

Within CCBS there is an option for a CCBS call to establish a bearer before setup in state "CC-establishment confirmed". In this case the MSC server shall check whether an access bearer assignment modification has to be performed after receiving the setup message from UE.

13.13.2.1.1 Example

For the network model, please refer to figure 6.1.

Figure 13.26 shows the message sequence chart for the network initiated mobile originating call using the option assignment after A and B party alerting. In the following, the case with backward bearer establishment is considered.



Figure 13.26: Network initiated mobile originating call establishment with assignment after A and B party alerting (message sequence chart)

13.13.3 CCBS Information conveyed by Call Signalling

For CCBS, application specific information needs to be conveyed via the call signalling. Specific details of the CCBS information are described in 3GPP TS 23.093 [16].

13.14 Multiple Subscriber Profile (MSP)

No impact.

13.15 Multicall

The procedures specified in 3GPP TS 23.135 [17] for the Multicall supplementary service shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network. If out-of-band transcoder control is applied for a speech call, it shall be performed in accordance with 3GPP TS 23.153 [3].

13.15.1 Mobile Originating

When the UE is in active mode and it makes a request for the multicall function on a new traffic channel, call and bearer establishment shall be as described for mobile originating call.

When the UE is in active mode and it makes a request for the multicall function using an existing traffic channel, call and bearer establishment shall be as described for call hold function. An active call will be placed on hold and the additional originating call will be initiated.

13.15.2 Mobile Terminating

When the UE is in active mode and it makes a request for the multicall function on a new traffic channel, call and bearer establishment shall be as described for mobile terminating call. Access bearer assignment shall occur either after a Call Confirmed or a Connect message is received from the UE.

When the UE is in active mode and it makes a request for the multicall function using an existing traffic channel, call and bearer establishment shall be as described for call hold function. An active call will be placed on hold and the additional terminating call will be initiated.

13.16 Calling Name Presentation (CNAP)

No impact.

13.17 Alternate Speech/Fax

The procedures for facsimile group 3 transparent/non-transparent shall be followed in accordance with 3GPP TS 43.045 [24] and 3GPP TS 23.146 [25]. The following clauses describe the additional requirements for the bearer independent CS core network. If out-of-band transcoder control is applied for a speech call, it shall be performed in accordance with 3GPP TS 23.153 [3].

Call and bearer establishment shall be handled as described in the Call Establishment clause. In order to change from speech to fax (or vice versa), the MSC server shall modify the access bearer as described in subclause 13.18.1.

If the MGW responds with an error to any of the procedures initiated by the MSC server, or the MSC server receives a Bearer Failure procedure from the MGW, the MSC server may either clear the call or reject the change from speech to fax (or vice versa).

After this possible modification, the MGW shall seize an interworking function if a PLMN Bearer Capability [4] has been supplied to the access side bearer termination. When the MSC server receives an answer indication, it shall request activation of the interworking function using the Activate Interworking Function procedure.

13.18 Modification of the Access Bearer

13.18.1 Modification of Bearer Characteristics

The modification of the access bearer is possible during a call establishment and during an active call. If the MSC server needs to modify the access bearer, the existing bearer termination in the MGW is modified or a new access side bearer termination is created. The modification of the access bearer shall be performed in accordance with 3GPP TS 25.413 [26] or 3GPP TS 48.008 [27].

13.18.1.1 lu mode

If the link characteristics for the existing access bearer need to be changed and the MSC server previously received a notification from the MGW that modification of link characteristics of the current transport connection is supported [refer to 26], the MSC server shall use the Modify Bearer Characteristics procedure to provide the MGW with the new bearer characteristics for the existing access side bearer termination. After the MGW has replied, the MSC server shall initiate the access bearer modification towards RAN.

If the MSC server has not previously received a notification from the MGW that modification of existing link characteristics is supported, the MSC server shall use the Prepare Bearer procedure to request the MGW to add a new context and a new access side bearer termination, and to provide a bearer address and a binding reference. After the MGW has replied, the MSC server shall initiate the access bearer modification towards RAN using the provided bearer address and the binding reference. Upon successful access bearer modification, the MSC server shall connect the new access side bearer termination to the old context and release the old access side bearer termination.

If the user plane mode of the modified access bearer is "Support Mode", the Iu UP will also be re-initialised as defined in [20].

13.18.1.2 A/Gb mode

The MSC server shall use the Modify Bearer Characteristics procedure for A interface TDM termination and may use the Modify Bearer Characteristics procedure or the Configure RTP Connection procedure for AoIP termination to the MGW to provide the new bearer characteristics for the existing access side bearer termination. After the MGW has replied, the MSC server shall initiate the access bearer modification towards GERAN.

13.18.2 IWF Protocol Change

If the MSC server has requested indication on IWF protocol events, the MGW informs the MSC server about changes related to IWF protocol, using the Protocol Negotiation Result and Rate Change procedures.

For AoIP terminations there shall be no change to the AoIP Transport Layer Address (MGW) and AoIP Transport Layer Address (BSS) for Rate Changes in data calls. Instead the MSC shall include the previously returned Local Connection Address in any subsequent Assignment request to the BSS.

13.19 GSM Fax

The procedures for facsimile group 3 transparent service toward a GSM BSS shall be followed in accordance with 3GPP TS 43.045 [24]. The following clauses describe the additional requirements for the bearer independent CS core network.

Call and bearer establishment shall be handled as described in the Call Establishment clause with the addition that the Rate Change Event Notification shall be requested. If the MGW detects a mismatch between the radio channel rate to the facsimile transmission speed then it shall indicate this to the MSC server by use of the rate change procedure. The MSC server shall initiate a new Assignment towards the GSM RAN and if successful modify the PLMN Bearer Capability and GSM Channel Coding at the associated access bearer termination using the Modify Bearer Characteristics procedure as described in 16.2.41 to signal to the MGW that fax transmission can continue. The sequence is shown in Figure 13.19/1.



Figure 13.19/1: GSM access Channel Mode Modify message

For AoIP terminations the Configure RTP Connection Point Procedure may be used instead of the Modify Bearer Characteristics procedure.

13.20 Voice group call service (VGCS), Voice broadcast service (VBS)

13.20.1 Beginning the Voice group call

The procedures specified in 3GPP TS 43.068 [39] for the Voice group call service shall be followed. For a Voice broadcast call the procedures in 3GPP TS 43.069 [40] shall be followed. When the served mobile subscriber invokes a Voice group call service the MSC server selects an MGW that provides the conference bridge capabilities with enhancements to support Voice group calls. The MSC server shall estimate the size of the Conference terminations needed for the VGCS call from the data retrieved from GCR and reserves the requested number of Conference terminations for that call. The number of needed conference terminations is provided within prepare bearer procedure.

13.20.2 Talker change in Voice group call (1 channel model)

If the MSC server decides that the talker shall use the uplink associated to the downlink of the ASCI broadcast (common group call) channel (1channel model, see 3GPP TS 43.068 [39]), the talkers speech is transferred via the uplink channel of the ASCI broadcast channel of the cell. The stream arrives within the MGW on the termination of the traffic channel serving the cell or BSS (enhanced A-interface) and is transferred to the conference function where the summarized conference signal is calculated. The resulting stream is then transferred to all conference terminations. The normal behaviour of a conference bridge is to substract for each conference member its own speech (incoming stream) from the summarized signal otherwise it hears itself (outgoing stream). Therefore the Talker context need to be modified by adding a new termination to the context which is connected to the listener context (see figure 13.20.5/3 and 13.20.5/5). The topology within the talker context is set oneway from uplink of the ASCI broadcast channel in direction to Conference bridge (old termination of the talker context) and the topology is set to Oneway from the new added termination towards the termination connected to the downlink of the ASCI broadcast channel. This is performed using establish bearer procedure and change flowdirection. To avoid, that the talker hear himself, the UE internally switch off the loudspeaker (muting/unmuting functionality, 3GPP TS 43.068 [39]) on request of the MSC server.

13.20.3 Disconnect

A dispatcher can leave or reenter the call at any time without any influence of the ongoing call. Only the dispatcher is disconnected as described in the clause for call clearing (see 3GPP TS 43.068 [39], 3GPP TS 43.069 [40]).

The Voice group call can only be released by an authorised subscriber (see 3GPP TS 43.068 [39], 3GPP TS 43.069 [40]). If an authorised subscriber triggers the release of the group call all other parties are disconnected as described in the clause for call clearing.

13.20.4 Failure handling in MSC server

If resources for the Voice group call service cannot be allocated in any of the MGW resources assigned to the MSC server (e.g. not enough conference terminations are available with cause "510 - Insufficient resources"), then the MSC server shall reject the VGCS request.

13.20.5 Example

Figure 13.20.5/1 shows a possible network model of an anchor MSC for a voice group call when the Talker is on the dedicated channel. At call setup Dispatchers, R-MSC and the talker can been seen as a multi party call. The listners are connected oneway in their context (bullet 1 and 2 in figure 13.20.5/2). In the example every kind of termination is shown in its own context.



Figure 13.20.5/1: VGCS call (termination overview for talker on dedicated channel)

If the talker is on a dedicated channel, the talker context contains two terminations which are connected bothway a termination overview is given in figure 13.20.5/1.

Figure 13.20.5/2 shows the message flow establishment of a VGCS call with two cells and the talker is located in cell1 when the Talker is on the dedicated channel. To connect the parties to the VGCS Multi Party Bridge the MSC server requests the MGW to establish internal Nb connections between the bearer terminations in the VGCS Multi Party bridge context and the call leg contexts, using the standard external bearer setup procedures. VGCS call acknowledge (bullet 3 in figure 13.20.5/2) should be send after the ASCI broadcast channel is established in the cell where the talker is located (see 3GPP TS 43.068 [39], 3GPP TS 43.069 [40]).



Figure 13.20.5/2: Talker on dedicated channel (Message sequence chart)

Figure 13.20.5/3 shows a possible network model of an anchor MSC for a voice group call when the Talker is on one channel mode.



Figure 13.20.5/3: VGCS call (termination overview for Talker on one channel model)

Figure 13.20.5/4 shows the message flow establishment of a VGCS call with two cells and the talker is located in cell1 when the Talker is on one channel mode.


Figure 13.20.5/4: Talker on one channel mode (Message sequence chart)

Figure 13.20.5/5 shows the message flow of the talker move to uplink.

If the MSC server decides that the talker shall use the uplink of the cell. The MSC server adds a termination to the talker context which is connected to the Listener context and changes the topology of the talker context (bullet 1 and 2 in figure 13.20.5/5). Substract of the termination of the talker is only performed when the MSC server decides to move an active talker to the ASCI broadcast channel (bullet 3 in figure 13.20.5/5). The termination towards the cell where the talker is located is moved to the talker context. Further talker changes or cell changes of the current talker would be handled with move commands related to the termination towards the cell where the talker is located (bullet 4 in figure 13.20.5/5). Thus the talker's speech to all the listeners in that cell is transmitted (broadcast channel). The termination overview after the talker is moved to the uplink is given in figure 13.20.5/3.



Figure 13.20.5/5: Information flow talker move to uplink (Message sequence chart)

14 Interactions with Other Network Features and Services

NOTE: All message sequence charts in this clause are informative examples.

14.1 Customised Applications for Mobile network Enhanced Logic (CAMEL)

If the gsmSRF is co-located with the (G)MSC server, the gsmSRF is divided into a gsmSRF server and an MGW. The gsmSRF server terminates the CAP protocol and signals over the Mc interface to instruct its MGW to provide the required resource. All the logic of the gsmSRF is located in the gsmSRF server. The MGW provides only simple resources for playing a single announcement or tone, or detection of single DTMF tone pair. If one single resource in the MGW does not fulfil the requirement of the gsmSCF, the gsmSRF server has to use different resources in sequence to fulfil the whole requirement.

The gsmSSF uses the capabilities of the (G)MSC server and the MGW to play announcements or send tones to the server.

- NOTE 1: In the subsequent figures within clause 14.1, the "Connect To Resource" scenario is used. However the other CAMEL Intelligent Peripheral (IP) scenarios are not intended to be excluded. No impacts are identified when applying these other CAMEL scenarios.
- NOTE 2: The gsmSRF functionality may be deployed within the MSC server, and either the current serving MGW or any MGW resource under the control of the current MSC server.

14.1.1 Play Announcement/Send Tone

The playing of an announcement or sending of a tone shall be performed in accordance with 3GPP TS 23.078 [10]. It is assumed that the MGW selected for the call has the capabilities to provide announcements and tones.

When the gsmSCF requests the gsmSRF to play a specified announcement or tone, the gsmSRF orders the MGW to play the announcement or tone as described in clause 14.6.

After the gsmSRF has received the announcement or tone completed notification from its MGW, it reports the announcement or tone completion to the gsmSCF.

If the gsmSCF requests the gsmSRF to cancel the earlier started announcement or tone, the gsmSRF orders the MGW to stop playing the announcement or tone as described in clause 14.6.

14.1.1.1 Example of playing announcement by the gsmSRF



Figure 14.1: CAMEL Announcement Playing (message sequence chart)

14.1.2 User Interaction

The user interaction shall be performed in accordance with 3GPP TS 23.078 [10]. It shall be assumed that the MGW selected for the call has the capabilities to provide announcements. In bearer independent CS core network the DTMF digits can be propagated inband or out-of-band.

14.1.2.1 Play announcement

When the gsmSCF requests the gsmSRF/SSF to play a specified announcement and to collect digits that are sent by the user the gsmSRF/SSF requests the MGW to play the announcement as described in clause 14.6.

14.1.2.2 Detect DTMF tones

The gsmSRF/gsmSSF starts detecting DTMF tones, as describes in clause 14.4.2, before it receives the announcement or tone completed notification (see clause 14.6).

14.1.2.3 Report DTMF tones

The DTMF tones are reported to the gsmSRF/SSF as described in clause 14.4.2. After all requested digits are received the gsmSRF/SSF reports the digits to the gsmSCF.

14.1.2.4 Cancel prompt and collect user information

If the gsmSCF requests the gsmSRF to cancel the prompt and collect user information procedure, which had been started earlier, the gsmSRF orders the MGW to stop playing the announcement or sending tone, if they are still in progress, using the Stop Announcement or the Stop Tone procedure. The gsmSRF shall also order the MGW to stop detecting DTMF tones using the Stop DTMF Detection procedure.



Figure 14.2: CAMEL User Interaction (message sequence chart)

NOTE: Since gsmSRF don not know whether DTMF digits are provided inband or out-of-band the gsmSRF has to be able to collect DTMF tones both inband and out-of-band.

14.1.3 Call Party Handling (CPH)

The procedures specified in 3GPP TS 23.078 [10] for Call Party Handling (CPH) shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network.

In contrast with HOLD and MPTY, the call parties created on instruction from the gsmSCF are not seen as separate calls in the Mobile Station, i.e. all call parties in a CPH configuration use the same *transaction id* towards the MS. In addition, CPH may take place in an MSC-S different from the one where the CAMEL served subscriber is registered. Furthermore, in CPH it is possible to have multiple call parties in separate *call segments* whereas the call hold supplementary service has a limit of one held call party and one active call.

The gsmSCF always triggers the elementary procedures which are described in this subclause. CPH elementary procedures can be used in more complex procedures to provide useful services, but the more complex procedures are out of scope of this specification.

NOTE: For simplicity, the figures below which show network models do not show the gsmSCF. The gsmSCF is in the HPLMN of the served (CAMEL) party. The GMSC-S is in the interrogating network (IPLMN). The MSC-S is in the VPLMN of the served party.

Use of a multi-party (conference) bridge

When the gsmSCF invokes a CPH procedure which requires the connection of three or more legs in a multi-party configuration, the MSC server selects an MGW which provides multi-party bridge capabilities. The timing of the selection of the MGW with the multi-party bridge capabilities is vendor specific. If the selected MGW is not the MGW which is used for the active call, the MSC server requests the MGW(s) to connect the bearer terminations of the participants to the selected MGW. The bearer terminations are connected together.

14.1.3.1 Call Party Handling concepts

The relationship between Call Segments and voice connections is explained in 3GPP TS 23.078 [10] subclause 4.5.1.

14.1.3.2 Initiate Call Attempt procedure

The Initiate Call Attempt (ICA) procedure is used either:

- To create a new call (out-of-the-blue), in which case the gsmSCF makes the initial contact with the MSC-S, or
- To create an additional call party in an existing call. The new call party is always created in a new call segment. The existing call may have triggered contact with the gsmSCF based on CAMEL subscription information (MO or MT in VMSC, MT in GMSC, call forwarding etc), or the gsmSCF may have initiated the contact using the ICA procedure (out-of-the-blue).

The gsmSCF may create additional call parties before it establishes the bearer to the calling party. The MSC-S/GMSC-S shall establish bearers to the additional call parties independently of the other parties, including the calling party.

The leg which is created by the Initiate Call Attempt procedure is initially in the held state.

14.1.3.2.1 Example

Figure 14.1.3/1 shows an example network model for the Initiate Call Attempt procedure with an establishment of a new call leg. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling and the bearer. The MSC-S seizes a new context with one bearer termination in the MGW which is used for the new call leg.



Figure 14.1.3/1 Initiate Call Attempt procedure (Network model)

Figure 14. 1.3/2 shows an example message sequence for the Initiate Call Attempt procedure.

In this example a new call leg to the C-party is established.



Figure 14. 1.3/2 Information flow for Initiate Call Attempt (message sequence chart)

14.1.3.3 Move Leg procedure

The Move Leg procedure is used to move a leg from its current call segment to the (existing) target call segment.

Using Move Leg to add a leg to a call segment which already includes 2 call legs requires the establishment of a multiparty call (if it does not already exist for the served CAMEL subscriber) as described in clause 13.7. Other call parties may be involved in independent Multiparty calls due to MPTY SS or CPH. If the call segment to which the specified leg is added is already using a multi-party bridge, the MSC server requests the MGW to establish the connection between the specified leg and the multi-party bridge.

Example

Figure 14. 1.3/3 shows an example network model for the Move Leg procedure. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling and the bearer



Figure 14. 1.3/3 Move Leg procedure (Network model)

Figure 14. 1.3/4 shows an example message sequence for the Move Leg procedure.

In this example the leg of the C-party is moved to an existing call segment. The MSC-S requests the MGW to move the bearer termination for the C-party to the same Context which contains the bearer termination for the A-party



Figure 14. 1.3/4 Information flow for Move Leg (message sequence chart)

14.1.3.4 Split Leg procedure

The Split Leg procedure is used to separate a call leg from a source call segment and place it in a (new) target call segment.

When the gsmSCF uses the Split Leg procedure to put a call leg on hold, the MSC server instructs the MGW to interrupt the connection between the specified call leg and the other party/parties in the call segment. If the call segment is using a multi-party bridge, the connection from the specified call leg to the multi-party bridge is interrupted.

Example

Figure 14. 1.3/5 shows an example network model for Split Leg procedure. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling and the bearer.



Figure 14. 1.3/5 Split Leg procedure (Network model)

Figure 14. 1.3/6 shows an example message sequence for the Split Leg procedure.

In this example the leg of the B-party is moved to a new call segment. The MSC-S requests the MGW to move the bearer termination for the B-party in the active call to a new context.





14.1.3.5 CAMEL User interaction procedure

In accordance with 3GPP TS 23.078 [10] the gsmSCF may order the MSC-S/gsmSSF to play an announcement or control user interaction as specified in subclauses 14.1.1 and 14.1.2 respectively. The tones are provided in accordance with subclause 14.6 of the present document. As part of Call Party Handling, announcements or tones can be played to an individual party or to all the parties connected in the call segment.

The gsmSCF may also instruct the MSC-S to establish a temporary connection to an external device which provides the user interaction.

14.1.3.6 Failure handling in the MSC-S

If resources for Call Party Handling cannot be allocated in any of the MGWs assigned to the MSC-S, then the MSC-S shall reject the request for the Call Party Handling procedure.

14.2 IST

The handling of IST shall be performed in accordance with 3GPP TS 42.032 [19]. This clause describes the additional requirements for the Bearer Independent CS Core Network.

The clearing of calls due to IST is the same as for (G)MSC server initiated call clearing, refer to clause 7.3,(G)MSC server Initiated.

14.3 Operator Determined Barring (ODB)

NOTE: The subsequent clauses in 14.3 describe the impacts of "Barring of Outgoing Calls" and "Barring of Incoming Calls" on Bearer Independent CS CN. Other flavours of Operator Determined Barring may be supported by the Bearer Independent CS CN. However no impacts caused by these other flavours are identified.

14.3.1 Barring of Outgoing Calls

If the mobile station attempts to connect to an address determined to be barred by the Operator Determined Barring service, the call shall be cleared as described in clause 7, Call Clearing.

Otherwise, the call is established as described in clause 6, Call Establishment.

14.3.2 Barring of Incoming Calls

If the incoming call to the mobile station is determined to be barred by the Operator Determined Barring service, the call shall be barred. Otherwise the call shall be delivered as described in clause 6, Call Establishment.

If the GMSC connects the call to a recorded announcement due to Operator Determined Barring, the GMSC server selects the MGW before providing the in-band information. It is possible that the MGW selection is based on an MGW-Id received from the preceding node.

The incoming side bearer establishment is handled in the GMSC server as described for the mobile terminating call using either forward or backward bearer establishment.

In-band information may be provided to the calling subscriber only when both of the following conditions are satisfied:

- 1. Either:
 - a. The incoming IAM indicated that the Continuity message will follow, and a Continuity message has been received, or
 - b. The incoming IAM did not indicate that the Continuity message will follow;
- 2. Notification indicating successful completion of the incoming side bearer set-up has been received from the MGW using the Bearer Established procedure.

The GMSC server provides the MGW with the announcement identification and requests the MGW to notify the announcement completion using the Play Announcement procedure as described in clause 4.6.

After the possible announcement has been completed the GMSC server initiates the call release as described in the clause 7, Call Clearing.

14.4 DTMF

DTMF information can be transported either inband or out of band. In order to minimise the interworking between out of band and in band DTMF signalling, the general principle is to use the DTMF signalling method of the preceding node whenever possible. A node supporting OoB DTMF shall also be able to receive inband DTMF digits, but no DTMF digits shall be duplicated, i.e. any detected digit shall either be sent forward by inband or out-of-band, but never by both methods. Transitions between inband and out-of-band may occur due to changes to an ongoing call (Explicit Call Transfer for example) but digits shall not be sent both inband and OoB for the same link.

If out-of-band transcoder control is applied for a speech call, it shall be performed in accordance with 3GPP TS 23.153 [3]

14.4.1 DTMF Tone Generation

14.4.1.1 Inband DTMF Tone Generation

This option uses inband signalling to transport DTMF digits in the core network.

The DTMF tone generation shall be performed in accordance with 3GPP TS 23.108 [18]. The following clauses describe the additional requirements for the bearer independent CS core network.

14.4.1.1.1 Start DTMF

When the MSC server receives the Start DTMF message from the UE, it uses the Send DTMF procedure to request the MGW to modify the bearer termination to play a tone for the pressed digit. The result of the tone sending by the bearer termination will be received by the MSC server and sent to the UE (bullet 1 in figure 14.3).

14.4.1.1.2 Stop DTMF

When the MSC server receives the Stop DTMF message from the UE, it uses the Stop DTMF procedure to request the MGW to modify the bearer termination to stop digit playing. When the response is received from the MGW, the MSC server will acknowledge the Stop DTMF (bullet 2 in figure 14.3).

The MGW shall check the minimum duration and minimum interval in accordance with the DTMF timing defined in TS 23.014 [33].

14.4.1.1.3 Example

Figure 14.3 shows an example where out-of-band signalling of DTMF information is not supported by the call control protocol. When the UE sends Start DTMF and Stop DTMF messages , the MSC server uses resources in the MGW to generate tones by modifying the bearer termination.



Figure 14.3: Inband DTMF generation (message sequence chart)

14.4.1.2 Out-of-Band DTMF Tone Generation

This option uses out-of-band network signalling to transport DTMF digits in the core network, where the information is sent on a call control layer.

The DTMF Tone Generation shall be performed in accordance with 3GPP TS 23.108 [18]. The following clauses describe the additional requirements for the bearer independent CS core network.

14.4.1.2.1 Start DTMF

When the MSC server receives a Start DTMF message from the UE, it indicates digit playing using out-of-band signalling. The corresponding result received from the preceding/succeeding node will be sent to the UE (bullet 1 in figure 14.4).

14.4.1.2.2 Stop DTMF

When the MSC server receives a Stop DTMF message from the UE, it indicates stop digit playing using out-of-band signalling. The succeeding node will indicate that digit playing is stopped. The MSC server will send the result back to the UE (bullet 2 in figure 14.4).

14.4.1.2.3 Example

Figure 14.4 shows the message sequence example for the out-of-band DTMF during a call. When the MSC server receives the Start DTMF and Stop DTMF messages from the UE, it shall send the information using signalling on call control layer. The MSC server will not use any dedicated resources of the MGW.



Figure 14.4: Out-of-Band DTMF generation (message sequence chart)

14.4.2 DTMF Detection

14.4.2.1 Inband DTMF Detection

The (G)MSC server/gsmSSF/gsmSRF requests the MGW to detect DTMF tones using Detect DTMF procedure (bullet 1 in figure 14.5).

At detection of the DTMF tone the MGW reports the digit to the (G)MSC server/gsmSSF/gsmSRF using the Report DTMF procedure (bullet 2 in figure 14.5). At reception of the DTMF tone report the (G)MSC server/gsmSSF/gsmSRF either expects the MGW to detect other DTMF tones (in which case no new Detect DTMF request needs to be sent) or requests the MGW to stop the detection of DTMF tone (bullet 3 in figure 14.5) using the Stop DTMF Detection procedure.



Figure 14.5: Inband DTMF detection (message sequence chart)

14.4.2.2 Out-of-Band DTMF Detection

The (G)MSC server/gsmSSF/gsmSRF starts collecting out-of-band DTMF tones. One DTMF tone consists of Start DTMF (bullet 1 in figure 14.6) and Stop DTMF messages (bullet 2 in figure 14.6).



Figure 14.6: Out-of-Band DTMF detection (message sequence chart)

14.5 OR

The procedures specified in 3GPP TS 23.079 [11] for the Optimal Routeing network service shall be followed. The following clauses describe the additional requirements for the bearer independent CS core network.

14.5.1 Optimal routeing for basic mobile-to-mobile calls

The optimally routed call from one mobile subscriber to another mobile subscriber is established as a normal basic call.

14.5.2 Optimal routeing for conditional call forwarding; Early call forwarding

For early call forwarding the same procedures as described for CFU and CFNRc (rerouting by HLR) shall apply.

14.5.3 Optimal routeing for conditional call forwarding; Late call forwarding

14.5.3.1 MSC server

14.5.3.1.1 Resume Call Handling and clearing of connection to GMSC server

When the MSC server determines that the call should be forwarded because the called mobile subscriber is busy (NDUB, UDUB), not reachable or has not replied to the call before the no-reply timer has expired, the MSC server sends a request to resume call handling to the GMSC server.

If the GMSC server determines that the call can be forwarded to the forwarded-to destination it sends a Release message to the MSC server. If no bearer has been established yet the MSC server handles the release only on call control level. If the bearer had been established, the MSC server handles the network side bearer release as described in the clause for the call clearing.

14.5.3.1.2 IU release

When the MSC server determines that the call should be forwarded because the called mobile subscriber is busy (UDUB) or it has not replied to the call before the no-reply call timer has expired, the MSC server shall release the call and bearer connection to the served mobile subscriber as described in the clause for call clearing.

14.5.3.2 GMSC server

14.5.3.2.1 Resume Call Handling and Clearing of Connection to visited MSC server

If the GMSC server determines that the call can be forwarded to the forwarded-to destination it sends a Release message to the MSC server and handles the outgoing side bearer release as described in the clause for call clearing, if the bearer had already been established.

14.5.3.2.2 MGW selection

The GMSC server shall select an MGW for the bearer connection as described for the CFU and CFNRc (in HLR) supplementary services, if not already selected by the mobile terminating call procedures.

14.5.3.2.3 Incoming side bearer establishment

The bearer establishment towards the preceding MGW is handled in the GMSC server as described for the mobile terminating call, if not already established by the mobile terminating call procedures.

14.5.3.2.4 Notification to the Calling Subscriber

The GMSC server sends the possible notification towards the calling subscriber according to the procedures described for the CFU and CFNRc (in HLR) supplementary services.

14.5.3.2.5 Establishment of call and bearer towards the forwarded-to subscriber

The GMSC server establishes the call and bearer towards the forwarded-to subscriber according to the procedures described for the CFU and CFNRc (in HLR) supplementary services.

14.5.3.2.6 Example

Figure 14.7 shows the network model for optimal routeing when no bearer has been established before the invocation of late call forwarding. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling and the bearer. The GMSC server seizes one context with two bearer terminations in the MGW. The bearer termination T_A is used for the bearer towards the preceding MGW (calling subscriber) and the bearer termination T_C is used for the bearer towards the succeeding MGW (forwarded-to subscriber).



Figure 14.7: Optimal routeing; late call forwarding (CFB (NDUB), CFNRc) (Network model)

Figure 14.8 shows the message sequence example for the optimal routeing with late call forwarding without any notification to the calling party. In the example below no bearer has been established for the connection when the MSC server sends the Resume Call Handling request to the GMSC server. After the call towards the visited MSC server has been released the GMSC server establishes the call and the bearer as described for Call Forwarding Unconditional.



Figure 14.8: Information flow for optimal routeing; late call forwarding (CFB (NDUB), CFNRc) (message sequence chart)

Figure 14.9 shows the network model for optimal routeing when a bearer has been established before the invocation of late call forwarding. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling and the bearer. The GMSC server replaces the bearer termination towards the visited MSC server (T_{MSC}) with the bearer termination for the forwarded-to subscriber (T_C) in an existing context in the MGW. The bearer termination T_A is used for the bearer towards the preceding MGW (calling subscriber).



Figure 14.9: Optimal routeing; late call forwarding (CFB (UDUB), CFNRy, CD) (Network model)

Figure 14.10 shows the message sequence example for the optimal routeing for late call forwarding with a forward bearer release. In the example the MSC server requests the MGW to remove the termination towards the served mobile subscriber after the bearer towards the RNC has been released. At reception of the release message from the GMSC server requests the MGW to be prepared for the bearer release. When the GMSC server receives the Release Complete it requests the MGW to release the bearer.



CD Call Deflection

Figure 14.10: Information flow for optimal call routeing; late call forwarding (CFB (UDUB), CFNRy, CD), forward bearer release (message sequence chart)

Figure 14.11 shows the message sequence example for the optimal routeing for late call forwarding with a backward bearer release. In the example the MSC server requests the MGW to remove the termination towards the served mobile subscriber after the bearer towards the RNC has been released. At reception of the release message from the GMSC server the MSC server requests the MGW to be release the bearer. When the GMSC server receives the Release Complete it requests the MGW to remove the bearer termination.



CD Call Deflection

Figure 14.11: Optimal call routeing; late call forwarding (CFB (UDUB), CFNRy, CD), backward bearer release (message sequence chart)

14.6 Providing tones or announcements_

It shall be assumed that the MGW selected for the call has the capabilities to provide announcements and tones.

14.6.1 Preconditions when providing in-band information to the calling subscriber

For a mobile terminating/forwarded call, announcements/tones may be provided to the calling subscriber only when both of the following conditions are satisfied:

- 1. Either:
 - a. The incoming IAM indicated that the Continuity message will follow, and a Continuity message has been received, or
 - b. The incoming IAM did not indicate that the Continuity message will follow;
- 2. Notification indicating successful completion of the incoming side bearer set-up has been received from the MGW using the Bearer Established procedure.

For a mobile originating call, the traffic channel assignment shall be completed before providing the in-band information to the calling subscriber.

14.6.2 Preconditions when providing in-band information to the called subscriber

The called party is selected by the calling party, or a supplementary service (call forwarding, call deflection, CAMEL redirection etc), or a call is initiated by the gsmSCF using the Initiate Call Attempt procedure. The called party may also be in the PSTN.

Announcements/tones may be provided to the called subscriber only when both of the following conditions are satisfied:

- 1. The called party has answered and is still active in the call.
- 2. Notification indicating successful completion of the outgoing side bearer set-up has been received from the MGW using the Bearer Established procedure.

14.6.3 Preconditions when providing in-band information to multiple subscribers

The gsmSCF may instruct the MSC-S/gsmSSF to provide announcements or tones for multiple subscribers. For each calling and called subscriber the precondition for calling and called subscriber (respectively) shall be fulfilled. If the preconditions are not fulfilled for all subscribers (e.g. one of the called parties is in the alerting phase), then the announcements/tones shall not be played to the subscribers who do not meet the preconditions, but the announcements/tones shall be played to the subscribers (if any) who meet the preconditions.

14.6.4 Request to play an announcement/tone

The (G)MSC server/gsmSSF/gsmSRF provides the MGW with the announcement/tone identification and optionally requests the MGW to notify the announcement/tone completion using the Play Announcement or Send Tone procedure (bullet 1 in figure 14.13).

14.6.5 Stopping an announcement/tone

The (G)MSC server/gsmSSF/gsmSRF can order the MGW to stop the current announcement/tone using the Stop Announcement or Stop Tone procedure (bullet 2 in figure 14.13).

14.6.6 Announcement/tone completed

If notification of the announcement/tone completion was requested in the Play Announcement or Send Tone procedure, the MGW notifies the (G)MSC server/gsmSSF/gsmSRF when the announcement/tone has been completed using the Announcement Completed or Tone Completed procedure (bullet 3 in figure 14.13).

14.6.7 Example 1

Figure 14.12 shows the network model for providing in-band information to the calling subscriber where the (G)MSC-S requests to play the announcement/tone directly on the desired termination from which the signal shall be sent. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling and the bearer. The bearer termination Tx is used for the bearer towards the preceding MGW (calling subscriber).

A bearer termination Ty may exist in the same context towards the succeeding MGW (called subscriber) and may be both-way connected and through-connected with Tx. In that case, the exchange of media streams between the terminations Tx and Ty is assumed to be normally interrupted during playing the announcement in the direction towards which the announcement is played.



Figure 14.12: Providing in-band information (Network model)

Figure 14.13 shows the message sequence example for providing the calling party with an announcement/tone. In the example the (G)MSC server requests the MGW to play an announcement/tone and to notify the announcement/tone completion. The (G)MSC server may stop the announcement while the current announcement/tone is ongoing.



NOTE: OR1: Stop the announcement/tone (Y: yes N:no) OR2: Notification of completion required (Y: yes N:no)

Figure 14.13: Playing an announcement/tone (message sequence chart)

14.6.8 Example 2

Figure 14.6.8/1 shows the network model for providing in-band information to the calling subscriber where the (G)MSC-S requests to play the announcement/tone on another termination in the context connected via topology information to the desired termination(s). E.g. during the call setup, the MSC-S requests the MGW to send a ring back tone to the calling subscriber when receiving the indication that the called subscriber is being alerted.

The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling and the bearer. The in-band information applied on the T0 termination is forwarded to the bearer termination(s) towards which the topology association is either 'oneway' or 'bothway', i.e. towards the Ta termination in this example.





Figure 14.6.8/2 shows the message sequence example for providing the calling party with the announcement/tone. In the example the MSC requests the MGW to play an announcement/tone by seizing a new ephemeral

announcement/tone termination with the indication that a tone or announcement shall be forwarded internally to the bearer termination(s) towards which the topology association is either 'oneway' or 'bothway'.



NOTE OR1: Notification of completion required (Y: yes, N::no)

OR2: Stop the announcement/tone (Y: yes, N::no)

Figure 14.6.8/2: Playing an announcement/tone (message sequence chart)

14.7 Global Text Telephony

3GPP TS 23.226 [26] describes the high level architecture and functionality of GTT. When text based conversation is needed by a subscriber, the call is established with general call control functions like any other call. Within the call control transactions the UE indicates the need for text conversation (see 3GPP TS 24.008 [4]), which then requires actions in the core network where the pooling mechanism is chosen for GTT feature. This section describes only the option where the CTM pool is provided in the Media Gateway in the Core Network.

MSC Server receives an indication by the UE about the need of text conversation, allocates terminations in MGW with CTM (Cellular Text telephony Modem) capabilities for the detection of CTM signals from radio access network. The default action of the call path in the CTM-detection/conversion function in MGW is to transfer audio transparently while monitoring for text telephone signals. When valid text telephone signals are detected, the converting action of the channel takes effect. The path converts between the detected CTM at the access termination and PSTN text telephony signalling methods on the network side. This mode of operation continues until text signalling ceases. Then transparent audio transport is re-established, again monitoring for text signals.

The CTM channel is created with Prepare bearer for Iu/ATM terminations, Prepare IP Transport for Iu/IP terminations, Reserve circuit procedure for A interface over TDM terminations or with Reserve RTP Connection Point

for A interface over IP by including Cellular Text Telephone package properties. The core network bearer shall be established with default PCM codec to carry the T.140 protocol.

The outcome of CTM negotiation towards the CTM user in the user plane is reported in the CTM report procedure.

At release the MGW may report the number of bits of Global Text Telephony data in accordance with ITU recommendation T.140 sent for the call.

14.8 Emergency Calls

Emergency Calls shall be handled as in clause 6.1 Basic Mobile Originating Call. and clause 6.2 Basic Mobile Terminating Call. The Procedure Emergency Call Indication may be used for informing the MGW about the emergency call.

14.9 Subscriber and equipment trace

For the subscriber and equipment trace the MSC server may be activated for tracing (see 3GPP 32.421 [34]). Besides performing trace record generation, if Signalling based activation the MSC server shall forward the Trace Session activation to GERAN with BSSAP signaling, UTRAN with RANAP signaling (see 3GPP TS 25.413) and if the trace control and configuration parameters are defined so, the MGW. For detailed description of trace session activation mechanism see 3GPP TS 32.422 [35]. The activation to the MGW is sent using a trace package which is included either to the Add or Modify command(s) in mobile originated call or mobile terminating call. In the case of handovers where new termination is created the trace package is also included into the Add command. If the MSC does not receive the optional trace interface list IE in the trace activation request it shall request tracing on all terminations that support the total MGW trace interface that the termination supports and optionally the Mc interface trace value if requested from the HLR. The content of the trace records generated in the MSC server and the MGW will follow the rules of 3GPP TS 32.422[35], 3GPP TS 32.423 [36]. The content of the MGW and MSC Server trace record is described in TS 32.423 [36].

14.10 Customized Alerting Tone

14.10.1 General

Customised Alerting Tone may be provided from a "CAT Server" switched into the call at appropriate trigger points. The "CAT Server" is out of the scope of the 3GPP architecture and therefore the signalling procedures towards the CAT Server (if not collocated) are not described.

The Forward CAT indicators and Backward CAT indicators information elements are specified in 3GPP TS 29.205 [22].

The following sub-clauses describe extensions to basic call handling and unless explicit exceptions are made all basic call protocol handling support is unchanged.

14.10.2 Audio CAT

14.10.2.1 Audio CAT Activated by the Calling Party

If the calling party subscriber has activated the CAT service (CAT-A) for audio and the subscriber is in its Home PLMN and the calling party subscriber gives priority to CAT-A, the originating MSC Server supporting the audio CAT-A service shall connect that calling party UE towards the CAT Server during alerting phase. The MSC Server is responsible for switching the user plane towards the CAT Server during alerting phase and then on receipt of answer message from the called party the MSC Server shall switch the user plane towards the called party and releasing the connection to the CAT Server.

The originating MSC may indicate the CAT priority to the terminating network by setting the CAT Priority indicator of the Forward CAT indicators information element when sending an IAM to the succeeding node.

If the calling party subscriber supports CAT-A but gives priority to CAT-B and the originating MSC receives an ACM or CPG from the succeeding network with the CAT content Indicator in the Backward CAT indicators information element set to "inband media content available" then it shall not switch the user plane towards the CAT Server. If however the originating MSC does not receive an indication that CAT-B is present then it shall switch the user plane towards the CAT server as described above for the CAT-A priority case.

NOTE: Providing alerting tone at the calling party rather than from the called party network can cause voice clipping if the called party answers and speaks while the calling party network has not disconnected the inband alerting tone.

14.10.2.2 Audio CAT Activated by the Called Party

If the called party subscriber has activated the CAT service (CAT-B) for audio, the GMSC Server supporting the audio CAT-B service shall check the CAT Priority indicator if the Forward CAT indicators information element is received. If CAT-A priority is requested the GMSC Server shall handle the call in accordance with normal call handling not supporting CAT-B service. If no CAT-A priority is received the GMSC Server shall connect the calling party network to the CAT Server during alerting phase. The GMSC Server should indicate in ACM or CPG message if CAT-B is inserted by setting the CAT content Indicator in the Backward CAT indicators information element. The GMSC Server is responsible for switching the user plane towards the CAT Server during alerting phase and then on receipt of answer message from the called party the GMSC Server shall switch the user plane towards the called party and releasing the connection to the CAT Server.

14.10.3 Multimedia CAT

14.10.3.1 Introduction

The following subclauses specify the mobile originated and mobile terminated multimedia call procedures to generate a multimedia CAT to a UE supporting the multimedia CAT capability.

A UE supporting multimedia CAT will indicate support of this capability to the MSC server as specified in 3GPP TS 24.008 [4].

To play a multimedia CAT to such a UE, upon receiving the indication that the called party is being alerted and that a multimedia CAT is available, the MSC Server supporting multimedia CAT shall request the MGW to both-way through-connect the bearer towards the source of the multimedia CAT using the Change Through-Connection procedure, and then request the calling UE to attach to the user connection for multimedia as soon as an appropriate channel in multimedia mode is available and to set up an H.324 call by indicating "inband multimedia CAT available" within an ALERTING message or a PROGRESS message as specified in 3GPP TS 24.008 [4] subclause 5.3.6.4.

14.10.3.2 Multimedia CAT Activated by the Calling Party

If the calling party subscriber has activated CAT service (CAT-A) for multimedia and the subscriber is in its Home PLMN and the calling party subscriber gives priority to CAT-A, the originating MSC Server supporting multimedia CAT-A service for multimedia it shall connect that calling UE towards the CAT Server during alerting phase. The MSC Server is responsible for switching the user plane towards the CAT Server during alerting phase and then on receipt of answer message from the called party the MSC Server shall switch the user plane towards the called party and release the connection to the CAT Server.

The originating MSC may indicate CAT-A has priority to the terminating network by setting the CAT Priority indicator of the Forward CAT indicators information element to 'Priority given to Calling party (CAT-A)' when sending an IAM to the succeeding node.

NOTE: As the Forward CAT indicators information element is required to be sent by the originating MSC with the Multimedia CAT capability Indicator set to to "MCAT supported" if it permits multimedia CAT-B, the inclusion of the Forward CAT indicators information element is not strictly required since if no CAT parameters are included it has the same effect as indicate CAT-A has priority however it does provide explicit indication that the calling party supports CAT-A.

If CAT-B is given priority then the Multimedia CAT capability Indicator of the Forward CAT indicators information element shall be set as described in subclause 14.10.3.3.1.

If the calling party subscriber supports CAT-A but gives priority to CAT-B and the originating MSC receives an ACM or CPG from the succeeding network with the CAT content Indicator in the Backward CAT indicators information element set to "inband media content available" then it shall not switch the user plane towards the CAT Server and shall follow the procedures defined in subclause 14.10.3.3.1. If however the originating MSC does not receive an indication that CAT-B is present then it shall switch the user plane towards the CAT server as described above for the CAT-A priority case.

14.10.3.3 Multimedia CAT Activated by the Called Party

14.10.3.3.1 Mobile originated multimedia call

If the calling UE supports the multimedia CAT service and no CAT-A priority exists, the originating MSC supporting the multimedia CAT service shall inform the succeeding node that multimedia CAT is supported by setting the Multimedia CAT capability Indicator of the Forward CAT indicators information element to "MCAT supported" in the IAM message sent to the succeeding node.

The MSC Server supporting the multimedia CAT service shall consider that a multimedia CAT is available if it receives an ACM or CPG message including the Multimedia CAT content Indicator in the Backward CAT indicators information element set to "inband media content available".

14.10.3.3.2 Mobile terminated multimedia call

If the called party subscriber supports the "CAT-B" service for multimedia and if the GMSC Server supporting the optional CAT-B service for multimedia receives the Multimedia CAT capability Indicator in the Forward CAT indicators information element set to "MCAT supported" in the IAM from the preceding node, the GMSC Server shall connect the calling party to the CAT Server during the alerting phase.

To play a multimedia CAT to the calling subscriber, upon receiving the indication that the called party is being alerted, the GMSC supporting multimedia CAT shall request the MGW to establish, and both-way through-connect, a bearer connection towards a CAT Server.

The GMSC shall inform the originating MSC that a multimedia CAT is available (if any) by setting the Multimedia CAT content Indicator of the Backward CAT indicators information element to "inband media content available" in the ACM or CPG message.

14.10.3.3.3 Example

Figure 14.10.3.3.3.1 shows the message sequence chart example for providing multimedia CAT activated by the called party during a basic multimedia call establishment. In the example, the calling UE, the originating MSC and GMSC support the multimedia CAT capability.

The calling UE initiates a multimedia call establishment during which it indicates that it supports multimedia CAT capability (bullet 1). The originating MSC sends an Initial Address message to the GMSC of the called party indicating that Multimedia CAT is supported.

The GMSC server determines that the called party has a multimedia CAT subscription (bullet 6) e.g. based on OSS code retrieved from the HLR.

Upon receiving the indication that the called party is being alerted, the GMSC sets up a UDI call towards the CAT Server. Upon reception of an Address Complete message from the CAT server, it returns to the originating MSC an Address Complete or Call Progress message indicating that multimedia CAT is available (bullet 13). This leads the originating MSC to request the calling UE (supporting multimedia CAT) to attach to the user connection and to set up a H.324 call to the CAT server (bullet 14).

When the called party accepts the call, the GMSC releases the call towards the CAT server (bullet 18), waits for the Release Complete message from the CAT Server, and then through-connects the calling and called legs and returns an answer indication to the originating MSC.

Upon reception of the CONNECT message, the calling UE releases any on-going H.324 call and establishes a new H.324 call towards the called party.

3GPP TS 23.205 version 9.4.0 Release 9





15 Tunnelling

NOTE: All message sequence charts in this clause are examples. All valid call establishment message sequences can be derived from the example message sequences and associated message pre-conditions.

15.1 Forward Bearer Establishment

The following clauses describe the requirements for tunnelling transport mechanism within the bearer independent CS core network. These requirements are supplementary to those already stated in the Call Establishment clause. If out of-band transcoder control is applied for a speech call, it shall be performed in accordance with 3GPP TS 23.153 [3].

15.1.1 Outgoing Side

15.1.1.1 Tunnel Selection

If the MGW selection occurs before the IAM is sent, the (G)MSC server uses the Prepare Bearer procedure to indicate a tunnel support to the MGW. Depending upon the received value, the MGW shall determine whether tunnelling shall actually be used and when to send the tunnel data (bullet 1 in figure 15.2).

If the (G)MSC server indicated that tunnelling is not supported, the bearer will be established as described in clause Call Establishment.

If the (G)MSC server indicated that fast tunnelling is supported, the MGW may select which tunnelling method to use. In this case the MGW shall select either fast tunnelling or the non-tunnelling method.

If the (G)MSC server indicated that delayed tunnelling is supported, the MGW may select which tunnelling method to use. In this case the MGW shall select either delayed tunnelling or the non-tunnelling method.

If the MGW is allowed to choose whether tunnelling is to be used, it shall select either fast, delayed, or the non-tunnelling method.

The MGW shall respond to the Prepare Bearer procedure with the used tunnel indication, when the type of tunnelling mechanism has been decided.

NOTE: For a given bearer type, other specifications may describe the mechanism to be used to transport bearer control information. An MGW is only required to comply with that specification.

15.1.1.2 Initial addressing

If the MGW selection has occurred, the MGW shall respond to the Prepare Bearer procedure indicating whether tunnelling is allowed and what type of tunnelling is used – fast or delayed forward. The (G)MSC server provides a tunnel indicator to the succeeding node in the IAM to indicate that tunnelling is to be used. For fast tunnelling, the (G)MSC server waits for the MGW to use the Tunnel Information Up procedure to provide the tunnel data before the IAM is sent.

If the MGW indicates that tunnelling is not to be used, then tunnel indicator is not included in the Initial Address message and the bearer will be established as described in clause Call Establishment.

If the MGW has not been selected yet, then the (G)MSC server decides whether delayed tunnelling is supported or not. If the delayed tunnelling is supported the tunnel indicator is included to the Initial Address message to indicate that. Otherwise the tunnel indicator is not included to the Initial Address message and the bearer will be established as described in clause Call Establishment.

15.1.1.3 Fast forward tunnelling

The tunnel data is transferred in the IAM and the subsequent Tunnel Information message(s).

Before the IAM is sent, the (G)MSC server waits for the MGW to use the Tunnel Information Up procedure to supply the tunnel data. The (G)MSC server sends the received tunnel data to the succeeding node in the IAM (bullet 2 in example, clause 15.2).

When the (G)MSC server receives a Tunnel Information message from the succeeding node the (G)MSC server uses the Tunnel Information Down procedure to supply the MGW with the received tunnel data (bullet 5 in example sequence 15.2).

15.1.1.4 Delayed forward tunnelling

The tunnel data is transferred in the Tunnel Information messages following the Bearer Information message.

If tunnel indicator was included in the IAM indicating that delayed tunnelling is supported, the succeeding node may include the tunnel indicator to the Bearer Information message. If the tunnel indicator is received the (G)MSC server indicates the delayed tunnel support in the Establish Bearer procedure.

When the MGW sends the Tunnel Information Up procedure, the (G)MSC server sends the received tunnel data in the Tunnel Information message to the succeeding node.

When the (G)MSC server receives a Tunnel Information message from the succeeding node, the (G)MSC server uses the Tunnel Information Down procedure to send the received tunnel data to the MGW.

15.1.1.5 Bearer control signalling transfer

The tunnelling of the bearer control signalling is transported transparently through the (G)MSC server during the call establishment and at any other time until Release is sent or received.

15.1.2 Incoming Side

15.1.2.1 Initial addressing

The (G)MSC server receives the possible tunnel indicator and the tunnel data in IAM. Based on received information it provides the tunnel support indication and the tunnel data to the MGW.

15.1.2.2 Tunnel Selection

If the tunnel indicator was received in the IAM, the (G)MSC server uses the received tunnel indicator to indicate the support of tunnel to the MGW. If the tunnel indicator is received in the IAM without tunnel data, the (G)MSC server checks the value of the tunnel indicator. If the tunnel indicator indicates that tunnel mechanism is to be used then delayed tunnelling is indicated to the MGW. If the tunnel indicator indicates that tunnel mechanism is supported the (G)MSC server decides whether the delayed tunnel is supported or non tunnelling mechanism is used. If both tunnel indicator and tunnel data are received in the IAM, fast tunnelling is indicated to the MGW.

If no tunnel indicator was received in the IAM, then the preceding node has indicated that non-tunnelling mechanism is to be used.

The (G)MSC server uses the Prepare Bearer procedure to supply the tunnel support indication to the MGW.

The MGW decides based on the received tunnel support indication from the (G)MSC server whether to use delayed tunnelling or not. In the response the MGW provides the used tunnel indication to the (G)MSC server.

15.1.2.3 Fast forward tunnelling

The tunnel data is transferred in the IAM and the subsequent Tunnel Information message(s).

The (G)MSC server sends the tunnel data received in the IAM to the MGW using the Tunnel Information Down procedure (bullet 3 in example, clause 15.2).

When the MGW sends the Tunnel Information Up procedure, the (G)MSC server sends the received tunnel data in the Tunnel Information message to the preceding node (bullet 4 in example, clause 15.2).

15.1.2.4 Delayed forward tunnelling

The tunnel data is transferred in the Tunnel Information messages following the Bearer Information message.

If tunnel indicator was received in the IAM indicating that delayed tunnel is supported and delayed tunnelling was indicated by the MGW, the (G)MSC server shall include the tunnel indicator to the Bearer Information message which is sent to the preceding node.

When the (G)MSC server receives a Tunnel Information message from the preceding node, the (G)MSC server uses the Tunnel Information Down procedure to send the received tunnel data to the MGW.

When the MGW sends the Tunnel Information Up procedure, the (G)MSC server sends the received tunnel data in the Tunnel Information message to the preceding node.

15.1.2.5 Bearer control signalling transfer

The tunnelling of bearer control signalling is transported transparently during the call establishment and at any other time until Release is sent or received.

15.1.2.6 Example

Figure 15.1 shows the network model for the forward tunnelling transport mechanism. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling. The (G)MSCa seizes one context with two bearer terminations in MGWa. The bearer termination T1 is used for the bearer towards the incoming side of (G)MSCa and the bearer termination T2 is used for the tunnelling towards the succeeding MGW. The (G)MSCb seizes one context with two bearer terminations in MGWb. The bearer termination T3 is used for the bearer towards the outgoing side of (G)MSCb and the bearer termination T4 is used for the tunnelling towards the preceding MGW.



Figure 15.1: Forward Tunnelling Transport Mechanism (network model)

Figure 15.2 shows the message sequence example for fast forward tunnelling transport mechanism. In the example (G)MSCa indicates to MGWa that fast tunnelling is requested. After MGWa has notified the (G)MSCa of the tunnel data, the IAM is sent to the (G)MSCb. The (G)MSCb indicates to MGWb that fast tunnelling is supported and sends the received tunnel data to MGWb. Once MGWb has sent the tunnel data to the (G)MSCb, the (G)MSCb sends a Tunnel Information message with the tunnel data to the (G)MSCa. The (G)MSCa sends the received tunnel data to MGWa. The handling of Continuity message, through-connection and answer is as normal for non-tunnelled forward bearer establishment.



Figure 15.2: Fast Forward Tunnelling Transport Mechanism (message sequence chart)

15.2 Backward Bearer Establishment

The following clauses describe the additional requirements for tunnelling transport mechanism within the bearer independent CS core network. If out-of-band transcoder control is applied for a speech call, it shall be performed in accordance with 3GPP TS 23.153[3].

15.2.1 Outgoing Side

15.2.1.1 Tunnel Selection

The (G)MSC server uses the Prepare Bearer procedure to indicate a tunnel support to the MGW. Depending upon the received value, the MGW shall determine whether tunnelling shall actually be used and when to send the tunnel data (bullet 1 in example, clause 15.4).

If the (G)MSC server indicated that tunnelling will be not supported, the bearer is established as described in clause Call Establishment.

If the (G)MSC server indicated that fast tunnelling is supported, the MGW may select which tunnelling method it can use. In this case the (G)MSC may select either fast tunnelling or the non-tunnelling method.

If the (G)MSC server indicated that delayed tunnelling is supported, the MGW may select which tunnelling method it can use. In this case the (G)MSC server may select either delayed tunnelling the non-tunnelling method.

If the MGW is allowed to choose whether tunnelling is to be used, it shall select either fast, delayed or the non-tunnelling method.

After MGW has decided which tunnelling mechanism to use , it responds to the Prepare Bearer procedure with the used tunnel indication.

15.2.1.2 Initial addressing

The MGW shall respond to the Prepare Bearer procedure to indicate whether tunnelling is allowed and what type of tunnelling is used – fast or delayed forward. The (G)MSC server provides a tunnel indicator to the succeeding node in the IAM. For fast tunnelling, the (G)MSC server waits for the MGW to use the Tunnel Information Up procedure to provide the tunnel data before the IAM is sent.

If the MGW indicates that tunnelling is not to be used, the bearer will be established as described in clause Call Establishment.

15.2.1.3 Fast forward tunnelling

The tunnel data is transferred in the IAM and the subsequent Tunnel Information message(s).

Before the IAM is sent, the (G)MSC server waits for the MGW to use the Tunnel Information Up procedure to supply the tunnel data. The (G)MSC server sends the received tunnel data to the succeeding node in the IAM.

When the (G)MSC server receives a Tunnel Information message from the succeeding node the (G)MSC server uses the Tunnel Information Down procedure to supply the MGW with the received tunnel data.

15.2.1.4 Delayed backward tunnelling

The tunnel data is transferred in the Tunnel Information messages following the IAM.

When the (G)MSC server receives a Tunnel Information message from the succeeding node the (G)MSC server uses the Tunnel Information Down procedure to supply the MGW with the received tunnel data (bullet 4 in example, clause 15.4).

When the MGW sends the Tunnel Information Up procedure, the (G)MSC server sends the received tunnel data in the Tunnel Information message to the succeeding node (bullet 5 in example, clause 15.4).

15.2.1.5 Bearer control signalling transfer

The tunnelling of bearer control signalling is transported transparently through the (G)MSC server during the call establishment and at any other time until Release is sent or received.

15.2.2 Incoming Side

15.2.2.1 Initial addressing

The (G)MSC server receives the possible tunnel indicator and the tunnel data in the IAM. Based on received information it provides the tunnel support indication and the tunnel data to the MGW.

15.2.2.2 Tunnel Selection

If the tunnel indicator was received in the IAM, the (G)MSC server uses the received tunnel indicator to indicate the support of tunnel to the MGW. If the tunnel indicator is received in the IAM without tunnel data, delayed tunnelling is indicated to the MGW. If tunnel indicator and tunnel data are received in the IAM, fast tunnelling is indicated to the MGW.

The (G)MSC server uses the Establish Bearer procedure to supply the tunnel support indication to the MGW (bullet 2 in example, clause 15.4).

15.2.2.3 Fast forward tunnelling

The tunnel data is transferred in the IAM and the subsequent Tunnel Information message(s).

The (G)MSC server sends the tunnel data received in the IAM to the MGW using the Tunnel Information Down procedure.

When the MGW sends the Tunnel Information Up procedure, the (G)MSC server sends the received tunnel data in the Tunnel Information message to the preceding node.

15.2.2.4 Delayed backward tunnelling

When the MGW sends the Tunnel Information Up procedure, the (G)MSC server sends the received tunnel data in the Tunnel Information message to the preceding node (bullet 3 in example, clause 15.4).

When the MGW receives an Nb User Plane Initialisation message (bullet 5a in example sequence 15.4). before the subsequent Tunnel Information Down procedure (bullet 6 in example sequence 15.4), t he MGW may acknowledge this Nb User Plane Initialisation message without waiting for the Tunnel Information Down procedure, and send the acknowledge message towards the IP address and port that were supplied as source within the IP packet transporting the Nb User Plane Initialisation message (bullet 5b in example sequence 15.4). The MGW shall use the same RTP Payload-Type number for the acknowledge message, which was used in the RTP header of the packet transporting the Nb User Plane Initialisation message.

Alternatively, when the MGW receives an Nb User Plane Initialisation message (bullet 5a in example sequence 15.4). before the subsequent Tunnel Information Down procedure (bullet 6 in example sequence 15.4), the MGW may wait for the Tunnel Information Down procedure, before sending the Nb User Plane Initialisation acknowledge message (bullet 6a in example sequence 15.4). While waiting for the Tunnel Information Down procedure, the MGW may receive repetition(s) of the Nb User Plane Initialisation message and shall not treat this as error case.

When the (G)MSC server receives a Tunnel Information message from the preceding node, the (G)MSC server uses the Tunnel Information Down procedure to send the received tunnel data to the MGW (bullet 6 in example, clause 15.4).

After receiving the Tunnel Information Down procedure (bullet 6 in example sequence 15.4) and acknowledging the NbFP Init message (bullet 5b or 6a in example sequence 15.4), the MGW shall notify the MSC server about the bearer establishment (bullet 7 in example sequence 15.4).

15.2.2.5 Bearer control signalling transfer

The tunnelling of bearer control signalling is transported transparently through the (G)MSC server during the call establishment and at any other time until Release is sent or received.

15.2.2.6 Example

Figure 15.3 shows the network model for the backward delayed tunnelling transport mechanism. The "squared" line represents the call control signalling. The "dotted" line represents the bearer control signalling. The (G)MSCa seizes one context with two bearer terminations in MGWa. The bearer termination T1 is used for the bearer towards the incoming side of (G)MSCa and the bearer terminations T2 is used for the tunnelling towards the succeeding MGW. The (G)MSCb seizes one context with two bearer terminations in MGWb. The bearer termination T3 is used for the bearer towards the outgoing side of (G)MSCb and the bearer termination T4 is used for the tunnelling towards the preceding MGW.



Figure 15.3: Delayed Backward Tunnelling Transport Mechanism (network model)

Figure 15.4 shows the message sequence example for backward tunnelling transport mechanism. In the example the (G)MSCa indicates to MGWa that delayed tunnelling is requested. After MGWa has responded the (G)MSCa of tunnelling, the IAM is sent to (G)MSCb. The (G)MSCb indicates to MGWb that delayed tunnelling is supported. Once MGWb has sent the tunnel data to the (G)MSCb, the (G)MSCb sends the received tunnel data in the Tunnel Information message to the (G)MSCa. The (G)MSCa sends the received tunnel data to MGWa. Once MGWa has sent the tunnel data to the (G)MSCa, the (G)MSCa sends the received tunnel data in the Tunnel Information message to the (G)MSCa sends the received tunnel data in the Tunnel Information message to the (G)MSCa, the received tunnel data to MGWb. The handling of Continuity message, through-connection and answer is as normal for non-tunnelled backward bearer establishment.



Figure 15.4: Delayed Backward Tunnelling Transport Mechanism (message sequence chart)

16 Messages/Procedures and their contents

This clause contains the detailed description of the information flows used in bearer independent CS core network.

Each Information Element, IE, is marked as (M) Mandatory, (C) Conditional or (O) Optional. A mandatory information element shall always be present. A conditional information shall be present if certain conditions are fulfilled; if those conditions are not fulfilled it shall be absent. An optional information element may be present or absent, at the discretion of the application at the sending entity. This categorisation is a functional classification, i.e., stage 2 information and not a stage 3 classification to be used for the protocol.

The stage 2 and stage 3 message and information element names are not necessarily identical.

16.1 Messages between (G)MSC servers

Table 16.1 indicates messages between (G)MSC servers in Nc interface. Only the new messages and information elements required by the bearer independent CS core network are shown.

Mossago	Mossago	Information cloment	Information	Information alamont description
message	direction	name	element required	mormation element description
Initial Address	Forward	Bearer Establishment Direction	М	This information element indicates that the direction of bearer establishment.
		Bearer Address	С	This information element indicates the bearer
				address of the MGW used by the preceding node. This information element is included when an AAL2 bearer is establish ed using
				backward bearer establishment control protocol. Otherwise the information element
				is optional.
		Binding Reference	С	This information element indicates the bearer identifier in the MGW used by the preceding node. This information element is included
				when an AAL2 bearer is established using backward bearer establishment control protocol. Otherwise the information element
			-	is optional.
		MGW-id	0	This information element indicates the MGW selected by the preceding node.
		Bearer Characteristics	М	This information element indicates the characteristics of the bearer.
		Tunnel Indicator	0	This information element indicates either that tunnelling is to be used or tunnelling is supported.
		Tunnel data	0	This information element contains the tunnel data that is provided between MGWs.
Bearer Information	Backward	Bearer Address	C	This information element indicates the bearer address of the MGW used by the succeeding node. This information element is included when an AAL2 bearer is established using forward bearer establishment control protocol. Otherwise the information element is optional.
		Binding Reference	C	This information element indicates the bearer identifier in the MGW used by the succeeding node. This information element is included when an AAL2 bearer is established using forward bearer establishment control protocol. Otherwise the information element is optional.
		MGW-id	0	This information element indicates the MGW selected by the succeeding node.
		Tunnel Indicator	0	This information element indicates that tunnelling is used.
Tunnel Information	Both	Tunnel Indicator	М	This information element indicates that the message contains tunnelling information.
		Tunnel data	М	This information element contains the tunnel data that is provided between MGWs.
Start DTMF	Both	Start DTMF indicator	М	This information element indicates that the message is used for Start DTMF.
		Digit	М	This information element indicates the digit for DTMF tone generation.
Start DTMF Ack	Both	Start DTMF Ack indicator	М	This information element indicates that the message is used for Start DTMF Ack.
Stop DTMF	Both	Stop DTMF indicator	М	This information element indicates that the message is used for Stop DTMF.
Stop DTMF Ack	Both	Stop DTMF Ack indicator	М	This information element indicates that the message is used for Stop DTMF Ack.

Table 16.1: Messages between (G)MSC servers

16.2 Procedures between (G)MSC server and MGW

The clauses below indicate the procedures used between (G)MSC server and MGW in Mc interface. The procedures are logical, i.e. message identifiers are not part of the protocol. Several logical procedures can be combined into one H.248 command in order to perform required transactions. If several logical procedures are combined, only one context/context request and only one bearer termination/bearer termination request is sent in the H.248 command. Exemption is the Change Flow Direction procedure, where the two bearer terminations are related to a change of the context and not to a command of the bearer termination. All the procedures below describe a successful operation. If the procedure is rejected, a Command Reject is sent back to the entity that sent the command request.

16.2.1 Change Flow Direction

This procedure is used to change the flow direction between bearer terminations within the context.

Procedure	Initiated	Information element name	Information element	Information element description
Change Flow Direction	(G)MSC-S	Context/Context Request	M	This information element indicates the existing context or a new context where the flow direction is changed.
		Bearer Termination 1/ Bearer Termination 1 Request	М	This information element indicates the existing bearer termination or a new bearer termination from where the new flow direction is applied.
		Bearer Termination 2/ Bearer Termination 2 Request	М	This information element indicates the existing bearer termination or a new bearer termination where to the new flow direction is applied.
		Flow Direction	М	This information element indicates the flow direction from the bearer termination 1 to bearer termination 2 within the context.
Change Flow Direction Ack	MGW	Context	М	This information element indicates the context where the command was executed.

Table 16.2: Procedures between (G)MSC server and MGW: Change Flow Direction

NOTE 1: This procedure may be combined with Prepare Bearer, Prepare_IP_Transport, Establish Bearer, Reserve Circuit, Reserve RTP Connection Point, Configure RTP Connection Point or Reserve and Configure RTP Connection Point, Join Bearer Termination or Isolate Bearer Termination procedure. This list of procedures is not exhaustive.

NOTE 2: Only one of the bearer terminations can be a new bearer termination within this procedure.
16.2.2 Join Bearer Termination

This procedure is used to join a bearer termination with other bearer terminations within the context.

Procedure	Initiated	Information element name	Information element required	Information element description
Join Bearer Termination	(G)MSC-S	Context	M	This information element indicates the context where the bearer termination is joined.
		Bearer Termination/Bearer Termination Request	М	This information element indicates the existing bearer termination or requests a new bearer termination to be joined with the other bearer terminations within the context.
Join Bearer Termination Ack	MGW	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	M	This information element indicates the bearer termination where the command was executed.

16.2.3 Isolate Bearer Termination

This procedure is used to isolate one bearer termination from the other bearer terminations within the context.

Procedure	Initiated	Information element name	Information element required	Information element description
Isolate Bearer Termination	(G)MSC-S	Context/Context Request	Μ	This information element indicates the existing context or a new context where to the bearer termination is isolated.
		Bearer Termination/Bearer Termination Request	М	This information element indicates the existing bearer termination or requests a new bearer termination to be isolated from the other bearer terminations within the context.
Isolate Bearer Termination Ack	MGW	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	М	This information element indicates the bearer termination where the command was executed.

|--|

16.2.4 Establish Bearer

This procedure is used to request a bearer establishment.

Table 16.5: Procedures between (G)MSC server and MGW: Establish Bearer

Procedure	Initiated	Information element	Information	Information element description
		name	element required	
Establish Be	earer	Context/Context	M	This information element indicates the
(G)MSC-	S	Request		existing context or requests a new context for
				the bearer termination.
		Bearer	М	This information element indicates the
		Termination/Bearer		existing bearer termination or requests a new
				established.
		Bearer Establishment	М	This information element requests
		Request		establishment of a bearer.
		Destination Binding	С	This information element indicates the bearer
		Reference		identifier in the destination MGW. This
				information element shall be included when
				requesting the establishment of an AAL2
				optional.
		Destination Bearer	C	This information element indicates the bearer
		Address		address of the destination MGW. This
				information element shall be included when
				hearer. Otherwise the information element is
				optional
		Bearer Characteristics	М	This information element indicates the
				characteristics of the bearer connection.
		Bearer Service	С	This information element indicates the bearer
		Characteristics		service requested by the user. This
				Codec information element is provided. It
				may be included for a non-speech / non
				SCUDIF multimedia call (see3GPP TS
				23.172 [38], 3GPP TS 29.007 [37] and 3GPP
				TS 29.232 [6]).
		Notify Bearer Event	0	This information element requests a
				notification of an established bearer, a
				released bearer, a modified bearer or a bearer
		Tunnel Support	0	This information element indicates the
				support of tunnel data transfer and when to
				send tunnel data. In 29.232 this information
				element is defined in the Tunnel Information
				down procedure.
		Circuit Switched Data	С	This information element indicates the
				PLMN bearer capabilities and when
				applicable GSM channel coding. This
				information element shall be included
				according to SGPP 15 $29.00/[3/]$ and $3GPP$ TS 29 232 [6] for a non-speech / non
				SCUDIF multimedia call (see 3GPP TS
				23.172 [38])by the MSC server. or by the
				anchor-MSC in case of inter-MSC handover,
				for an anchor MGW network termination.

		Codec	С	This information element indicates the speech coding format to be used for the bearer. This information element is included for a speech call for a network side bearer termination.
		Framing Protocol	0	This information element indicates the framing protocol to be used for the bearer.
		Call Type Discrimination	0	This information element supports modem signalling for GTT feature. NOTE
		Text Telephone	0	This information element supports interworking with PSTN text telephone. NOTE
		Notify termination heartbeat	С	This information element requests termination heartbeat indications. This information element shall be included when requesting a new bearer termination. Otherwise the information element is optional.
		IP Realm Identifier	0	This information element indicates the IP realm of the IP termination.
		Number of desired listener context terminations	0	This information element indicates the number of desired terminations for a listener context in a VGCS or VBS call. It should be included in the first ADD request which establishes a Listener context.
Establish Bearer Ack	MGW	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	Μ	This information element indicates the bearer termination where the command was executed.
NOTE: Procedure	es for use of	Text Telephony indication	and Call Typ	pe discrimination are not defined.

16.2.5 Prepare Bearer

This procedure is used to prepare for a bearer establishment.

Table 16.6: Procedures between	(G)MSC server	and MGW: Prepare	Bearer
--------------------------------	---------------	------------------	--------

Procedure	Initiated	Information element	Information	Information element description
		name	element	
			required	
Prepare Bea	arer	Context/Context	М	This information element indicates the
(G)MSC-	S	Request		existing context or requests a new context for
				the bearer termination.
		Bearer Termination	М	This information element requests a new
		Request		bearer termination for the bearer to be
				established.
		Binding Reference	С	This information element requests the bearer
		Request	-	identifier in the MGW. This information
				element shall be included when preparing the
				establishment of an AAL2 bearer. Otherwise
				the information element is optional.
		Bearer Address	С	This information element requests the bearer
		Request	Ũ	address of the MGW. This information
		ricquest		element shall be included when preparing the
				establishment of an AAI 2 hearer. Otherwise
				the information element is optional
		Sondor Binding	0	This information element indicates the hearer
		Beforence	0	identifier of the conding MCW
		Reference Sondor Dooror	0	This information element indicates the bearer
		Sender Bearer	0	This information element indicates the bearer
		Address		address of the sending MGW.
		Bearer Characteristics	M	I his information element indicates the
				requested characteristics of the bearer
				connection.
		Bearer Service	С	This information element indicates the bearer
		Characteristics		service requested by the user. This
				information element is not included if the
				Codec information element is provided. It
				may be included for a data call (see
			ics M This information element indicates the requested characteristics of the bearer connection. C This information element indicates the bearer connection. C This information element is not included if the Codec information element is provided. may be included for a data call (see 3GPP TS 29.007 [37] and 3GPP TS 29.[6]). nt O This information element requests a notification of an established bearer, a released bearer, a modified bearer or a	3GPP TS 29.007 [37] and 3GPP TS 29.232
				[6]).
		Notify Bearer Event	0	This information element requests a
				notification of an established bearer, a
				released bearer, a modified bearer or a
				bearer modification failure.
		Notify Bearer	0	This information element requests a
		Modification		notification that bearer modification of the
				established bearer is allowed. This
				information element may be included for
				access bearer assignment.
			0	This information element indicates the
				support of tunnel data transfer and when to
				send tunnel data. In 3GPP TS 29.232 [6] this
				information element is defined in the Tunnel
				Information down procedure.
		Circuit Switched Data	С	This information element indicates the PLMN
				bearer capabilities and when applicable GSM
				channel coding and user bit rate. This
				information element shall be included
				according to 3GPP TS 29.007 [37] and 3GPP
				TS 29.232 [6] for a non-speech call by the
				MSC server, or by the anchor-MSC in case of
				inter-MSC handover, for a radio access
				network termination and for an anchor MGW
				network termination.

		Codec	С	This information element indicates the speech or multimedia coding format to be used for the bearer. This information element is included for a speech call and may be included for a SCUDIF multimedia call (see3GPP TS 23.172 [38], 3GPP TS 29.007 [371 and 3GPP TS 29.232 [6]).
		Framing Protocol	0	This information element indicates the framing protocol to be used for the bearer.
		Cellular Text telephony modem	С	This information element indicates the need of CTM function.
		Notify termination heartbeat	М	This information element requests termination heartbeat indications.
		Number of needed conference terminations	0	This information element indicates the number of conference ports needed for that VGCS or VBS call. It shall be included in the first ADD request which establishes the Multiparty context.
		IP Realm Identifier	0	This information element indicates the IP realm of the IP termination.
Prepare Bearer Ack	MGW	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	М	This information element indicates the bearer termination where the command was executed.
		Binding Reference	С	This information element indicates the bearer identifier in the MGW. This information element is included if requested in the Prepare Bearer request.
		Bearer Address	C	This information element indicates the bearer address of the MGW. This information element is included if requested in the Prepare Bearer request.

16.2.6 Reserve Circuit

This procedure is used to select a TDM circuit in the MGW.

Table 16.7: Procedures between (G)MSC server and MGW: Reserve (Circuit
---	---------

Procedure	Initiated	Information element name	Information element	Information element description
			required	
Reserve Ci	rcuit	Context/Context	M	This information element indicates the
(G)MSC-	S	Request		existing context or requests a new context for
				the bearer termination.
		Bearer Termination	M	This information element indicates the
				physical bearer termination for the TDM
				circuit.
		Circuit Switched Data	С	This information element indicates the PLMN
				bearer capabilities and GSM channel coding.
				This information element shall be included
				according to 3GPP TS 29.007 [37] and 3GPP
				15 29.232 [6] for a non-speech call by the
				MSC server, or by the anchor-MSC in case of
				Inter-MSC handover, for a fadio access
		Pooror Sonvice	<u> </u>	The information element indicates the bearer
		Characteristics	C	service requested by the user. This
		Characteristics		information element is included if no Circuit
				Switched Data information element is
				provided.
		Cellular Text telephony	С	This information element indicates the need
		modem	_	of CTM function.
		Notify Released	0	This information element requests a
		Bearer		notification of a released bearer
		Notify termination	0	This information element requests
		heartbeat		termination heartbeat indications.
		Number of desired	0	This information element indicates the
		listener context		number of desired terminations for a listener
		terminations		context in a VGCS or VBS call. It should be
				included in the first ADD request which
Deserve Oire it		Orantaurt		establishes a Listener context.
Reserve Circuit	MGW	Context	IVI	I his information element indicates the
АСК			N4	context where the command was executed.
		Dearer Termination	IVI	tormination where the command was
				evoluted
				executed

16.2.7 Change Through-Connection

This procedure is used to change the through-connection in the bearer termination.

Procedure	Initiated	Information element name	Information element	Information element description
			required	
Change Through-	(G)MSC-S	Context/Context	M	This information element indicates the
Connection		Request		existing context or requests a new context for
				the bearer termination.
		Bearer	М	This information element indicates the
		Termination/Bearer		existing bearer termination or requests a new
		Termination Request		bearer termination where the through-
				connection is changed.
		Through-Connection	М	This information element indicates the
		-		through-connection of the bearer termination.
Change Through-	MGW	Context	М	This information element indicates the
Connection Ack				context where the command was executed.
		Bearer Termination	М	This information element indicates the bearer
				termination where the command was
				executed.

Table 16.8: Procedures between (G)MSC server and MGW: Change Through-Connection

NOTE: This procedure may be combined with Prepare Bearer, Prepare_IP_Transport, Establish Bearer, Reserve Circuit, Reserve RTP Connection Point, Configure RTP Connection Point or Reserve and Configure RTP Connection Point, Join Bearer Termination, Isolate Bearer Termination or Release Bearer procedure. This list of procedures is not exhaustive.

16.2.8 Activate Interworking Function

This procedure is used to activate the interworking function.

Table 16.9: Procedures between (G)MSC server and MGW: Activate Interworking Function

Procedure	Initiated	Information element name	Information element required	Information element description
Activate Interworking	(G)MSC-S	Context	М	This information element indicates the context for the bearer termination.
Function		Bearer Termination	М	This information element indicates the bearer termination where the interworking function is activated.
		Circuit Switched Data Activate	М	This information element requests to activate the interworking function.
		Circuit Switched Data	0	This information element indicates the request for IWF protocol negotiation result and rate change indication.
Activate Interworking	MGW	Context	М	This information element indicates the context where the command was executed.
Function Ack		Bearer Termination	М	This information element indicates the bearer termination where the command was executed.

16.2.9 Release Bearer

This procedure is used to release the bearer.

Procedure	Initiated	Information element name	Information element required	Information element description
Release Bearer	(G)MSC-S	Context	M	This information element indicates the context for the bearer termination.
		Bearer Termination	М	This information element indicates the bearer termination for the bearer to be released.
		Bearer Release Request	М	This information element requests release of a bearer.
		Release Cause	0	This information element indicates the cause of a bearer release.
Release Bearer Ack	MGW	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	М	This information element indicates the bearer termination where the command was executed.

Table 16.10: Procedures between (G)MSC server and MGW: Release Bearer

16.2.10 Bearer Established

This procedure is used to notify the established bearer.

Table 16.11: Procedures between (G)	ASC server and MGW: Bearer Established
-------------------------------------	---

Procedure	Initiated	Information element name	Information element required	Information element description
Bearer Established	MGW	Context	М	This information element indicates the context for the bearer termination.
		Bearer Termination	M	This information element indicates the bearer termination where the bearer was established.
		Bearer Established	М	This information element notifies a bearer establishment.
Bearer Established Ack	(G)MSC-S	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	M	This information element indicates the Bearer Termination where the command was executed.

16.2.11 Bearer Released

This procedure is used to notify the released bearer or failed bearer establishment.

able 16.12: Procedures betweer	(G)MSC server and	MGW: Bearer	Released
--------------------------------	-------------------	-------------	----------

Procedure	Initiated	Information element name	Information element required	Information element description
Bearer Released	MGW	Context	М	This information element indicates the context for the bearer termination.
		Bearer Termination	М	This information element indicates the bearer termination where the bearer was released.
		Bearer Released	М	This information element notifies a bearer release.
		Release Cause	М	This information element indicates the cause of a bearer release.
Bearer Released Ack	(G)MSC-S	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	М	This information element indicates the Bearer Termination where the command was executed.

16.2.12 Release Termination

This procedure is used to release the bearer termination.

Table 16.13: Procedures between (G)MSC server and MGW: Release Terr

Procedure	Initiated	Information element name	Information element required	Information element description
Release Termination	(G)MSC-S	Context	М	This information element indicates the context for the bearer termination.
		Bearer Termination	М	This information element indicates the bearer termination to be released.
Release Termination Ack	MGW	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	M	This information element indicates the bearer termination where the command was executed.
		T.140 data statistics	С	Number of t.140 data bits transmitted over the termination

16.2.13 Tunnel Information Up

This procedure is used to transfer tunnel data from the MGW to the (G)MSC server.

Fable 16.14: Procedures between	(G)MSC server and MGW	Tunnel Information Up
---------------------------------	-----------------------	------------------------------

Procedure	Initiated	Information element name	Information element required	Information element description
Tunnel Information Up	MGW	Context	М	This information element indicates the context for the bearer termination.
		Bearer Termination	M	This information element indicates the bearer termination from where the tunnel data is sent.
		Tunnel Data	М	This information element contains the tunnel data that is provided between MGWs.
Tunnel InformationUp Ack	(G)MSC-S	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	M	This information element indicates the Bearer Termination where the command was executed.

16.2.14 Tunnel Information Down

This procedure is used to transfer tunnel data from the (G)MSC server to the MGW.

Procedure	Initiated	Information element name	Information element required	Information element description
Tunnel Information Down	(G)MSC-S	Context	M	This information element indicates the context for the bearer termination.
		Bearer Termination	М	This information element indicates the bearer termination where to the tunnel data is sent.
		Tunnel Data	М	This information element contains the tunnel data that is provided between MGWs.
Tunnel Information Down	MGW	Context	М	This information element indicates the context where the command was executed.
Ack		Bearer Termination	М	This information element indicates the bearer termination where the command was executed.

NOTE: This procedure may be combined with Prepare Bearer or Establish Bearer procedure.

16.2.15 Send Tone

This procedure is used to send a tone.

Procedure	Initiated	Information element name	Information element required	Information element description
Send Tor (G)MSC:	ne -S	Context	М	This information element indicates the context for the bearer termination.
(0)			М	This information element indicates the existing bearer termination or requests a new bearer termination where the tone is sent.
		Tone	М	This information element indicates the tone to be generated.
		Notify Tone Completion	0	This information element requests a notification of a completed tone.
		Tone Direction	0	This information element indicates the tone direction in the bearer termination.
		Tone Timing	0	This information element indicates the time for the tone.
		Notify termination heartbeat	С	This information element requests termination heartbeat indications. This information element shall be included when requesting a new ephemeral bearer termination.
Send Tone Ack	MGW	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	М	This information element indicates the bearer termination where the command was executed.

Table 16.16: Procedures between (G)MSC server and MGW: Send Tone

NOTE: This procedure may be combined with Join Bearer Termination or Isolate Bearer Termination procedure.

16.2.16 Stop Tone

This procedure is used to stop the tone.

Procedure	Initiated	Information element name	Information element required	Information element description
Stop Tone	(G)MSC-S	Context	М	This information element indicates the context for the bearer termination.
		Bearer Termination	М	This information element indicates the bearer termination where the tone is stopped.
		Stop Tone	М	This information element requests that tone generation is stopped.
Stop Tone Ack	MGW	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	М	This information element indicates the bearer termination where the command was executed.

16.2.17 Play Announcement

This procedure is used to play an announcement.

Procedure	Initiated	Information element name	Information element required	Information element description
Play Annound (G)MSC-	Play Announcement		М	This information element indicates the context for the bearer termination.
		Bearer Termination/Bearer Termination Request	М	This information element indicates the existing bearer termination or requests a new bearer termination where the announcement is sent.
		Announcement	М	This information element indicates the announcement to be played.
		Notify Announcement Completion	0	This information element requests a notification of a completed announcement.
			0	This information element indicates the announcement direction in the bearer termination.
		Announcement Timing	0	This information element indicates the time for the announcement.
		Notify termination heartbeat	С	This information element requests termination heartbeat indications. This information element shall be included when requesting a new ephemeral bearer termination.
Play Announcement	MGW	Context	М	This information element indicates the context where the command was executed.
Ack		Bearer Termination	М	This information element indicates the bearer termination where the command was executed.

NOTE: This procedure may be combined with Join Bearer Termination or Isolate Bearer Termination procedure.

16.2.18 Stop Announcement

This procedure is used to stop the announcement.

Procedure	Initiated	Information element name	Information element required	Information element description
Stop Announcement	(G)MSC-S	Context	М	This information element indicates the context for the bearer termination.
		Bearer Termination	M	This information element indicates the bearer termination where the announcement is stopped.
		Stop Announcement	М	This information element requests that announcement playing is stopped.
Stop Announcement	MGW	Context	М	This information element indicates the context where the command was executed.
Ack		Bearer Termination	M	This information element indicates the bearer termination where the command was executed.

16.2.19 Announcement Completed

This procedure is used to notify the completed announcement.

Table 16.20: Procedures between	(G)MSC server a	nd MGW: Announcement	Completed
---------------------------------	-----------------	----------------------	-----------

Procedure	Initiated	Information element name	Information element required	Information element description
Announcement Completed	MGW	Context	М	This information element indicates the context for the bearer termination.
		Bearer Termination	M	This information element indicates the bearer termination where the announcement was completed.
		Announcement Completed	М	This information element indicates completion of the announcement.
Announcement Completed Ack	(G)MSC-S	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	M	This information element indicates the Bearer Termination where the command was executed.

16.2.20 Tone Completed

.

This procedure is used to notify the completed tone.

Table 16.21: Procedures betwo	en (G)MSC server and	MGW: Tone Comp	leted

Procedure	Initiated	Information element name	Information element required	Information element description
Tone Completed	MGW	Context	М	This information element indicates the context for the bearer termination.
		Bearer Termination	М	This information element indicates the bearer termination where the tone was completed.
		Tone Completed	М	This information element indicates completion of the tone.
Tone Completed Ack	(G)MSC-S	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	М	This information element indicates the Bearer Termination where the command was executed.

16.2.21 Detect DTMF

This procedure is used to request detection of a DTMF tone.

Procedure	Initiated	Information element name	Information element required	Information element description
Detect DTMF	(G)MSC-S	Context	М	This information element indicates the context for the bearer termination.
		Bearer Termination/Bearer Termination Request	М	This information element indicates the existing bearer termination or requests a new bearer termination where the DTMF tone detection is requested.
		Digit	М	This information element requests MGW to detect a DTMF tone.
Detect DTMF Ack	MGW	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	М	This information element indicates the bearer termination where the command was executed.

Table 16.22: Procedures between (G)MSC server and MGW: Detect DTMF

NOTE This procedure may be combined with Prepare Bearer, Prepare_IP_Transport, Establish Bearer, Reserve Circuit, Reserve RTP Connection Point procedure, Configure RTP Connection Point or Reserve and Configure RTP Connection Point. This list of procedures is not exhaustive.

16.2.22 Stop DTMF Detection

This procedure is used to stop detection of the DTMF tone.

Table 16.23: Procedures between (G)MS(server and MGW: S	Stop DTMF Detection
--	-------------------	---------------------

Procedure	Initiated	Information element name	Information element required	Information element description
Stop DTMF Detection	(G)MSC-S	Context	М	This information element indicates the context for the bearer termination.
		Bearer Termination	M	This information element indicates the bearer termination where the DTMF tone detection is stopped.
		Stop DTMF Detection	М	This information element requests that DTMF tone detection is stopped.
Stop DTMF Detection Ack	MGW	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	M	This information element indicates the bearer termination where the command was executed.

16.2.23 Report DTMF

This procedure is used to report a detected DTMF tone.

Table 16.24: Procedures	between	(G)MSC server	and MGW:	Report DTMF
		(-)		

Procedure	Initiated	Information element name	Information element required	Information element description
Report DTMF	MGW	Context	М	This information element indicates the context for the bearer termination.
		Bearer Termination	М	This information element indicates the bearer termination where the DTMF tone was detected.
		Digit	М	This information element reports the detected DTMF tone.
Report DTMF Ack	(G)MSC-S	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	M	This information element indicates the Bearer Termination where the command was executed.

16.2.24 Send DTMF

This procedure is used to request sending of a DTMF tone.

Table 16.25: Procedures be	etween (G)MSC server	and MGW: Send DTMF

Procedure	Initiated	Information element name	Information element required	Information element description
Send DTMF	(G)MSC-S	Context	М	This information element indicates the context for the bearer termination.
		Bearer Termination/Bearer Termination Request	М	This information element indicates the existing bearer termination or requests a new bearer termination where the DTMF tone generation is requested.
		Digit	М	This information element requests MGW to generate a DTMF tone.
		DTMF Tone Timing	0	This information element indicates the time for the DTMF tone in the bearer termination.
Send DTMF Ack	MGW	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	М	This information element indicates the bearer termination where the command was executed.

16.2.25 Stop DTMF

This procedure is used to stop sending of the DTMF tone.

Procedure	Initiated	Information element name	Information element required	Information element description
Stop DTMF	(G)MSC-S	Context	М	This information element indicates the context for the bearer termination.
		Bearer Termination	M	This information element indicates the bearer termination where the DTMF tone generation is stopped.
		Stop DTMF	М	This information element requests that DTMF tone generation is stopped.
Stop DTMF Ack	MGW	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	M	This information element indicates the bearer termination where the command was executed.

16.2.26 MGW Out-of-Service/ Maintenance locking

This procedure is used to indicate that the MGW will go out of service or is maintenance locked.

Table 16.27: Procedures between (G)MSC server and MGW: MGW Out-of-Service or Maintenance
Locked

Procedure	Initiated	Information element name	Information element required	Information element description
MGW Out-of- Service/	MGW	Context	М	This information element indicates the context for the command.
Maintenance locked		Root Termination	М	This information element indicates the root termination for the command.
		Reason	М	This information element indicates the reason for service change.
		Method	М	This information element indicates the method for service change.
MGW Out-of- Service/	(G)MSC-S	Context	М	This information element indicates the context where the command was executed.
Maintenance locked Ack		Root Termination	М	This information element indicates the root termination where the command was executed.

16.2.27 MGW Communication Up

This procedure is used to indicate that the MGW is back in service.

Up
ļ

Procedure	Initiated	Information element name	Information element required	Information element description
MGW Communication	MGW	Context	M	This information element indicates the context for the command.
Up		Root Termination	М	This information element indicates the root termination for the command.
		Reason	М	This information element indicates the reason for service change.
		Method	М	This information element indicates the method for service change.
MGW Commur Up Ack	nication	Context	М	This information element indicates the context where the command was executed.
(G)MSC-	S	Root Termination	M	This information element indicates the root termination where the command was executed.
		(G)MSC-S Address	0	This information element indicates the (G)MSC server signalling address to which the MGW should preferably attempt to reregister.

16.2.28 MGW Restoration

This procedure is used to indicate the MGW failure or recovery.

Procedure	Initiated	Information element name	Information element required	Information element description
MGW Restoration	MGW	Context	М	This information element indicates the context for the command.
		Root Termination	М	This information element indicates the root termination for the command.
		Reason	М	This information element indicates the reason for the service change.
		Method	М	This information element indicates the method for service change.
MGW Restorat (G)MSC-	ion Ack S	Context	М	This information element indicates the context where the command was executed.
		Root Termination	M	This information element indicates the root termination where the command was executed.
		(G)MSC-S Address	0	This information element indicates the (G)MSC server signalling address to which the MGW should preferably attempt to reregister.

16.2.29 MGW Register

This procedure is used to register the MGW.

Table 16.30: Procedures between	n (G)MSC server	r and MGW: MGW	Register
---------------------------------	-----------------	----------------	----------

Procedure	Initiated	Information element name	Information element required	Information element description
MGW Register	MGW	Context	M	This information element indicates the context for the command.
		Root Termination	М	This information element indicates the root termination for the command.
		Reason	М	This information element indicates the reason for the service change.
		Method	М	This information element indicates the method for service change.
		Protocol Version	M	This information element indicates the protocol version for Mc interface requested by the MGW.
		Service Change Profile	М	This information element indicates the profile for the Mc interface requested by the MGW.
MGW Registe (G)MSC-	er Ack S	Context	М	This information element indicates the context where the command was executed.
		Root Termination	М	This information element indicates the root termination where the command was executed.
		Protocol Version	0	This information element indicates the protocol version for Mc interface supported by the (G)MSC server.
		Service Change Profile	0	This information element indicates the profile for the Mc interface supported by the (G)MSC Server.
		(G)MSC-S Address	0	This information element indicates the (G)MSC server signalling address to which the MGW should preferably attempt to reregister.

16.2.30 MGW Re-register

This procedure is used to re-register the MGW.

Table 16	.31: Procedures	between	(G)MSC	server ar	nd MGW:	MGW	Re-register
----------	-----------------	---------	--------	-----------	---------	-----	--------------------

Procedure	Initiated	Information element name	Information element required	Information element description
MGW Re-register	MGW	Context	M	This information element indicates the
				context for the command.
		Root Termination	М	This information element indicates the root termination for the command.
		Reason	М	This information element indicates the reason for the service change.
		Method	М	This information element indicates the method for service change.
		Protocol Version	М	This information element indicates the protocol version for Mc interface requested by the MGW.
		Service Change Profile	М	This information element indicates the profile for the Mc interface requested by the MGW.
MGW Re-regis (G)MSC-	iter Ack S	Context	М	This information element indicates the context where the command was executed.
		Root Termination	М	This information element indicates the root termination where the command was executed.
		Protocol Version	0	This information element indicates the protocol version for Mc interface supported by the (G)MSC server.
		Service Change Profile	0	This information element indicates the profile for the Mc interface supported by the (G)MSC Server.
		(G)MSC-S Address	0	This information element indicates the (G)MSC server signalling address to which the MGW should preferably attempt to reregister.

16.2.31 (G)MSC Server Ordered Re-register

This procedure is used by the (G)MSC server to request the MGW to register itself.

Table 16.32: Procedures between (G)MSC server and MGW: (G)MSC Server Ordered Re-register

Procedure	Initiated	Information element name	Information element required	Information element description
(G)MSC Server Ordered Re-	(G)MSC-S	Context	М	This information element indicates the context for the command.
register		Root Termination	М	This information element indicates the root termination for the command.
		Reason	М	This information element indicates the reason for the service change.
		(G)MSC-S Address	0	This information element indicates the (G)MSC server signalling address.
(G)MSC Server Ordered Re-	MGW	Context	М	This information element indicates the context where the command was executed.
register Ack		Root Termination	М	This information element indicates the root termination where the command was executed.

16.2.32 (G)MSC Server Restoration

This procedure is used to indicate the (G)MSC server failure or recovery.

Table 16.33: Procedures between (G)MSC server and MGW: (G)MSC Server Restoration

Procedure	Initiated	Information element name	Information element required	Information element description
(G)MSC Server Restoration	(G)MSC-S	Context	М	This information element indicates the context for the command.
		Root Termination	М	This information element indicates the root termination for the command.
		Reason	М	This information element indicates the reason for the service change.
		Method	М	This information element indicates the method for service change.
(G)MSC Server Restoration Ack	MGW	Context	М	This information element indicates the context where the command was executed.
		Root Termination	М	This information element indicates the root termination where the command was executed.

16.2.33 (G)MSC Server Out of Service

This procedure is used to indicate that (G)MSC server has gone out of service.

Table 16.34: Procedures between (G)MSC server and MGW: (G)MSC Server Out of Service

Procedure	Initiated	Information element name	Information element required	Information element description
(G)MSC Server Out of Service	(G)MSC-S	Context	М	This information element indicates the context for the command.
		Root Termination	М	This information element indicates the root termination for the command.
		Reason	М	This information element indicates the reason for the service change.
		Method	М	This information element indicates the method for service change.
(G)MSC Server Out of Service	MGW	Context	М	This information element indicates the context where the command was executed.
Ack		Root Termination	M	This information element indicates the root termination where the command was executed.

16.2.34 Termination Out-of-Service

This procedure is used to indicate that physical termination(s) will go out of service.

Table 16.35: Procedures between (G)MSC	server and MGW: Termination Out-of-Service
--	--

Procedure	Initiated	Information element name	Information element required	Information element description
Termination Out- of-Service	MGW	Context	М	This information element indicates the context for the command.
		Bearer Termination	М	This information element indicates the bearer termination(s) for the command.
		Reason	М	This information element indicates the reason for service change.
		Method	М	This information element indicates the method for service change.
Termination Out- of-Service Ack	(G)MSC-S	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	М	This information element indicates the bearer termination where the command was executed.

16.2.35 Termination Restoration

This procedure is used to indicate that physical termination(s) are back in service.

Table 16.36: Procedures betwee	en (G)MSC server and MGW:	Termination Restoration
--------------------------------	---------------------------	--------------------------------

Procedure	Initiated	Information element name	Information element required	Information element description
Termination Restoration	MGW	Context	М	This information element indicates the context for the command.
		Bearer Termination	М	This information element indicates the bearer termination(s) for the command.
		Reason	М	This information element indicates the reason for service change.
		Method	М	This information element indicates the method for service change.
Termination Restoration Ack	(G)MSC-S	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	M	This information element indicates the bearer termination where the command was executed.

16.2.36 Audit Value

This procedure is used to audit values of different object(s).

Procedure	Initiated	Information element	Information element	Information element description
		namo	required	
Audit Value	(G)MSC-S	Context	М	This information element indicates the context for the command.
		Bearer Termination	М	This information element indicates the bearer termination(s) for the command.
		Object(s)	М	This information element indicates the object(s) to be audited.
Audit Value Ack	MGW	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	M	This information element indicates the bearer termination where the command was executed.
		Value(s)	М	This information element indicates the value(s) of the object(s).

Table 16.37: Procedures between (G)MSC server and MGW: Audit Value

16.2.37 Audit Capability

This procedure is used to audit capabilities of different object(s).

Table	16.38:	Procedures	between	(G)MSC	server ar	nd MGW:	Audit (Capability
				(-)				

Procedure	Initiated	Information element name	Information element required	Information element description
Audit Capability	(G)MSC-S	Context	М	This information element indicates the context for the command.
		Bearer Termination	М	This information element indicates the bearer termination(s) for the command.
		Object(s)	М	This information element indicates the object(s) which capability is requested.
Audit Capability Ack	MGW	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	M	This information element indicates the bearer termination where the command was executed.
		Capabilities(s)	М	This information element indicates the capabilities of the object(s).

16.2.38 Capability Update

This procedure is used to indicate update of an object capability.

Table 16.39: Procedures between (G)MSC server and MGW: Capability Update

Procedure	Initiated	Information element name	Information element required	Information element description
Capability Update	MGW	Context	М	This information element indicates the context for the command.
		Bearer Termination	М	This information element indicates the bearer termination(s) for the command.
		Reason	М	This information element indicates the reason for service change.
		Method	М	This information element indicates the method for service change.
Capability Update Ack	(G)MSC-S	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	М	This information element indicates the bearer termination where the command was executed.

16.2.39 Command Reject

This command is used to reject the received command request. It may be used as response to any of the above procedures.

Table 1	6.40:	Procedures	between	(G)M\$	SC serve	r and	MGW:	Command	Reject
				· ·					

Procedure	Initiated	Information element name	Information element required	Information element description
Command Reject	Both	Context	М	This information element indicates the context where the command was rejected.
		Bearer Termination	М	This information element indicates the bearer termination where the command was rejected.
		Error	М	This information element indicates the error that caused command rejection.

16.2.40 Activate Voice Processing Function

This procedure is used to activate the voice processing (echo cancellation) function.

Table 16.41: Procedures between (G)MSC server and MGW: Activate Voice Processing Function

Procedure	Initiated	Information element name	Information element required	Information element description
Activate Voice Processing	(G)MSC-S	Context	М	This information element indicates the context for the bearer termination.
Function		Bearer Termination	М	This information element indicates the bearer termination where the voice processing function is activated.
		Activate Voice Processing Function	М	This information element requests to activate the voice processing function.
Activate Voice Processing	MGW	Context	М	This information element indicates the context where the command was executed.
Function Ack		Bearer Termination	M	This information element indicates the bearer termination where the command was executed.

16.2.41 Modify Bearer Characteristics

This procedure is used to modify the bearer characteristics.

Table 16.42: Procedures between (G)MSC server and MGW: Modify Bearer Characteristics

Procedure	Initiated	Information element	Information	Information element description
		name	element	
			required	
Modify Bearer	(G)MSC-S	Context	M	This information element indicates the
Characteristics				existing context for the bearer termination.
		Bearer Termination	М	This information element indicates the bearer
				termination for the bearer to be modified.
		Bearer Service	С	This information element indicates the bearer
		Characteristics		service requested by the user. This
				information element is not included if the
				Codec information element is provided. It
				may be included for a data call (see
				3GPP TS 29.007 [37] and 3GPP TS 29.232
				[6]).
		Circuit Switched Data	С	This information element indicates the PLMN
				bearer capabilities and when applicable GSM
				channel coding and user bit rate. This
				information element shall be included
				according to 3GPP TS 29.007 [37] and 3GPP
				TS 29.232 [6] for a non-speech call by the
				MSC server, or by the anchor-MSC in case of
				inter-MSC handover, for UMTS radio access
				network side bearer termination and for
				network side bearer termination.
		Codec	С	This information element indicates the
				speech or multimedia coding format to be
				used for the bearer. This information element
				is included for the speech call and may be
				included for a SCUDIF multimedia call
				(see3GPP TS 23.172 [38], 3GPP TS 29.007
				[37] and 3GPP TS 29.232 [6])., for UMTS
				radio access network side bearer termination
				and for network side bearer termination.
		Framing Protocol	0	This information element indicates the
				framing protocol to be used for the bearer.
Modify Bearer	MGW	Context	М	This information element indicates the
Characteristics				context where the command was executed.
Ack		Bearer Termination	М	This information element indicates the bearer
				termination where the command was
				executed.

16.2.42 Protocol Negotiation Result

This procedure is used to inform the MSC about protocol negotiation result.

Procedure	Initiated	Information element name	Information element required	Information element description
IWF Protocol Indication	MGW	Context	М	This information element indicates the existing context for the bearer termination.
		Bearer Termination	М	This information element indicates the bearer termination for the bearer to be modified.
		Protocol Negotiation result	М	This information element indicates the IWF protocol negotiation result
IWF Protocol Indication Ack	MSC	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	М	This information element indicates the bearer termination where the command was executed.

Table 16.43: Procedures between MSC server and MGW: Protocol Negotiation Result

16.2.43 MGW Resource Congestion Handling - Activate

This procedure is used to activate the congestion handling mechanism.

Table 16.44: Procedures between (G)MSC server and MGW: MGW Resource Congestion Handling Activate

Procedure	Initiated	Information element name	Information element required	Information element description
MGW Resource Congestion Handling – Activate	(G)MSC-S	Context	М	This information element indicates that all context are applicable for the root termination.
		Root Termination	М	This information element indicates that root termination is where the congestion mechanism is activated.
		Congestion Activate	М	This information element requests to activate the congestion mechanism.
MGW Resource Congestion Handling -	MGW	Context	М	This information element indicates that all context are where the command was executed.
Activate Ack		Root Termination	М	This information element indicates that root termination is where the command was executed.

16.2.44 MGW Resource Congestion Handling - Indication

This procedure is used to inform the (G)MSC server that traffic restriction is advised.

Table 16.45: Procedures between (G)MSC server and MGW: MGW Resource Congestion Handling -Indication

Procedure	Initiated	Information element name	Information element required	Information element description
MGW Resource Congestion	MGW	Context	М	This information element indicates all context are applicable for the root termination.
Handling - Indication		Root Termination	М	This information element indicates that root termination is where the congestion mechanism was activated.
		Reduction	М	This information element indicates the load percentage to be reduced.
MGW Resource Congestion Handling - Indication Ack	(G)MSC	Context	М	This information element indicates all context are where the command was executed.
		Root Termination	М	This information element indicates that root termination is where the command was executed.

16.2.45 Bearer Modification Support

This procedure is used to notify that the established bearer can be modified.

Table 16.46: Procedures between (G)MSC server and MGW: Bearer Modification

Procedure	Initiated	Information element name	Information element required	Information element description
Bearer Modification	MGW	Context	М	This information element indicates the context for the bearer termination.
Support		Bearer Termination	М	This information element indicates the bearer termination where the bearer was established.
		Bearer Modification	М	This information element notifies that the established bearer can be modified.
Bearer Modification	(G)MSC-S	Context	М	This information element indicates the context where the command was executed.
Support Ack		Bearer Termination	M	This information element indicates the Bearer Termination where the command was executed.

16.2.46 CTM report

This procedure is used to notify the outcome of the CTM negotiation in the user plane.

Table 16.47: Procedures between (G)	MSC server and MGW: CTM report
-------------------------------------	--------------------------------

Procedure	Initiated	Information element name	Information element required	Information element description
CTM report	MGW	Context	М	This information element indicates the context for the bearer termination.
		Bearer Termination	М	This information element indicates the bearer termination where the CTM function was activated
		Outcome of CTM negotiation	M	This information element indicates whether the CTM negotiation in user plane was successful or not.
CTM report Ack	(G)MSC-S	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	M	This information element indicates the Bearer Termination where the command was executed.

16.2.47 Prepare_IP_Transport

This procedure is used to prepare for a bearer establishment when IuCS on IP is supported by the MSC.

Procedure	Initiated	Information element	Information	Information element description
		name	element	
			required	
Prepare_IP_Tr	ansport	Context/Context	M	This information element indicates the
(G)MSC-	S	Request		existing context or requests a new context for
		Deexex Terrein etier	N.4	the IP Access bearer termination.
		Bearer Termination	IVI	I his information element requests a new
		Request		bearer termination for the IP Access bearer to
		In LIDB Port Poguest	Ν.4	This information element requests the lu LIDD
		IU ODF FOIT Request	IVI	Port in the MGW
		IP Transport Address	М	This information element requests the IP
		Request	IVI	address of the MGW
		Bearer Characteristics/	М	This information element indicates the
		Bearer Characteristics		preferred characteristics of the bearer
		Requests		connection or requests the MGW to select
				and provide the bearer characteristics.
		Bearer Service	С	This information element indicates the bearer
		Characteristics		service requested by the user. This
				information element is not included if the
				Codec information element is provided. It
				may be included for a data call (see
				3GPP TS 29.007 [37] and 3GPP TS 29.232
				[6]).
		Notify Bearer Event	0	This information element requests a
				notification of an established bearer, a
				released bearer, a modified bearer or a
			-	bearer modification failure.
		Circuit Switched Data	C	I his information element indicates the PLMN
				bearer capabilities and when applicable the
				be included according to 2000 TS 20.007
				[37] and 3GPP TS 29 232 [6] for a non-
				speech call by the MSC server, or by the
				anchor-MSC in case of inter-MSC handover
				for a UMTS radio access network side bearer
				termination.
		Codec	С	This information element indicates the
				speech or multimedia coding format to be
				used for the bearer. This information element
				is included for a speech call and may be
				included for a SCUDIF multimedia call
				(see3GPP TS 23.172 [38], 3GPP TS 29.007
				[37] and 3GPP TS 29.232 [6]), for a UMTS
				radio access network side bearer termination.
		Framing Protocol	0	This information element indicates the
			-	traming protocol to be used for the bearer.
		Cellular Text telephony	C	I his information element indicates the need
			0	Of CIM function.
		IP Realm Identifier	0	realm of the IP termination
		Notify termination	Ν.4	This information element requests
		heartheat	IVI	termination heartheat indications
Prenare IP	MGW/	Context	M	This information element indicates the
Transport Ack	101000	JOINEAL	111	context where the command was executed
		Bearer Termination	М	This information element indicates the bearer
				termination where the command was
				executed.
		IP Transport Address	М	This information element indicates the IP
		,		address of the MGW

	lu UDP Port	М	This information element requests the lu UDP
			Port in the MGW

16.2.48 Modify IP Transport Address

This procedure is used when IuCS on IP is supported by the MGW and IuUP in transparent mode is configured.

Table 16.49: Procedures between (G)MSC server and MGW: RNC IP address notification

Procedure	Initiated	Information element name	Information element required	Information element description
Modify IP Transport Address	MSC-S	Context	М	This information element indicates the context for the IP bearer termination.
		Bearer Termination	Μ	This information element indicates the IP bearer termination where the RNC IP Address is needed.
		IP Transport address	М	This information element indicates the IP address of the RNC
		lu UDP Port	М	This information element indicates the lu UDP Port in the RNC
Modify IP Address Ack	MGW	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	M	This information element indicates the IP bearer termination where the command is executed.

16.2.49 Emergency Call Indication

This procedure is used to indicate that the call is an emergency call.

Table 16.50: Procedures between (G)MSC server and MGW: Emergency Call Indication

Procedure	Initiated	Information element name	Information element required	Information element description
Emergency Call Indication	(G)MSC-S	Context Request	M	This information element indicates the existing context or requests a new context for the bearer termination.
		Emergency Call Indicator	Μ	This information element indicates the emergency call information
Emergency Call Indication Ack	MGW	Context	М	This information element indicates the context where the command was executed.

16.2.50 Trace Activation

This procedure is used for activation of a Trace Session in a MGW.

Procedure	Initiated	Information element name	Information element required	Information element description
Trace Activation	MSC-S	Context	М	This information element indicates the
				context for the command.
		Bearer Termination	М	This information element indicates the bearer termination(s) for the command.
		Trace Reference	С	Defined in 3GPP TS 32.421 and 3GPP TS 32.422
		Trace Session	С	Defined in 3GPP TS 32.421 and 3GPP TS
		Recording Reference	-	32.422
		Triggering Events	С	Defined in 3GPP TS 32.421 and 3GPP TS 32.422
		Trace Depth	С	Defined in 3GPP TS 32.421 and 3GPP TS 32.422
		List of interfaces	0	Defined 3GPP TS 32.422
		Trace Activity Control	М	This information element indicates the trace activation)
		IMSI	С	This information element shows the IMSI of the traced subscriber. Either IMSI or IMEI(SV) must be provided.
		IMEI(SV)	С	This information element shows the IMEI(SV) of the traced equipment. Either IMSI or IMEI(SV) must be provided.
		Notify Trace Activation result	0	This information element requests a notification of the result of the trace activation.
Trace Activation Reply	MGW	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	M	This information element indicates the bearer termination where the command was executed.

Table 16.51: Procedures between (G)MSC server and MGW: Trace Activation

16.2.51 Trace Deactivation

This procedure is used for deactivation of a Trace Session in a MGW..

Table 16.52: Procedures	between (G)MSC	server and MGW:	Trace Deactivation
-------------------------	----------------	-----------------	---------------------------

Procedure	Initiated	Information element name	Information element required	Information element description
Trace Deactivation	MSC-S	Context	М	This information element indicates the context for the command.
		Bearer Termination	М	This information element indicates the bearer termination(s) for the command.
		Trace Reference	М	Defined in 3GPP TS 32.421 and 3GPP TS 32.422
		Trace Activity Control	М	This information element indicates the trace deactivation
Trace Deactivation Reply	MGW	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	М	This information element indicates the bearer termination where the command was executed

16.2.52 Trace Activation result notification

This procedure is used for informing the MSC Server about the result of the Trace Session Activation.

Procedure	Initiated	Information element name	Information element required	Information element description
Trace Activation result notification	MGW	Context	М	This information element indicates the context for the command
		Bearer Termination	М	This information element indicates the bearer termination(s) for the command
		Result	М	This information element defines the result of the Trace Session Activation
Trace Activation result notification Ack	MSC-S	Context	М	This information element indicates the context where the command was executed
		Bearer Termination	М	This information element indicates the bearer termination where the command was

Table 16.53: Procedures between MGW and (G)MSC Server: Trace Activation result notification

16.2.53 Continuity Check Tone

This procedure is used by the (G)MSC to order the MGW to generate a continuity check tone at a TDM termination and to inform the (G)MSC about the result of the continuity check as soon as the continuity check tone is received or a time-out occurs.

executed

Procedure	Initiated	Information element name	Information element required	Information element description
Continuity Check Tone	(G)MSC	Context/Context Request	М	This information element indicates the existing context or requests a new context for the bearer termination.
		TDM Termination	М	This information element indicates the bearer termination
		Request for continuity tone sending	М	This information request the MGW to apply the continuity check procedure on the indicated TDM termination
		Request for continuity check tone detection	М	This information request the MGW to inform of result of the continuity check procedure on the indicated TDM termination
Continuity Check Tone Ack	IMGW	Context	М	This information element indicates the context where the command was executed.

Table 16.54: Continuity Check Tone procedure

16.2.54 Continuity Check Verify

This procedure is used by the MGW to indicate towards the (G)MSC that the continuity check at a TDM termination has been completed and to return the result of the check: success or failure.

248

Procedure	Initiated	Information element name	Information element required	Information element description
Continuity check Verify	MGW	Context/t	М	This information element indicates the context where the command was executed.
		TDM Termination	М	This information element indicates the TDM termination involved in the procedure
		Outcome of the continuity check	М	This information element indicates the outcome of the continuity check (successful/unsuccessful)
Continuity Check Verify Ack	(G)MSC	Context	М	This information element indicates the context where the command was executed.

Table 16.55: Continuity Check Verify procedure

16.2.55 Continuity Check Response

This procedure is used by the (G)MSC to order the MGW to loop back an incoming continuity check tone at a TDM termination.

Procedure	Initiated	Information element name	Information element required	Information element description
Continuity Check Response	(G)MSC	Context/Context Request	Μ	This information element indicates the existing context or requests a new context for the bearer termination.
		TDM Termination	М	This information element indicates the bearer termination
		Request for loop back of the continuity tone	М	This information request the MGW to loop back the continuity check tone on the indicated TDM termination
Continuity Check Response Ack	IMGW	Context	М	This information element indicates the context where the command was executed.

Table 16.56: Continuity Check Response procedure

16.2.56 Rate Change

This procedure is used to notify that the CS data rate should be changed.

Table 16.57: Procedures between (G)MSC server and MGW: Rate Change

Procedure	Initiated	Information element name	Information element required	Information element description
Rate Change	MGW	Context	М	This information element indicates the context for the bearer termination.
		Bearer Termination	М	This information element indicates the bearer termination where the rate change is required.
		Rate	М	This information element notifies the current rate.
Rate Change Ack	(G)MSC-S	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	М	This information element indicates the bearer termination where the command was executed.

16.2.57 Bearer Modifed

This procedure is used to notify the modified bearer.

Table 16.2.57/1: Procedures betwee	n (G)MSC server an	nd MGW: Bearer Modified
------------------------------------	--------------------	-------------------------

Procedure	Initiated	Information element name	Information element required	Information element description
Bearer Modified	MGW	Context	М	This information element indicates the context for the bearer termination.
		Bearer Termination	М	This information element indicates the bearer termination where the bearer was modified.
		Bearer Modified	М	This information element notifies a bearer modification.
Bearer Modified Ack	(G)MSC-S	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	М	This information element indicates the bearer termination where the command was executed.

16.2.58 Bearer Modification Failed

This procedure is used to notify the failed bearer modification.

Table 16.2.58/1: Procedures between	(G)MSC server and MGW:	. Bearer Modification failed
	(-)	

Procedure	Initiated	Information element name	Information element required	Information element description
Bearer Modification Failed	MGW	Context	М	This information element indicates the context for the bearer termination.
		Bearer Termination	М	This information element indicates the bearer termination where the bearer modification failed.
		Bearer Modification failure	М	This information element notifies a bearer modification failure
Bearer Modification Failed	(G)MSC-S	Context	М	This information element indicates the context where the command was executed.
Ack		Bearer Termination	М	This information element indicates the bearer termination where the command was executed.

16.2.59 Termination heartbeat indication

This procedure is used to report indication of hanging termination.

Table 16.2.59/1: Procedures between (G)MSC server and MGW: Hanging termination indication

Procedure	Initiated	Information element name	Information element required	Information element description
Termination heartbeat	MGW	Context	М	This information element indicates the context for the bearer termination.
indication		Bearer Termination	М	This information element indicates the bearer termination for which the termination heartbeat is reported.
		Termination heartbeat	М	Hanging termination event as defined in 3GPP TS 29.232 [6].
Termination heartbeat	(G)MSC-S	Context	М	This information element indicates the context where the command was executed.
indication Ack		Bearer Termination	М	This information element indicates the Bearer Termination where the command was executed.

16.2.60 Inactivity timeout activation

This procedure is used to activate the inactivity timeout mechanism.

Table 16.2.60/1: Procedures between (G)MSC server and MGW: Inactivity timeout activation

Procedure	Initiated	Information element name	Information element required	Information element description
Inactivity timeout activation	(G)MSC-S	Context	М	This information element indicates the context for the command.
		Root Termination	М	This information element indicates that root termination is where the inactivity timeout mechanism is activated.
		Inactivity Timeout	М	Inactivity timeout event as defined in 3GPP TS 29.232 [6].
Inactivity timeout activation Ack	MGW	Context	М	This information element indicates the context where the command was executed.
		Root Termination	М	This information element indicates that root termination is where the command was executed.

16.2.61 Inactivity timeout indication

This procedure is used by MGW to report indication of inactivity timeout to MGC.

Table 16.2.61/1: Procedures between (G)MSC server and MGW: Inactivity timeout indication

Procedure	Initiated	Information element name	Information element required	Information element description
Inactivity timeout indication	MGW	Context	М	This information element indicates the context for the command.
		ROOT Termination	М	This information element indicates that root termination is where the inactivity timeout mechanism was activated.
		Inactivity Timeout	М	Inactivity timeout event as defined in 3GPP TS 29.232 [6].
Inactivity timeout indication Ack	(G)MSC-S	Context	М	This information element indicates the context where the command was executed.
		ROOT Termination	М	This information element indicates that root termination is where the command was executed.

16.2.62 Reserve RTP Connection Point

This procedure is used to reserve an RTP bearer termination.
Procedure	Initiated	Information element name	Information element required	Information element description
Reserve RTP Connection Point	(G)MSC- Server	Context/Context Request	M	This information element indicates the existing context or requests a new context for the bearer termination.
		Termination Request	М	This information element requests a new termination for the bearer to be established.
		IP Interface	0	This information element specifies the type of external interface to be used for the IP termination (e.g. AoIP).
		Local IP Resources	М	This information element indicates the resource(s) (e.g. codec, auxiliary payload types) for which the MGW shall be prepared to receive user data,
		ReserveValue	C	This information element indicates if multiple local resources are to be reserved. This information element shall be included if a speech codec and auxiliary pavload types are required.
		Local Connection Address Request	М	This information element requests an IP address and port number on the MGW that the remote end can send user plane data to.
		Notify termination	М	This information element requests
		Notify Released	0	This information element requests a notification of a released bearer
		IP Realm Identifier	0	This information element indicates the IP
		Cellular Text telephony modem	C	This information element indicates the need of CTM function. (NOTE 1)
		Number of desired listener context terminations	0	This information element indicates the number of desired terminations for a listener context in a VGCS or VBS call. It should be included in the first ADD request which establishes a Listener context. (NOTE 1)
		Circuit Switched Data	С	This information element indicates the PLMN bearer capabilities and when applicable GSM channel coding. This information element shall be included according to 3GPP TS 29.007 [37] and 3GPP TS 29.232 [6] for a non-speech / non SCUDIF multimedia call (see 3GPP TS 23.172 [20]) by the MSC server, or by the anchor-MSC in case of inter-MSC handover, for an anchor MGW network termination.
		IP Version	C	This information element indicates the version of the internet protocol to be applied at the termination for the user plane. This is required for IP address translation. (NOTE 4)
		DiffServ Code Point	0	This information element indicates a specific DiffServ code point to be used in the IP header in packets sent on the IP termination. (NOTE 4)

Table 16.2.62.1: Reserve RTP Connection Point

DiffServ Tagging Behaviour	0	This information element indicates whether the Diffserv code point in the IP header in packets sent on the IP termination should be copied from the received value or set to a specific value. (NOTE 4) This information element indicates that
Address Filtering	0	remote source address filtering is required. (NOTE 4)
Remote Source Address Mask	С	This information element provides information on the valid remote source addresses. This may be included if remote source address filtering is included. It shall not be included if remote source address filtering is not included. (NOTE 4)
Remote Source Port Filtering	0	This information element indicates that remote source port filtering is required. (NOTE 4)
Remote Source Port	C	This information element identifies the valid remote source port. This may be included if remote source port filtering is included. It shall not be included if remote source port filtering is not included. (NOTE 2, NOTE 4)
Remote Source Port Range	C	This information element identifies a range of valid remote source ports. This may be included if remote source port filtering is included. It shall not be included if remote source port filtering is not included. (NOTE 2, NOTE 4)
Media Inactivity Detection Required	0	This information element indicates that detection of inactive media flows is required. (NOTE 4)
Inactivity Detection Time	С	This information element may be present if Inactive Media Detection is required and specifies the Inactivity Detection time. (NOTE 4)
Inactivity Detection Direction	С	This information element may be present if Inactive Media Detection is required and specifies the Inactivity Detection direction. (NOTE 4)
RTCP handling	0	Indicates whether or not the CS-MGW shall reserve a port for an RTCP flow. (NOTE 4)
Traffic Policing Required	0	This information element indicates that policing of the media flow is required. (NOTE 4)
Peak Data Rate	0	This information element may be present if Policing is required and specifies the permissible peak data rate for a media stream. (NOTE 3, NOTE 4)
Sustainable Data Rate	0	This information element may be present if Policing is required and specifies the permissible sustainable data rate for a media stream. (NOTE 3, NOTE 4)
Delay Variation Tolerance	0	This information element may be present if Policing on Peak Data Rate is required and specifies the maximum expected delay variation tolerance for the corresponding media stream. (NOTE 4)
Maximum Burst Size	С	This information element shall be present if Policing on Sustainable Data Rate is required and specifies the maximum expected burst size for the corresponding media stream. (NOTE 4)

Reserve RTP Connection Point Ack	MGW	Context	М	This information element indicates the context where the command was executed.
		Termination	М	This information element indicates the termination where the command was executed.
		Local IP Resources	М	This information element indicates the resources that the MGW has reserved to receive the user plane data from the remote peer.
		Local Connection Address	М	This information element indicates the IP address and port on the MGW that shall receive user plane data from the remote peer.
NOTE 1: Only app NOTE 2: Remote NOTE 3: At least	Source Port and one of these IEs	side terminations d Remote Source Port I s shall be present when	Range are mutually	y exclusive. d.

NOTE 4: This information element is used for border control functions as defined in Annex A of 3GPP TS 29.235 [44]

16.2.63 Configure RTP Connection Point

This procedure is used to configure or reconfigure an RTP bearer termination.

Procedure	Initiated	Information element name	Information element required	Information element description						
Configure RTP Connection Point	(G)MSC- Server	Context	M	This information element indicates the existing context.						
		Termination	М	This information element indicates the existing bearer termination.						
		IP Interface	0	This information element specifies the type of external interface to be used for the IP termination (e.g. AoIP).						
		Local IP Resources	0	This information element indicates the resources (e.g. codec, auxiliary payload types) that the MGW may use on the reception of user plane data.						
		Remote IP Resources	0	This information element indicates the resources (e.g. codec, auxiliary payload types) that the MGW may send user plane data to.						
		Local Connection Address	0	This information element indicates the IP address and port on the MGW that the remote peer can send user plane data to.						
		Remote Connection Address	0	This information element indicates the IP address and port that the MGW can send user plane data to.						
		Reserve Value	С	This information element indicates if multiple resources are to be reserved. This information element shall be included if a speech codec and auxiliary payload types are required.						
		Cellular Text telephony modem	C	This information element indicates the need of CTM function. (NOTE 1)						
		Number of desired listener context terminations	0	This information element indicates the number of desired terminations for a listener context in a VGCS or VBS call. It should be included in the first ADD request which establishes a Listener context. (NOTE 1)						
		Circuit Switched Data	Ο	This information element indicates the PLMN bearer capabilities and when applicable GSM channel coding. This information element shall be included according to 3GPP TS 29.007 [37] and 3GPP TS 29.232 [6] for a non-speech/ non SCUDIF multimedia call by the MSC server, or by the anchor-MSC in case of inter-MSC handover for an anchor MGW network termination.						
								DiffServ Code Point	0	This information element indicates a specific DiffServ code point to be used in the IP header in packets sent on the IP termination. (NOTE 4)
				DiffServ Tagging Behaviour	0	This information element indicates whether the Diffserv code point in the IP header in packets sent on the IP termination should be copied from the received value or set to a specific value. (NOTE 4)				
		Remote Source Address Filtering	0	This information element indicates that remote source address filtering is required. (NOTE 4)						

Table 16.2.63.1: Configure RTP Connection Point Procedure

		Remote Source Address Mask	С	This information element provides information on the valid remote source addresses. This may be included if remote source address filtering is included. It shall not be included if remote source address filtering is not included. (NOTE 4)
		Remote Source Port Filtering	0	This information element indicates that remote source port filtering is required. (NOTE 4)
		Remote Source Port	С	This information element identifies the valid remote source port. This may be included if remote source port filtering is included. It shall not be included if remote source port filtering is not included. (NOTE 2, NOTE 4)
		Remote Source Port Range	С	This information element identifies a range of valid remote source ports. This may be included if remote source port filtering is included. It shall not be included if remote source port filtering is not included. (NOTE 2, NOTE 4)
		Media Inactivity Detection Required	0	This information element indicates that detection of inactive media flows is required. (NOTE 4)
		Inactivity Detection Time	C	This information element may be present if Inactive Media Detection is required and specifies the Inactivity Detection time. (NOTE 4)
		Inactivity Detection Direction	C	This information element may be present if Inactive Media Detection is required and specifies the Inactivity Detection direction. (NOTE 4)
		RTCP handling	0	Indicates whether or not the CS-MGW shall reserve a port for an RTCP flow. (NOTE 4)
		Traffic Policing Required	0	This information element indicates that policing of the media flow is required. (NOTE 4)
		Peak Data Rate	0	This information element may be present if Policing is required and specifies the permissible peak data rate for a media stream.(NOTE 3, NOTE 4)
		Sustainable Data Rate	0	This information element may be present if Policing is required and specifies the permissible sustainable data rate for a media stream.(NOTE 3, NOTE 4)
		Delay Variation Tolerance	0	This information element may be present if Policing on Peak Data Rate is required and specifies the maximum expected delay variation tolerance for the corresponding media stream. (NOTE 4)
		Maximum Burst Size	С	This information element shall be present if Policing on Sustainable Data Rate is required and specifies the maximum expected burst size for the corresponding media stream. (NOTE 4)
Configure RTP Connection Point Ack	MGW	Context	М	This information element indicates the context where the command was executed.
		Termination	М	This information element indicates the termination where the command was executed.
		Local IP Resource	0	This information element indicates the resources that the MGW has reserved to receive the user plane data from the far end.

		Remote IP Resource	С	This information element indicates the resource (i.e. codec) that the MGW shall use to send user data to. Shall be present if corresponding IE is present in the request.	
		Local Connection Address	0	This information element indicates the IP address and port on the MGW that the remote end can send user plane data to.	
		Remote Connection Address	С	This information element indicates the IP address and port that the MGW can send user plane data to. Shall be present if corresponding IE is present in the request.	
NOTE 1: Only app	olies for Access	side terminations			
NOTE 2: Remote	NOTE 2: Remote Source Port and Remote Source Port Range are mutually exclusive.				
NOTE 3: At least	NOTE 3: At least one of these IEs shall be present when policing is required.				
NOTE 4: This info	NOTE 4: This information element is used for border control functions as defined in Annex A of 3GPP TS 29.235 [44]				

16.2.64 Reserve and Configure RTP Connection Point

This procedure is used to reserve and configure multimedia-processing resources for an RTP termination.

 Table 16.2.64.1: Reserve and Configure RTP Connection Point

Procedure	Initiated	Information element name	Information element required	Information element description
Reserve and Configure RTP Connection Point	(G)MSC- Server	Context/Context Request	М	This information element indicates the existing context or requests a new context for the bearer termination.
		Termination/ Termination Request	М	This information element indicates the existing bearer termination or requests a new termination for the bearer to be established.
		IP Interface	0	This information element specifies the used interface type for the IP termination (e.g. NboIP).
		Local IP Resources	М	This information element indicates the resource(s) (e.g. codec, auxiliary payload types) for which the MGW shall be prepared to receive user data,
	Remote IP Resources	М	This information element indicates the resources (e.g. codec, auxiliary payload types) that the MGW shall use to send user data.	
	Reserve Value	С	This information element indicates if multiple IP resources are to be reserved. This information element shall be included if a speech codec and auxiliary payload types are required.	
	Cellular Text telephony modem	С	This information element indicates the need of CTM function. (NOTE 1)	
		Local Connection Address Request	М	This information element requests an IP address and port number on the MGW that the remote end can send user plane data to.

Remote Connection Address	М	This information element indicates the IP address and ports of the remote party that the MGW can send user plane data to.
Notify termination heartbeat	М	This information element requests termination heartbeat indications.
Notify Released Bearer	0	This information element requests a notification of a released bearer.
IP Realm Identifier	0	This information element indicates the IP realm of the IP termination.
Circuit Switched Data	0	This information element indicates the PLMN bearer capabilities and when
		applicable GSM channel coding. This information element shall be included according to 3GPP TS 29.007 [6] and 3GPP TS 29.232 [8] for a non-speech / non SCUDIF multimedia call (see 3GPP TS 23.172 [20]) by the MSC server, or by the anchor-MSC in case of inter-MSC handover, for an anchor MGW network termination.
IP Version	С	This information element indicates the version of the internet protocol to be applied at the termination for the user plane. This is required for IP address translation. (NOTE 4)
DiffServ Code Point	0	This information element indicates a specific DiffServ code point to be used in the IP header in packets sent on the IP termination. (NOTE 4)
DiffServ Tagging Behaviour	0	This information element indicates whether the Diffserv code point in the IP header in packets sent on the IP termination should be copied from the received value or set to a specific value. (NOTE 4)
Remote Source Address Filtering	0	This information element indicates that remote source address filtering is required. (NOTE 4)
Remote Source Address Mask	С	This information element provides information on the valid remote source addresses. This may be included if remote source address filtering is included. It shall not be included if remote source address filtering is not included. (NOTE 4)
Remote Source Port Filtering	0	This information element indicates that remote source port filtering is required. (NOTE 4)
Remote Source Port	С	This information element identifies the valid remote source port. This may be included if remote source port filtering is included. It shall not be included if remote source port filtering is not included. (NOTE 2, NOTE 4)
Remote Source Port Range	С	This information element identifies a range of valid remote source ports. This may be included if remote source port filtering is included. It shall not be included if remote source port filtering is not included. (NOTE 2, NOTE 4)
Media Inactivity Detection Required	0	This information element indicates that detection of inactive media flows is required. (NOTE 4)
Inactivity Detection Time	С	This information element may be present if Inactive Media Detection is required and specifies the Inactivity Detection time. (NOTE 4)
Inactivity Detection Direction	C	This information element may be present if Inactive Media Detection is required and specifies the Inactivity Detection direction. (NOTE 4)

		RTCP handling	0	Indicates whether or not the CS-MGW shall reserve a port for an RTCP flow. (NOTE 4)
		Traffic Policing Required	0	This information element indicates that policing of the media flow is required. (NOTE 4)
		Peak Data Rate	0	This information element may be present if Policing is required and specifies the permissible peak data rate for a media stream.(NOTE 3, NOTE 4)
		Sustainable Data Rate	0	This information element may be present if Policing is required and specifies the permissible sustainable data rate for a media stream.(NOTE 3, NOTE 4)
		Delay Variation Tolerance	0	This information element may be present if Policing on Peak Data Rate is required and specifies the maximum expected delay variation tolerance for the corresponding media stream. (NOTE 4)
		Maximum Burst Size	С	This information element shall be present if Policing on Sustainable Data Rate is required and specifies the maximum expected burst size for the corresponding media stream. (NOTE 4)
Reserve and Configure RTP	MGW	Context	M	This information element indicates the context where the command was executed.
Connection Point		Termination	М	This information element indicates the termination where the command was executed.
		Local IP Resources	М	This information element indicates the resources that the MGW has reserved to receive the user plane data from the remote side.
		Remote IP Resources	M	This information element indicates the resource (i.e. codec) that the MGW shall use to send user data.
		Local Connection Addresses	M	This information element indicates the IP address and port on the MGW that shall receive user plane data.
NOTE 1: Only app	lies for Acces	ss side terminations		
NOTE 2: Remote	Source Port a	and Remote Source Port F	Range are mut	ually exclusive.

NOTE 3: At least one of these IEs shall be present when policing is required. NOTE 4: This information element is used for border control functions as defined in Annex A of 3GPP TS 29.235 [44]

16.2.65 Realm Availability Change - Activation

This procedure is used to activate the realm availability notification mechanism.

Table 16.2.65.1: Realm	Availability Chang	Je - Activation
------------------------	--------------------	-----------------

Procedure	Initiated	Information element name	Information element required	Information element description
Realm Availability Change – Activate	(G)MSC- Server	Context	М	This information element indicates the NULL context.
		Root Termination	М	This information element indicates that root termination is where the realm availability change notification mechanism is activated.
		Realm Availability Change Activate	M	This information element requests to activate the realm availability change notification mechanism.
Realm Availability Change -Activate	MGW	Context	М	This information element indicates the NULL context.
Ack		Root Termination	M	This information element indicates that root termination is where the command was executed

NOTE: This procedure is used for border control functions as defined in Annex A of 3GPP TS 29.235 [44]

16.2.66 Realm Availability Change - Indication

This procedure is used to inform the (G)MSC-Server that there has been a change to the list of currently available realms.

Initiated	Information element name	Information element required	Information element description
MGW	Context	М	This information element indicates the NULL context.
	Root Termination	M	This information element indicates that root termination is where the realm availability change notification mechanism was activated.
	Changes to Realm Availability	М	This information element indicates the changes to the list of available realms.
(G)MSC- Server	Context	М	This information element indicates the NULL context.
	Root Termination	М	This information element indicates that root termination is where the command was executed.
	Initiated MGW (G)MSC- Server	InitiatedInformation element nameMGWContextRoot TerminationChanges to Realm Availability(G)MSC- ServerContextRoot Termination	InitiatedInformation element nameInformation element requiredMGWContextMMGWRoot TerminationMRoot TerminationMChanges to Realm AvailabilityM(G)MSC- ServerContextMRoot TerminationM

Table 16.2.66.1: Realm Availability Change - Indication

16.2.67 Media Inactivity Notification

This procedure is used to notify the (G)MSC-Server of media inactivity on the MGW.

Procedure	Initiated	Information element name	Information element required	Information element description
Media Inactivity MGW Notification		Context	М	This information element indicates the context for the bearer termination.
		Bearer Termination	M	This information element indicates the bearer termination where the media inactivity detection was activated.
		Media Inactivity	M	This information element notifies the (G)MSC-Server of Media inactivity detection on the bearer termination.
Media Inactivity Notification Ack	(G)MSC- Server	Context	М	This information element indicates the context where the command was executed.
		Bearer Termination	M	This information element indicates the bearer termination where the command was executed.

NOTE: This procedure is used for border control functions as defined in Annex A of 3GPP TS 29.235 [44]

17 Bearer Redirect

The BICC [22] Bearer Redirect mechanism may be used for optimising the bearer path when an endpoint of a call changes due to the operation of an application layer service.

17.1 Example of use of Bearer Redirect with Call Forwarding on No Reply (CFNRy)

Before CFNRy:



Figure 17.1A: Before CFNRy (Network model)

After CFNRy and Bearer Redirection:



Figure 17.1B: CFNRy and Bearer Redirect (Network model)

Figure 17.2 shows the message sequence example for the call forwarding on no reply with Bearer Redirect. In the example, after the call and the bearer towards the access have been released MSC server A requests the MGW A to remove the bearer termination for the served mobile subscriber. MSC server A requests the GMSC server to redirect the bearer to the forwarded to subscriber (interactions towards the access are not shown), using MSC server A as a call control anchor. Once the bearer towards MGW B is established the GMSC server instructs MGW G to connect the incoming bearer to the new bearer (towards MGW B) and informs MSC server A. Once informed that the new bearer has been established MSC server A instructs the GMSC server to removes the old bearer termination (towards MGW A).



Figure 17.2: Information flow for CFNRy with Bearer Redirect (message sequence chart)

18 (G)MSC MGW Tandeming

In all call flow examples a (G)MSC server may tandem (either during call setup or during an active call, as part of an application layer service invocation) the bearer through one or more MGWs under its control, in order to access bearer resources which may be distributed over a number of specialised MGWs.

18.1 Example of use of MSC MGW Tandeming during call setup to provide bearer access to specialised MGW resources

Before MGW tandeming:



Figure 18.1A: MSC before MGW Tandeming during call setup to provide bearer access to specialised MGW resources (Network model)

After MGW tandeming:



Figure 18.1B: MSC after MGW Tandeming during call setup to provide bearer access to specialised MGW resources (Network model)

The figure 18.2 below shows the message sequence example for MSC MGW Tandeming during call setup to provide bearer access to specialised MGW resources. In the example, after the signalling towards MSC server A and the bearer towards the MGW B is established, MSC server A requests the MGW B to tandem the bearer and terminate it at MGW C, where specialised bearer resources which are not available at MGW B may be provided.



Figure 18.2: MSC MGW Tandeming during call setup to provide bearer access to specialised MGW resources (message sequence chart)

19 Timers for bearer independent CS core network

Timer identity	Timer value	Timer started	Timer stopped	Timer expiry
Start_Bearer_E	1 – 14 seconds	Paging procedure is started.	Paging procedure is	The network side bearer
stablishment		Applied only if network side bearer establishment is delayed until paging procedure is completed.	completed or optionally when Call Confirmed message is received.	establishment is started.

20 Multiple IP Realms

Figure 20/1 shows a scenario to support multiple IP realms when the Iu interface or Nb interface supports IP bearer.



Figure 20/1 Multiple IP realms scenario for lu or Nb interface

The RAN1, RAN2, CN1 and CN2 represents separate IP realm. The definition of IP realm is specified in IETF RFC 2663 [41].

The Logical Port 1/2/3/4 represents a logical port connected with the IP realm in the MGW.

For establishing session when multiple IP realms are used in the MGW, the MGC may indicate the IP realm identifier to the MGW. The MGW shall assign the IP termination in the IP realm indicated.

A default IP realm may be configured such that if the MGW has not received the IP realm identifier and the MGW supports multiple IP realms then the default IP realm shall be used.

If the MGW does not support the option to indicate an IP realm then it is free to select an IP port as per implementation permitted in previous releases.

Annex A (informative): Change History

Change history							
Date	TSG #	TSG Doc.	CR	R e v	Subject/Comment	Old	New
Jul 2000					Initial draft	0.0.1	0.0.2
Aug 2000					Comments from TSG-CN WG4#3 incorporated and draft further elaborated	0.0.2	0.1.0
Sep 2000					Contributions and comments from TSG-CN WG4#4 incorporated and draft further elaborated	0.1.0	0.2.0
Oct 2000					Contributions and comments from TSG-CN WG4 Ad Hoc incorporated and draft further elaborated	0.2.0	1.0.1
Nov 2000	CN#10				For information to TSG-CN #10	1.0.1	1.1.0
Jan 2001					Contributions and comments from TSG-CN WG4 #6 incorporated and draft further elaborated.	1.1.0	1.1.1
Feb 2001					New clause structure for handover, editorial corrections.	1.1.1	1.2.0
Feb 2001					Contributions and comments from TSG-CN WG4 Ad Hoc incorporated.	1.2.0	2.0.0
Mar 2001	CN#11	NP-010081			Contributions and comments from TSG-CN WG4 #7 incorporated. Approved in CN#11	2.0.0	4.0.0
Jun 2001	CN#12	N4-010676	002	1	Voice Processing Function Alignment/Clean Up for Call Handover and Relocation	4.0.0	4.1.0
Jun 2001	CN#12	N4-010678	004	1	Corrections to Call Clearing	4.0.0	4.1.0
Jun 2001	CN#12	N4-010680	006	1	Alignment of procedure names to TS 29.232 and editorial changes	4.0.0	4.1.0
Sep 2001	CN#13	NP-010452	800	1	Slight Misalignment of Continuity Message Handling	4.1.0	4.2.0
Sep 2001	CN#13	NP-010452	009		Updates to Clause 9.1	4.1.0	4.2.0
Sep 2001	CN#13				Editorial clean up	4.1.0	4.2.0
Dec 2001	CN#14	NP-010619	012		Correction of Handover/Relocation for Speech and Non- speech calls	4.2.0	4.3.0
Dec 2001	CN#14	NP-010619	014	2	New timer to support long paging in bearer independent network	4.2.0	4.3.0
Dec 2001	CN#14	NP-010619	016	1	Correction for Release of Network Bearer	4.2.0	4.3.0
Dec 2001	CN#14	NP-010631	011		Introduction of MGW Congestion Handling	4.3.0	5.0.0
Dec 2001	CN#14	NP-010631	013	1	Maintenance locking of MG	4.3.0	5.0.0
Mar 2002	CN#15	NP-020029	021		(G)MSC restoration	5.0.0	5.1.0
Mar 2002	CN#15	NP-020029	023	2	Correction of Bearer Modification Handling	5.0.0	5.1.0
Jun 2002	CN#16	NP-020260	024		MSC server GTT enhancement	5.1.0	5.2.0
Jun 2002	CN#16	NP-020260	025	1	Alignment of terminology regarding GERAN access	5.1.0	5.2.0
Jun 2002	CN#16	NP-020248	028		Correction of an incorrect reference in Section 8.3.3.2	5.1.0	5.2.0
Sep 2002	CN#17	NP-020458	026	4	Introduction of GERAN lu-mode	5.2.0	5.3.0
Sep 2002	CN#17	NP-020459	029	1	Misalignment between TS 23205 and TS 29232 for Global Text Telephony	5.2.0	5.3.0
Sep 2002	CN#17	NP-020459	030	1	Misalignment between TS 23.226 and TS 23.205 for Global Text Telephony.	5.2.0	5.3.0
Sep 2002	CN#17	NP-020463	033	1	Correction on wrong message handling for subsequent Handover	5.2.0	5.3.0

Dec 2002	CN#18	NP-020507	031	3	lu-cs over IP related corrections for 23 205	530	540
Dec 2002	CN#10	NP 020596	031	3 2	CAMELA Call Party Handling interworking with Board	520	5.4.0
Dec 2002	CIN#10	NF-020390	035	2	independent CS agree	5.5.0	5.4.0
D 0000	01//40		000	4	Obstition of the termination of the hybrid of an	500	5 4 0
Dec 2002	CN#18	NP-020596	036	1	Clarification of the termination of the lu interface	5.3.0	5.4.0
					components in the Bearer Independent Architecture		
Mar 2003	CN#19	NP-030108	039	1	Bearer Release for Iu CS on IP	5.4.0	5.5.0
Jun 2003	CN#20	NP-030212	042		Clarification of handling of DTMF	5.5.0	5.6.0
Jun 2003	CN#20	NP-030212	044	1	Clarification of handling of DTMF timing	5.5.0	5.6.0
Mar 2003	CN#23	NP-040052	050	1	Use Modify command for change of flow direction to	5.6.0	5.7.0
					bothway		
Dec 2004	CN#26				Release 6 after CN#26	570	600
Mar 2005	CN#27	NP-050057	045	7	Addition of the trace package	600	610
Mar 2005	CN#27	ND 050034	050	' 2	Stage 2 Presedure for Emergency Cell Indigetion	0.0.0	0.1.0
Mar 2005	CIN#Z7	NP-050034	052	2	Stage 2 Procedure for Emergency Call Indication	0.0.0	0.1.0
Mar 2005	CN#27	NP-050028	056		Solving contradiction for Release Cause in Release	6.0.0	6.1.0
					Bearer Procedure between stage 2 and stage 3		
Jun 2005	CT#28	CP-050101	065		Multi-Party Conference call Implementation	6.1.0	6.2.0
Sep 2005	CT#29	CP-050289	059	2	Introducing the ServiceChangeProfile also to the	6.2.0	6.3.0
					relevant stage 2 procedures		
Sep 2005	CT#29	CP-050289	061	2	MGW Recovery clarification	6.2.0	6.3.0
Sep 2005	CT#29	CP-050308	068	1	Alignment of Stage 2 with Continuity test Procedure	6.2.0	6.3.0
Sep 2005	CT#29	CP-050277	071		Introduction of chapter to describe CMM message for	620	630
000 2000	01//20	0. 000211	011		GSM fax	0.2.0	0.0.0
Sep 2005	CT#20	CP-050280	073	1	Notifications for IP Bearer Establishment	620	630
Sep 2005	CT#29	CF-050209	013	1	Miner correction in references	0.2.0	0.3.0
Sep 2005	01#29	CP-050308	082		Minor correction in references	6.2.0	6.3.0
Sep 2005	CT#29	CP-050308	083	1	I ransparent data call request in dual mode case	6.2.0	6.3.0
Dec 2005	CT#30	CP-050608	0095		Correction of Notify establish bearer information	6.3.0	6.4.0
					element	L	
Dec 2005	CT#30	CP-050603	0098		Missing stage 2 definition for Bearer Modified and	6.3.0	6.4.0
					Bearer Modification Failed procedures		
Dec 2005	CT#30	CP-050603	0102		Correction of term Bearer Termination in procedures	6.3.0	6.4.0
					that concern the whole MGW		
Dec 2005	CT#30	CP-050603	0104		Adding missing termination ID to Acknowledgement part	630	640
2000 2000	• • • • • •		0.0.		of notification procedures	0.0.0	
Dec 2005	CT#30	CP-050608	0107		Removing Notify Bearer Modification Information	630	640
2000	01#00	01 000000	0107		element from Prenare IP transport procedure	0.0.0	0.4.0
Dec 2005	CT#20		0100	1	Pearer Beleased Event to Beserve Circuit procedure	620	640
Dec 2005	CT#30	CP-050607	0108	I	stage 2	0.3.0	0.4.0
D 0005	OTUOO	00.050000	0000	_		0.4.0	7 0 0
Dec 2005	CT#30	CP-050629	0089	2	Clean-up of hanging contexts and terminations	6.4.0	7.0.0
Dec 2005	CT#30	CP-050629	0092	1	Corrections for Delayed Backward Bearer Establishment	6.4.0	7.0.0
					with IP Transport	ļ	
Dec 2005	CT#30	CP-050618	0110	2	Correct the UP initialization error in Figure 6.6/1	6.4.0	7.0.0
Mar 2006	CT#31	CP-060064	0117	3	Correction To Termination State Handling	7.0.0	7.1.0
Mar 2006	CT#31	CP-060078	0120	2	Clarification of periodic sending of Audits to recognize	7.0.0	7.1.0
			-		MGW unavailable	-	-
Mar 2006	CT#31	CP-060064	0123	1	Miscellaneous corrections to messages/procedures	700	710
Mar 2006	CT#31	CP-060078	0126	1	Internal bearer for multiparty service	7.0.0	710
Mar 2006	CT#21	CD 060078	0120	1	Wrong implementation of CD 050620 / C4 051767	7.0.0	7.1.0
IVIAI 2000	01#31		0127		(CB#090rov2)	1.0.0	1.1.0
	OTION	0.0.000.70	0.4.0.0		(CR#089feV2)		
Mar 2006	01#31	CP-060078	0128		ServiceUnangeversion in MGW Re-Register procedure	1.0.0	7.1.0
Mar 2006	CT#31	CP-060078	0132		Correct the termination error in figure 13.19	7.0.0	7.1.0
Mar 2006	CT#31	CP-060064	0135	1	Removal Of stream mode modification at bearer release	7.0.0	7.1.0
Mar 2006	CT#31	CP-060064	0141		Replacing Notify Bearer Modification information	7.0.0	7.1.0
					element with Notify Bearer Event in		
					Prepare IP Transport procedure.		
Jun 2006	CT#32	CP-060301	0150	1	Multimedia & CSD calls	7.1.0	7.2.0
lun 2006	CT#32	CP-060315	0153	1	Solving contradiction between tunnel ontion in 23 205	710	720
00112000	01#02	01 000010	0100	'	and 20 232	7.1.0	1.2.0
Dec 2006	CT#24		0165		Amend the call flow of handover precedure	720	720
Dec 2006	01#34		0100	~		1.2.0	1.3.0
		CP-060569	0159	5	Ennancements for VGCS/VBS in SPLIT architecture		
		CP-060569	0167	1	Example of in-band information sending with the	1	
					topology descriptor	ļ	
Mar 2007	CT#35	CP-070014	0172	3	IP domain connection indication	7.3.0	7.4.0
			0173	1	Clarify message sequence of VGCS call		
Jun 2007	CT#36	CP-070322	0177		ServiceChangeMGCId parameter	7.4.0	7.5.0
Sep 2007	CT#37	CP-070521	0180	3	Correction To Call Hold	750	760
1000	501	5. 010021	0.00	~			

	CP-070530	0185	2	Establish internal Nb connections via IPBCP in MPTY]	
	CP-070530	0190	3	Support inactivity timer (H.248.14) for MGWs detecting		
				the failure of MGC		
Sep 2007 CT#37	CP-070569	0191	2	Mandatory use termination heartbeat (H.248.36) when	7.6.0	8.0.0
				establish bearer/prepare bearer		
Dec 2007 CT#38	CP-070749	0196	1	Number of needed listener context terminations to ASCI	8.0.0	8.1.0
			_	package		
		0192	1	Clarifications to Send Tone / Play Announcement procedures		
Mar 2008 CT#39	CP-080011	0200		Detect DTMF event after reporting of DTMF	8.1.0	8.2.0
	CP-080025	0204	1	Private communication during a Multiparty call		
Jun 2008 CT#40	CP-080271	0207	3	Clarification of several inconsistencies in procedure	8.2.0	8.3.0
		0208	1	Clarification of subscribers" roles in CW procedure		
		0211	1	Protocol negociation result procedures between MSC		
		0000	~	Server and MGVV	-	
		0209	3	Support for A interface user plane over IP	-	
0 0000 OT#44	00.000400	0210	2	Revision of scope of 15 23.205	0.0.0	0.4.0
Sep 2008 C1#41	CP-080466	0213	1	Handover Scenarios for AoIP	8.3.0	8.4.0
		0215		Call Clearing for A Interface on IP	-	
		0216	1	IP interface type indicator	-	
		0217		Use of Global Text Telephony with A over IP		
		0218	1	Combining certain procedures with "Reserve RTP		
D 0000 0T#40	00 000705	0000			0.4.0	0.5.0
Dec 2008 C1#42	CP-080705	0220	1	Corrections to Call Forwarding procedures	8.4.0	8.5.0
	CP-080710	0222	1	Enhanced SRNS relocation	-	
	CP-080697	0224	2	BSS Internal Handover procedure	-	
	CP-080697	0225		Interface'		
	CP-080695	0226	4	Signalling of CAT Capability	1	
	CP-080697	0228	2	Impact on HandoverRelocation and Interactions with	1	
				Other Services for AoIP		
Jun 2009 CT#44	CP-090312	0231	2	Customized alerting tone indicators	8.5.0	8.6.0
	CP-090309	0232		BSS Internal Handover procedure		
				Correct previous entry in history table	8.6.0	8.6.1
Sep 2009 CT#45	CP-090541	0234		Clarification of Use of Reserve and Configure RTP	8.6.1	8.7.0
				Connection Point		
	CP-090541	0235		IWF Rate Change Procedure with AoIP		
	CP-090541	0236		Removal of Editor's Notes for IP Interface Type		
2009-12 -	-	-	-	Update to Rel-9 version (MCC)	8.7.0	9.0.0
2010-09 CT#49	CP-100450	0239		Terminating Access Bearer Assignment	9.0.0	9.1.0
2011-03 CT#51	CP-110046	0243	-	BSS Internal Handover	9.1.0	9.2.0
2011-06 CT#52	CP-110352	0246	1	Adding Local Connection Address request to Reserve	9.2.0	9.3.0
				and Configure RTP Connection point		
2011-09 CT#53	CP-110563	0248		Media inactivity notification procedure	9.3.0	9.4.0

History

Document history						
V9.0.0	February 2010	Publication				
V9.1.0	October 2010	Publication				
V9.2.0	April 2011	Publication				
V9.3.0	June 2011	Publication				
V9.4.0	October 2011	Publication				